

Fred James Chief Regulatory Officer Phone: 604-623-4046 Fax: 604-623-4407 bchydroregulatorygroup@bchydro.com

March 19, 2019

Mr. Patrick Wruck Commission Secretary and Manager Regulatory Support British Columbia Utilities Commission Suite 410, 900 Howe Street Vancouver, BC V6Z 2N3

Dear Mr. Wruck:

RE: British Columbia Utilities Commission (BCUC or Commission) British Columbia Hydro and Power Authority (BC Hydro) E^xponent Report on Electric and Magnetic Fields and Health: Review of the Scientific Research from January 2017 to December 2018 (the Report) Compliance with Directive 4 of Commission Order No. C-4-06

BC Hydro writes in compliance with Directive 4 of the Commission's Order No. C-4-06 dated July 7, 2006 in regards to the British Columbia Transmission Corporation Vancouver Island Transmission Reinforcement Project and encloses the Report.

The enclosed report prepared by Exponent is the fifth such report reviewing and updating the scientific research pertaining to the potential health impacts of exposure to Extremely Low Frequency Electric and Magnetic Fields (**ELF-EMF**) to be submitted to the Commission. BC Hydro notes that consistent with all previous reports, the recent research results, including the scientific literature that has been reviewed in this Report, does not provide new evidence to alter the conclusion that long-term exposure to ELF-EMF is not known to cause any adverse health effects, including cancer or other illnesses.

The cost of each of these Reports is approximately \$25,000 (USD). Given that the research results reviewed over the past 12 years has consistently concluded that long-term exposure to ELF-EMF is not known to cause any adverse health effects, including cancer or other illnesses, BC Hydro is requesting relief from Directive 4 of Order C-4-06, requiring BC Hydro to file a report every two years. Based on the results from the last five reports, BC Hydro respectfully submits incurring this frequency of reporting, and the inherent cost, is no longer warranted going forward.

The purpose of the Exponent report was to keep the Commission up to date on the research into the health impacts of exposure to ELF-EMF. Based on the consistent findings showing that long-term exposure to ELF-EMF is not known to cause any adverse health effects, continuing to require BC Hydro to provide a report every two



years imposes, in BC Hydro's view, unnecessary costs on BC Hydro ratepayers. As a result, BC Hydro recommends either eliminating the requirement to continue to report on the long-term health effects of exposure to ELF-EMF, or at the very least, reduce the frequency of the required reporting.

BC Hydro may produce the Exponent Report on an as-needed basis in support of stand-alone capital project applications that involve the addition to or modification of the bulk transmission system. In transmission project applications, stakeholders often question the impacts of such a project on the local community in relation to ELF-EMF exposure. If needed, BC Hydro would update the review of the literature on health effects to support these major project applications. BC Hydro would determine the appropriate timeframe to update the report in support of its transmission projects, and would include the updated report in its project applications. BC Hydro submits this should be sufficient to continue to keep the Commission updated on the research into the health effects of exposure to ELF-EMF.

For further information, please contact Geoff Higgins at 604-623-4121 or by email at <u>bchydroregulatorygroup@bchydro.com</u>.

Yours sincerely,

Fred James Chief Regulatory Officer

gh/tl

Enclosure

Exponent®

Health Sciences Practice

Electric and Magnetic Fields and Human Health: Review and Update of the Scientific Research from January 1, 2017 to December 31, 2018

Exponent

Electric and Magnetic Fields and Human Health: Review and Update of the Scientific Research from January 1, 2017 to December 31, 2018

Prepared for

BC Hydro 333 Dunsmuir St. Vancouver, B.C. V6B 5R3

Prepared by

Exponent 149 Commonweals Drive Menlo Park, California 94025

March 4, 2019

© Exponent, Inc.

1809838.000 - 7732

Contents

			I	Page
List	of Fig	gures		iv
List	of Ta	bles		v
Acr	onyms	s and Ab	obreviations	vi
Lim	itatio	ns		vii
Exe	cutive	Summa	ıry	viii
Intr	oduct	ion		1
1	Bac	kground	: Electric and Magnetic Fields	3
2	Met	hods for	• evaluating scientific research	6
	2.1	Heath	risk assessment approach	6
	2.2	Hazard	identification/weight-of-evidence review	6
	2.3	Evalua	tion of epidemiologic studies	8
		2.3.1	Association vs. causation	12
		2.3.2	Meta- and pooled analyses	14
		2.3.3	Assessment of EMF exposure in epidemiologic studies	15
	2.4	Evalua	tion of experimental research	17
		2.4.1	General research methods	17
		2.4.2	Experimental methods for cancer research	19
		2.4.3	Experimental methods for developmental toxicity	20
		2.4.4	Evaluating the cumulative body of experimental evidence	21
3	Con	clusions	of weight-of-evidence reviews of EMF and health	23
	3.1	Weight	t of evidence reviews by national and international scientific agencies	23
	3.2	Standa	rds and guidelines for limiting exposure to EMF	28
		3.2.1	Status of EMF guidelines	28
		3.2.2	Comparison of ICES and ICNIRP guidelines	30
		3.2.3	Implications for human health	31
	3.3	Precau	tionary approaches	32
		3.3.1	General definition	32

		3.3.2	WHO recommendations regarding precautionary measures for EMF	33
		3.3.3	Canadian perspective on precautionary approaches	34
4	4 Human Health Research			
	4.1	Cancer		37
		4.1.1	Childhood leukemia	37
		4.1.2	Childhood brain cancer	43
		4.1.3	Breast cancer	45
		4.1.4	Other adult cancers	46
		4.1.5	In vivo studies of carcinogenesis	51
	4.2	Reprodu	active and developmental effects	61
	4.3	Neurode	egenerative disease	65
	4.4	Electror	nagnetic hypersensitivity	70
5	Poss Devi	ible Effe ces	cts of ELF Electric and Magnetic Fields on Implantable Cardiac	73
6	Faur	na and F	lora Research	78
	6.1	Fauna		78
	6.2	Flora		80
Glos	sary			81
Refe	rence	S		85

List of Figures

		Page
Figure 1.	Basic design of cohort and case-control studies	10
Figure 2.	Interpretation of an odds ratio in a case-control study	10
Figure 3.	Basic IARC method for classifying exposures based on evidence for potential carcinogenicity	12

List of Tables

Page

Table 1.	Hill's guidelines for evaluating causation in epidemiologic data*	13
Table 2.	Criteria for evaluating experimental studies as applied to EMF exposures*	22
Table 3.	Reference levels for whole body exposure to 60-Hz fields: general public	29
Table 4.	Studies of childhood leukemia (2017-2018)	42
Table 5.	Studies of childhood brain cancer (2017-2018)	44
Table 6.	Studies of adult brain cancer (2017-2018)	49
Table 7.	Studies of adult leukemia/lymphoma (2017-2018)	51
Table 8.	Studies of in vivo carcinogenesis (2017-2018)	61
Table 9.	Studies of reproductive and developmental effects (2017-2018)	65
Table 10.	Studies of neurodegenerative diseases (2017-2018)	69
Table 11.	Studies of electromagnetic hypersensitivity (2017-2018)	72
Table 12.	Studies of EMI (2017-2018)	77

Acronyms and Abbreviations

r 111				
AC AI	Alternating current			
ACGIH A	American Conference of Governmental Industrial Hygienists			
ALL A	Acute lymphoblastic leukemia			
ALS A	Amyotrophic lateral sclerosis			
CI Co	Confidence interval			
CNS Ce	entral nervous system			
DMBA 7,	12-dimethylbenz[a]anthracene			
EFHRAN Eu	European Health Risk Assessment Network on Electromagnetic Fields			
Ex	xposure			
ELF Ex	xtremely low frequency			
EMF El	lectric and magnetic fields			
EMI El	lectromagnetic interference			
FPTRPC Fe	ederal-Provincial-Territorial Radiation Protection Commission			
G Ga	auss			
Hz He	ertz			
IARC In	ternational Agency for Research on Cancer			
ICD m	plantable cardioverter-defibrillator			
ICES In	ternational Committee for Electromagnetic Safety			
ICNIRP In	ternational Commission on Non-Ionizing Radiation Protection			
IEI Id	liopathic environmental intolerance			
kV/m Ki	ilovolts per meter			
MDA M	lalondialdehyde			
mG M	lilligauss			
OR O	dds ratio			
RR Re	elative risk			
SCENIHR Sc	cientific Committee of Emerging and Newly Identified Health Risks			
SOD Su	uperoxide dismutase			
TWA Ti	ime-weighted average			
V/m Ve	olts per meter			
WHO W	Vorld Health Organization			

Limitations

At the request of BC Hydro, Exponent prepared this summary report on the status of research related to power frequency electric and magnetic field exposure and health. The findings presented herein are made to a reasonable degree of scientific certainty. This report is limited to the papers reviewed and may not include all information in the public domain. Exponent reserves the right to supplement this report and to expand or modify opinions based on review of additional material as it becomes available, through any additional work, or review of additional work performed by others.

The scope of services performed during this investigation may not adequately address the needs of other users of this report, and any re-use of this report or its findings, conclusions, or recommendations presented herein are at the sole risk of the user. The opinions and comments formulated during this assessment are based on observations and information available at the time of the investigation. No guarantee or warranty as to future life or performance of any reviewed condition is expressed or implied.

Executive Summary

This report was prepared at the request of BC Hydro to provide a summary and overview on the status of scientific research related to extremely low frequency (ELF) electric and magnetic fields (EMF) exposure and health. This report also fulfills a recurring directive from the British Columbia Utilities Commission to monitor and report on ELF EMF research on a regular basis.

Electric and magnetic fields are produced by both natural and man-made sources that surround us in our daily lives. Power-frequency EMF, part of the ELF range of the electromagnetic spectrum that includes frequencies up to 300 Hertz (ICNIRP, 1998), are invisible fields surrounding all objects that generate, use, or transmit electricity. People are almost constantly exposed to ELF EMF in their homes, workplaces, schools, hospitals, and other environments, because the use of electricity and the supporting electricity network are essential parts of technologically advanced societies. Sources of ELF EMF in our everyday environment include, for example, appliances, wiring in homes, and electric motors, as well as distribution and transmission lines.

This report provides an overview of scientific methods used for studying potential health effects of environmental exposures, specifically reviews the scientific disciplines most relevant for human health (epidemiologic and laboratory animal studies), and reviews methods used for health risk assessments. This report then identifies relevant studies of childhood cancers, adult cancers, reproductive and developmental effects, neurodegenerative diseases, electromagnetic hypersensitivity, *in vivo* experimental studies on carcinogenesis, and fauna/flora studies published from January 1, 2017 to December 31, 2018. These studies were systematically reviewed and evaluated.

Since the late 1970s, potential health effects related to ELF EMF have been the focus of extensive scientific research. Because of the amount and complexity of the scientific studies in this area, comprehensive evaluations of the available scientific evidence have been performed for health and scientific agencies by panels comprised of independent scientists with expertise in relevant scientific disciplines. The general public and policy makers should look to the

conclusions of reviews such as these for guidance. In the past two decades a number of national and international health and scientific agencies have assembled panels of scientists that conducted comprehensive evaluations of the scientific literature to assess if the evidence supports a causal relationship between exposure to ELF EMF and adverse human health effects.

One of the most comprehensive health risk assessments of the EMF ELF literature that critically reviewed the cumulative epidemiologic and laboratory research was published by the World Health Organization in a 2007 report. Similar evaluations also were conducted in prior years by the National Institute for Environmental Health Sciences in the United States, the International Agency for Research on Cancer, the Federal-Provincial-Territorial Radiation Protection Committee in Canada, and the National Radiological Protection Board in the United Kingdom, among others. More recent evaluations have been conducted by the Swedish Radiation Safety Authority and the European Union's Scientific Committee on Emerging and Newly Identified Health Risks. Overall, none of these agencies has concluded that long-term exposure to ELF EMF is known to cause any adverse health effect, including cancer and other illnesses. Recent research results, including the scientific literature that have been reviewed in this report, do not provide new evidence to alter this conclusion.

Introduction

Exponent was requested by BC Hydro to prepare a summary of the current research related to extremely low frequency (ELF) electric and magnetic fields (EMF) and health. This report provides an update to Exponent's 2007, 2010, 2012, and 2017 reports.¹ The previous Exponent reports evaluated research results published up to December 31, 2016, and assessed their potential impact on the conclusions reached by the World Health Organization (WHO) in its comprehensive risk assessment that reviewed research through 2005 (WHO, 2007). The current report evaluates research published between January 1, 2017 and December 31, 2018, to determine if new research developments justify changes to the conclusions of previous weight-of-evidence reviews. This report also provides an update to the British Columbia Utilities Commission on the status of EMF health research since 2016. Health Canada also recognizes that "*the current state of scientific knowledge with respect to possible health effects from EMF exposure and a review of current exposure guidelines and/or position statements from health-related organizations*" is a key element for its assessment of the potential effects of [EMF exposure for] a proposed project on human health" (Health Canada, 2010, pp. 7, 4).

This report follows the general structure of the previous Exponent reports to BC Hydro and discusses the scientific topics covered in the previous reports. Sections 1 and 2 of this report provide the reader with a framework for understanding the discussion in later sections. Section 1 provides background information on EMF, and Section 2 outlines the standard scientific methods used to evaluate research. Section 3 summarizes the conclusions of recent weight-of-evidence reviews of ELF EMF prepared by scientific organizations. Section 4 provides an evaluation of epidemiologic studies on selected health outcomes (childhood cancers, adult cancers, reproductive and developmental effects, neurodegenerative diseases), *in vivo* experimental studies on carcinogenesis, and the phenomena of electromagnetic hypersensitivity,

¹ Electric and Magnetic Fields and Health: Review of the Scientific Research from March 1, 2012, to December 31, 2016 (Exponent, 2017); EMF and Health – Comprehensive Review and Update of the Scientific Research, January 15, 2010 through March 1, 2012 (Exponent, 2012); EMF and Health – Review and Update of the Scientific Research, September 2007 through January 2010 (Exponent, 2010); EMF and Health – Review and Update of the Scientific Research (Exponent, 2007).

published from January 1, 2017² to December 31, 2018, as identified through a systematic review of the literature. Sections 5 and 6 address additional topics with relevance to an EMF risk assessment. A glossary of scientific terms is included at the end of the report to provide additional clarification.

² A few studies published in 2016 were included in this review if they had not been included in the earlier review because they were not available at the time that review was completed.

1 Background: Electric and Magnetic Fields

Electric and magnetic fields are produced by both natural and man-made sources that surround us in our daily lives. Man-made EMF is found wherever electricity is generated, delivered, or used, including near power lines, wiring in homes, workplace equipment, electrical appliances, power tools, and electric motors. In North America, EMF from these sources changes direction and intensity 60 times, or cycles, per second—a frequency of 60 Hertz (Hz)—and are often referred to as power-frequency EMF.³ Power-frequency EMF is part of the ELF range that includes frequencies up to 300 Hz (ICNIRP, 1998). Natural sources of EMF include, for example, the earth's static magnetic field and the electric fields created by the normal functioning of our nervous and cardiovascular system.

Electric fields occur as the result of the voltage applied to electrical conductors and equipment. Electric-field levels are expressed in measurement units of volts per meter (V/m) or kilovolts per meter (kV/m); 1 kV/m is equal to 1,000 V/m. Electric fields are easily blocked by most objects such as buildings, walls, trees, and fences. As a result, the major indoor sources of electric fields are the many appliances and equipment we use within our homes and workplaces. Electric-field levels increase in strength as voltage increases and are present even if an electrical device is turned off but plugged in; field strength diminishes quickly, however, as one increases distance from the source.

Magnetic fields are produced by the movement of electricity. Magnetic-field levels are expressed as magnetic flux density in units called gauss (G), or in milligauss (mG), where 1 G equals 1,000 mG.⁴ The magnetic-field level associated with a particular object (e.g., an appliance or power line) depends largely on various operating characteristics of the source and on the amount of current (i.e., electricity) flowing through the object. Unlike electric fields, magnetic fields are only present when an appliance or electrical device is turned on or a power line is energized. Similar to electric fields, magnetic fields diminish in strength quickly as

³ Electrical facilities in many countries outside North America operate at a frequency of 50 Hz.

⁴ Scientists also refer to magnetic flux density at these levels in units of microtesla (μ T). Magnetic flux density in milligauss units can be converted to microtesla by dividing by 10 (i.e., 1 mG = 0.1 μ T).

distance increases from the source, but unlike electric fields they are not easily blocked by conductive objects.

ELF EMF is ubiquitous in modern society because of the abundance of electrical sources in our environments. Every person's average EMF exposure is defined by the environments where they spend time, the sources they encounter in those locations, and the duration of any exposure; any substantial changes to these variables may result in a change in average exposure. If someone worked as a welder or lived in a home with faulty wiring, for example, his or her average EMF exposure may be elevated during these periods. This ubiquitous and changing nature of EMF exposure makes it difficult to describe and quantify.

Electric fields in the home range up to approximately 0.01 kV/m in the center of rooms (away from appliances) and up to 0.25 kV/m near appliances (WHO, 1984). In most homes, the magnetic-field level measured in the center of rooms (away from appliances) is approximately 1 mG, resulting principally from indoor sources (Zaffanella, 1993). Based on a sample taken in the United States, the estimated daily average exposure to magnetic fields is approximately 1-2 mG for about 76% of the population (Zaffanella, 1997). In Canada, the average magnetic-field exposure in a sample of 382 children from five provinces, including British Columbia, was measured as 1.2 mG using wearable personal magnetic-field meters (Deadman et al., 1999). While higher magnetic-fields levels may be measured immediately under distribution and transmission lines, the distance of most buildings from a power line's right-of-way reduces the effect of these sources on magnetic-field levels measured inside a home or office, since the intensity of magnetic fields diminishes quickly with distance from the source. In fact, typical sources of the highest magnetic fields encountered indoors are electrical appliances. For example, a publication by the U.S. National Institute of Environmental Health Sciences (NIEHS, 2002) reported that the median magnetic field measured at 6 inches from a sample of appliances was 6 mG (baby monitor), 7 mG (color televisions), 9 mG (electric oven), 14 mG (computers), 90 mG (copier), 200 mG (microwave ovens), 300 mG (hair dryer), and 600 mG (can opener).⁵

⁵ Mobile phones and their antennas, wireless communication networks, and radios of all types (AM, FM, police, and fire) operate using radiofrequency fields, which represent a frequency (i.e., millions and billions of Hz) within the electromagnetic spectrum much higher than ELF EMF.

Because the frequency of electromagnetic energy is a key factor in determining its interaction with living things, and the interaction mechanisms relevant for ELF EMF are very different from those relevant for higher frequency fields (e.g., radio frequency or solar energy), only studies of ELF EMF are directly relevant to assessing the potential biological and health effects of power-frequency fields. The focus of this report is on power-frequency EMF (i.e., the ELF EMF fields produced by the generation, transmission, and use of electricity); thus, only ELF EMF studies will be reviewed in this report.⁶

⁶ The major focus of the review is magnetic-field exposure. Research has focused on magnetic fields because, among other reasons, conductive objects effectively shield electric fields, and power lines have little effect on the potential long-term average electric-field exposure of nearby residents.

2 Methods for evaluating scientific research

2.1 Heath risk assessment approach

The standard scientific method for determining whether an exposure in the environment (such as chemical, physical, or biological agents) can affect human health is a health risk assessment.⁷ Health risk assessments include four general steps: hazard identification, dose-response assessment, exposure assessment, and specific risk characterization. The process starts with a systematic identification and evaluation of the entire relevant body of research to determine if any health risks are associated with an exposure (hazard identification/weight-of-evidence review).⁸ A follow-up question to hazard identification is, "if the exposure does cause any health risks, at what level do they occur?" (dose-response assessment). A risk assessment then characterizes the exposure circumstances of the situation under consideration (exposure assessment). Finally, using the findings from the hazard identification and dose-response assessment as a basis, a summary evaluation is provided (risk characterization).

2.2 Hazard identification/weight-of-evidence review

Science is more than a collection of facts; rather, it is a method of obtaining information and of reasoning to ensure that the information is accurate and correctly describes physical and biological phenomena. Many misconceptions in human reasoning occur when people casually observe and interpret their observations and experience (e.g., if a person develops a headache after eating a particular food, he or she may mistakenly ascribe the headache to the food). The proximity or co-occurrence of events or conditions, however, does not necessarily indicate a causal relationship. Scientists use systematic methods to evaluate observations and assess the potential impact of a specific agent on human health.

⁷ Some of the scientific panels that have reviewed EMF research have described the risk assessment process in the introductory sections of their reviews or in separate publications (ICNIRP, 2002; IARC, 2006; SCENIHR, 2007; SSI, 2007; WHO, 2007; HCN, 2009a; SSM, 2010; SCENIHR, 2012; SCHEER, 2018).

⁸ The terms weight-of-evidence review and hazard identification are used interchangeably in this report to denote a systematic review process involving the review of experimental and epidemiologic research to arrive at conclusions about possible health risks.

The scientific process involves looking at *all* the evidence on a particular issue in a systematic and thorough manner (i.e., a weight-of-evidence review or hazard identification). This process is designed to ensure that more weight is given to studies of better quality and that studies with a given result are not selected out from the available evidence to advocate or suppress a preconceived idea of an adverse effect. Three broad steps define a weight-of-evidence review: a systematic search of the published literature to identify relevant studies, an evaluation of each identified study to determine its strengths and weaknesses, and an overall evaluation of the data, giving more weight to higher-quality studies.

Data from several types of studies must be evaluated *together* in a weight-of-evidence review, including epidemiologic observations in people, experimental studies in animals (*in vivo*), and experimental studies in isolated cells and tissues (*in vitro*). Epidemiologic and experimental studies complement one another because the inherent limitations of epidemiologic studies are addressed in experimental studies and *vice versa*. Similar to puzzle pieces, the results of epidemiologic and experimental studies are placed together to provide a picture of the possible relationship between exposure to a particular agent and disease.

Epidemiology is the scientific discipline that studies the patterns of disease occurrence in human populations and the factors that influence those patterns. Epidemiologic studies are critical for determining the causes of diseases and play a primary role in a human health risk assessment. Epidemiologic studies are observational in nature, in that they examine and analyze people in their normal lives with the investigators having little control over the many factors that affect disease. Such studies are designed to quantify and evaluate the association between exposures (e.g., a high fat diet) and health outcomes (e.g., coronary artery disease). An association is a statistical measure of how things vary together. Scientists may report, for example, that people with coronary artery disease eat a diet that is lower in fiber content compared to people without the disease (i.e., a negative association), or that persons with coronary artery disease eat a diet that is higher in fat compared to persons without the disease (i.e., a positive association). Epidemiologic studies can identify factors that may contribute to the development of disease but typically they are not used as the sole basis for drawing inferences about cause-and-effect relationships. Additional results from experimental research needs to be considered as well.

Page 17 of 116

In contrast to epidemiologic studies, experimental studies are conducted under controlled laboratory conditions designed to test specific hypotheses. *In vivo* studies can strictly control and measure the exposure levels in the exposed groups as well as control and measure other factors such as food intake, housing conditions, and temperature that may have an effect on the outcome in all groups of exposed and unexposed animals. Generally, experimental studies are required to establish cause-and-effect relationships, but the results of experimental studies by themselves may not always be directly extrapolated to predict effects in human populations. Therefore, it is both necessary and desirable that biological responses to agents that could present a potential health threat be explored by epidemiologic methods in human populations, as well as by experimental studies in the research laboratory.

A weight-of-evidence review is essential for arriving at a valid conclusion about causation because no individual study is capable of assessing causation independently. Rather, evaluating causation is an inferential process that is based on a comprehensive assessment of all the relevant scientific research. The final conclusion of a weight-of-evidence review is a conservative evaluation of the strength in support of a causal relationship. If a clear causal relationship is indicated by the data, the conclusion is that the exposure is a known cause of the disease. In most cases, however, because of limitations in study methods, the relationship is not clear and the exposure is characterized as probably related, possibly related, unclassifiable, or probably not related (IARC, 2006). Few exposures are categorized as either *known* or *unlikely* causes of cancer (IARC, 2006).

2.3 Evaluation of epidemiologic studies

This section briefly describes the two most commonly used and most informative designs used in epidemiologic studies (cohort and case-control designs) and the major issues that are relevant to evaluating their results.

A case-control study (Figure 1) compares the characteristics of people who have been diagnosed with a disease (i.e., cases) to a group of people who do not have the disease (i.e., controls). The prevalence and extent of past exposure to a particular agent is estimated in both groups to assess whether the cases have a higher exposure level than the controls, or *vice versa*.

In a case-control study, an odds ratio (OR) is used to estimate the association quantitatively. An OR is the ratio of the odds of being exposed among the cases to the odds of being exposed among the controls. If an OR is equal to 1.0, the general interpretation is that there is no association between the exposure and disease in the study. If the OR is greater than 1.0, there is a positive association between the exposure and disease in the study and the inference is that the exposure may increase the risk of the disease (Figure 2). A negative association is indicated when the OR is less than 1.0. Epidemiologists typically quantify the precision of the estimated measures of association by calculating confidence intervals (CI), which is the margin of error, usually set at 95% by convention, around the point estimates. The 95% CI represents a range of values that are expected to include the underlying effect estimate in the population 95% of the time if samples for studies were repeatedly drawn from the underlying population. When the 95% CI for the effect estimate excludes the null value of 1.0, the result is also commonly referred to as statistically significant.

In a cohort study, the researchers start with the identification of a pre-defined study population (i.e., individuals who are free of the disease), determine their exposure status, then follow them over time to see if persons with a certain exposure develop disease at a higher or lower rate compared to unexposed persons (Figure 1). Cohort studies are evaluated statistically in a similar manner as case-control studies, although the risk estimate is referred to as a relative risk (RR). The RR is equal to the risk of disease in the exposed group divided by the risk of disease in the unexposed group, with values greater than 1.0 suggesting that the exposed group has a higher risk of disease.



Figure 1. Basic design of cohort and casecontrol studies

Figure 2. Interpretation of an odds ratio in a case-control study

A RR or OR value is simply a measure of association (i.e., how often a disease and exposure occur together in a particular study population); it does not mean that there is a known or causal relationship. Before any conclusions can be drawn, all studies of the relationship between the exposure and disease must be identified and evaluated to determine the possible role that other factors such as chance, bias, and confounding may have played in the study's results.

- Chance in epidemiologic studies refers to random error that may result from sampling variability, or imprecision in measurements of study variables, including exposure, outcome, and confounders. The probability that a given finding is due to chance may be estimated by statistical methods, such as significance testing, or calculation of CIs.
- Bias refers to any error in the design, conduct, or analysis of a study resulting in a distorted estimate of an exposure's effect on the risk of disease. There are many different types of bias; for example, selection bias may occur if the characteristics of cases that participate in a study differ in a meaningful way from the characteristics of those subjects who do not participate (e.g., if cases who live near a power line are more likely to participate in the study than controls because they are concerned about a possible exposure, cases will end up living closer to power lines than controls in the study sample just because of the selection process and the differential willingness to participate between cases and controls).
- Confounding is a situation in which an association is distorted because the exposure being studied is associated with other risk factors for the disease. For example, a link between coffee drinking in mothers and low birth weight babies may be observed in a study. Some women who drink coffee, however, may also smoke cigarettes. When the smoking habits of mothers are taken into account, coffee drinking may not be associated with low birth weight babies because the confounding effect of smoking has been removed.

As part of the weight-of-evidence review process, each study's design and methods are critically evaluated to determine if and how chance, bias, and confounding may have affected the results, and, as a result, the weight that should be placed on the study's findings.

A formal procedure for classifying scientific data has been developed by the International Agency for Research on Cancer (IARC). The IARC classifies epidemiologic and *in vivo* studies as providing sufficient, limited, or inadequate evidence (Figure 3) in support of carcinogenicity, or evidence suggesting a lack of carcinogenicity. In epidemiologic studies, the role of chance, bias, and confounding on the observed association must be ruled out with reasonable confidence to designate the evidence as sufficient. If the role these factors may play in the calculated statistical association cannot be ruled out with reasonable confidence, then the data is classified as providing limited evidence. Inadequate evidence describes a data set that lacks quality, consistency, or power for conclusions to be drawn regarding causality. The categories on the left in Figure 3 (e.g., known, probable, etc.) are based on the combined evaluations of epidemiologic and *in vivo* studies. Other biological data relevant to the evaluation of carcinogenicity and its mechanisms are considered, depending on the relevance to the agent under study.

	Epidemiology Studies				Animal Studies			
	Sufficient evidence	Limited evidence	Inadequate evidence	Evidence suggesting lack of carcinogenicity	Sufficient evidence	Limited evidence	Inadequate evidence	Evidence suggesting lack of carcinogenicity
Known Carcinogen	~							
Probable Carcinogen		~			~			
Possible Carcinogen		~				~	~	
Not Classifiable			V			~	V	
Probably not a Carcinogen				V				~

Sufficient evidence in epidemiology studies—A positive association is observed between the exposure and cancer in studies, in which chance, bias and confounding were ruled out with "reasonable confidence."

Limited evidence in epidemiology studies—A positive association has been observed between the exposure and cancer for which a causal interpretation is considered to be credible, but chance, bias or confounding could not be ruled out with "reasonable confidence."

Inadequate evidence in epidemiology studies—The available studies are of insufficient quality, consistency or statistical power to permit a conclusion regarding the presence or absence of a causal association between exposure and cancer, or no data on cancer in humans are available.

Evidence suggesting a lack of carcinogenicity in epidemiology studies—There are several adequate studies covering the full range of levels of exposure that humans are known to encounter, which are mutually consistent in not showing a positive association between exposure to the agent and any studied cancer at any observed level of exposure. The results from these studies alone or combined should have narrow confidence intervals with an upper limit close to the null value (e.g. a relative risk of 1.0). Bias and confounding should be ruled out with reasonable confidence, and the studies should have an adequate length of follow-up. Sufficient evidence in animal studies—An increased incidence of malignant neoplasms is observed in (a) two or more species of animals or (b) two or more independent studies in one species carried out at different times or indifferent laboratories or under different protocols. An increased incidence of tumors in both sexes of a single species in a well-conducted study, ideally conducted under Good Laboratory Practices, can also provide sufficient evidence.

Limited evidence in animal studies—The data suggest a carcinogenic effect but are limited for making a definitive evaluation, e.g. (a) the evidence of carcinogenicity is restricted to a single experiment; (b) there are unresolved questions regarding the adequacy of the design, conduct or interpretation of the studies; etc.

Inadequate evidence in animal studies—The studies cannot be interpreted as showing either the presence or absence of a carcinogenic effect because of major qualitative or quantitative limitations, or no data on cancer in experimental animals are available

Evidence suggesting a lack of carcinogenicity in animal studies—Adequate studies involving at least two species are available which show that, within the limits of the tests used, the agent is not carcinogenic.

Figure 3. Basic IARC method for classifying exposures based on evidence for potential carcinogenicity

2.3.1 Association vs. causation

As discussed earlier, an association is a relationship between two events, a finding that they occur together more often than expected by chance. A reported association, even a statistically significant association, between a particular exposure and disease, however, is not sufficient evidence to conclude that the exposure is a cause of the disease. Rather, an association is a

finding from a particular study; evaluating causation is an inferential process that combines the totality of evidence (including epidemiologic studies that have measured associations) in a weight-of-evidence review.

In order to support a cause-and-effect relationship, the overall data, or evidence, must present a logically coherent and consistent picture. Various guidelines have been used to assist in the evaluation of the plausibility of a cause-and-effect relationship between a particular exposure and disease. These guidelines, commonly referred to as *Hill's criteria* after the British physician who outlined them (Hill, 1965), typically form the foundation of causal inference (Rothman and Greenland, 1998). Since the publication of *Hill's criteria* in 1965, numerous revisions and updates have been suggested (e.g., Susser, 1991), although the basic tenets remain the same. As described in Table 1, *Hill's criteria* are used as an analytic framework in the weight-of-evidence review process (e.g., ICNIRP, 2002; USEPA, 2005).

Each criterion cannot be addressed with a simple "yes" or "no," nor are the criteria as a whole meant to be an inflexible set of rules; rather, they serve as guidance for weighing the evidence to reach a decision about the plausibility of a cause-and-effect relationship. The more firmly these criteria are met by the data, the more convincing the evidence. Hill also noted that, while formal tests of significance do not establish causation, the proposed guidelines were intended for evaluation of associations where chance was eliminated as a potential explanation (Hill, 1965).

Strength	The stronger the association between the disease and the exposure in question, the more persuasive the evidence. Smaller relative risks are more likely to be result of bias or confounding.
Consistency Consistent results across different study populations and study designs are mo convincing than isolated observations.	
Specificity	The evidence for causation is stronger if the exposure produces a specific effect.
Dose-response	If the risk of disease increases as the exposure level increases (e.g., from low to high exposure), the exposure is more likely to be related to the disease.
Biological plausibility	Epidemiologic results are much more convincing if they are coherent with what is known about biology. That is, the evidence is stronger if scientists know of a biological mechanism that can explain the effect.
Temporality	The data must provide evidence of correct temporality. That is, the exposure must be documented to have occurred before the observed effect, with sufficient time for any induction period related to the disease.

Table 1. Hill's guidelines for evaluating causation in epidemiologic data*

Coherence	The association should be compatible with existing theory and knowledge.
Experimental evidence	Causation is likely if the disease has been shown to be prevented by the removal of the exposure through an intervention or prevention program.
Analogy	Established causal relationships observed with similar diseases and/or exposures provide more weight for a causal relationship.

*These guidelines were adapted from Hill (1965).

2.3.2 Meta- and pooled analyses

In epidemiologic research, the results of smaller studies are difficult to distinguish from the random variation that normally occurs in data. Meta-analysis is an analytic technique that combines the published results from a group of studies into one summary result. A pooled analysis, on the other hand, combines the raw, individual-level data from the original studies and analyzes the data from the studies together. These methods are valuable because they increase the number of individuals in the analysis, which allows for a statistically more robust and stable estimate of association. Meta- and pooled analyses are also important tools for qualitatively synthesizing the results of a large group of studies and to examine factors that may explain differences between the results of studies analyzed.

The disadvantage of meta- and pooled analyses is that they can convey a false sense of consistency across studies if *only* the combined estimate of effect is considered (Rothman and Greenland, 1998). These analyses typically combine data from studies with different study populations, methods for measuring and defining exposure, and definitions of disease. This is particularly true for analyses that combine data from case-control studies that use very different methods for exposure assessment and the selection of cases and controls. Therefore, in addition to the synthesis or combination of data, meta- and pooled analyses should be used to assess heterogeneity in the results, that is, to understand what factors cause the results of the studies to vary, and how these factors affect the associations calculated from the data of all the studies (Rothman and Greenland, 1998). In addition, the influence of individual studies on the overall results also could be assessed. For example, in a pooled analysis of childhood leukemia and magnetic-field exposure, Greenland et al. (2000) performed analyses to assess how excluding particular studies from the group impacted the overall results. Meta- and pooled analyses are a

valuable technique in epidemiology, but the quality of the underlying studies and the consistency and robustness of the results should always be taken into consideration.

2.3.3 Assessment of EMF exposure in epidemiologic studies

One of the most crucial aspects in the review of any epidemiologic study is an evaluation of how exposure was measured or assessed. A good exposure metric should measure the element that is hypothesized to cause the disease at the etiologically relevant time in the disease process. Estimating exposure to EMF is difficult because 1) EMF is ubiquitous; 2) exposure is often estimated retrospectively; and 3) there is currently no accepted biological mechanism for carcinogenicity or any other disease process, so the appropriate exposure metric and timing is unknown. In the absence of substantive knowledge about a specific mechanism by which magnetic fields could affect normal cells, the focus on long-term exposure is based upon the standard assumption that exposure that affects the development of cancer requires repeated exposure at elevated levels, as does tobacco smoke, alcohol, sunlight, chemicals, and other agents in the environment that are known to cause cancer. Investigators have used different types of magnetic-field assessment methods, including measurements and calculations, to estimate a person's long-term time-weighted average (TWA) exposure. One method of estimating a person's TWA exposure is to sum all magnetic-field exposure encountered during the day (e.g., while at work or school, at home, at a grocery store, shopping, etc.), weight each estimate by the time spent in that environment, and divide that value by the total time of interest.

Historical exposure to residential magnetic fields has been estimated in epidemiologic studies using a variety of surrogates, including:

- Classification of potential magnetic-field exposure from nearby power lines based on the number and thickness of power-line conductors and their distance to nearby residences (wire code categories);
- Simple distance from overhead or underground power lines;
- Instantaneous, spot (short-term) measurements in particular locations of a home;
- Long-term stationary measurements of magnetic fields (typically over 24- or 48-hour periods) in a room where a person spends most of his or her time, or measurements taken by a device that is carried by the person (personal monitoring); and

• Calculated magnetic-field levels based on information on loading, height, configuration, etc., of nearby transmission lines.

In general, long-term exposure using personal magnetic-field measurements are frequently considered as the most appropriate measures, because they estimate exposure from all magneticfield sources and directly estimate a person's total exposure. Personal monitoring results, however, are strongly influenced by behavior and the person's environment, thus, any change in behavior and the environment between the time of measurement and the etiologically relevant time period may still result in exposure misclassification. Also, even long-term measurements typically capture exposure during a 24- or 48-hour period, and may not fully represent average exposure over months or years. Other methods typically capture exposure from one type of source. Personal magnetic-field measurements are obtained by wearing a personal exposure meter, which can take single readings each minute to estimate average magnetic-field exposure over the measurement period. Since this type of measurement may be cost prohibitive in some locations, the investigators of a study of Canadian children evaluated what proxy exposure measures might best predict the child's 48-hour average magnetic-field exposure (Armstrong et al., 2001). Stationary 24-hour measurements in a child's bedroom were a good predictor of 48hour personal exposure, and spot measurements around the perimeter of the child's home were a moderately good predictor. Wire code categories, on the other hand, were not found to be an accurate predictor of a child's exposure (Armstrong et al., 2001).

It is important to note that estimates of magnetic-field exposure in epidemiologic studies represent estimates of long-term exposure potentially from all sources over months or years, and should not be compared to the magnetic-field values measured on a single occasion, and at a single, fixed location. It is evident that brief encounters with higher magnetic fields (for example, walking under a distribution or transmission line, at home in front of a refrigerator or television, or at a grocery store near the freezer) would not significantly alter the long-term exposure of a person to magnetic fields, as reflected in their TWA exposure, because they typically spend a very small fraction of their time at these locations.

Much of the research on EMF is related to occupational exposures, given the higher range of exposure levels encountered in the occupational environment. The main limitation of these studies, however, has been the methods used to assess exposure, with early studies relying

simply on a person's occupational title (often taken from a death certificate) and later studies linking a person's full or partial occupational history to representative average exposures for each occupation (i.e., a job exposure matrix). The latter method, while it represents advancement over earlier methods, still has some important limitations, as noted in the Scientific Committee of Emerging and Newly Identified Health Risk's (SCENIHR's) 2015 review and highlighted in a review by Kheifets et al. (2009) summarizing an expert panel's findings.⁹ While a person's occupation may provide some indication of the overall magnitude of their occupational magnetic-field exposure, it does not take into account the possible variation in exposure due to different job tasks within occupational titles, the frequency and intensity of contact to relevant exposure sources, or variation by calendar time. The 2015 review by SCENIHR noted that the validity of using an indirect indictor of exposure, such as job title, depends on the variability of exposure of people within similar occupations - the more the exposures differ across the same occupation, the higher the likelihood that the exposure level will be incorrectly estimated (SCENIHR, 2015). A study of the 48-hour exposure of 543 workers in Italy found that job exposure matrices were a poor indicator of actual occupational, magnetic-field exposure levels (Gobba et al., 2011). A study by Mee et al. (2009) also confirmed that job exposure matrices could be improved by linking occupational classifications with industry or information on participation in certain tasks of interest (e.g., use of welding equipment or work near power lines) based on their measurements of personal occupational magnetic-field exposures in the United Kingdom.

2.4 Evaluation of experimental research

2.4.1 General research methods

Experimental studies of humans, animals, and cells and tissues complement epidemiologic studies. Both epidemiologic and experimental approaches are needed because, although people are the species of interest, they have large variations in their genetic makeup, exposures, dietary intake, and health-related behaviors that may affect health outcomes. In laboratory animals,

⁹ Kheifets et al. (2009) reports on the conclusions of an independent panel organized by the Energy Networks Association in the United Kingdom in 2006 to review the current status of the science on occupational EMF exposure and identify the highest priority research needs.

these variables can be well controlled to provide more precise information regarding the effects of an exposure. In epidemiologic studies, it is difficult to control for these variables because scientists are merely observing individuals going about their ordinary lives. Taken together, epidemiology, *in vivo*, and *in vitro* studies provide a more complete picture of a possible disease etiology than any one of these study types alone.

A wide variety of approaches is available for assessing the possible adverse effects associated with exposures in experimental studies. The two general types of experimental studies are *in vivo* and *in vitro* studies. *In vivo* studies include studies that examine the potential effects of exposures on human volunteers (usually short-term studies examining short-term effects) and studies of whole animals that could also examine long-term effects. *In vitro* studies are designed to evaluate the way that the exposure may interact with cells and tissues outside of the body, which may provide information on mechanism of action.

2.4.1.1 In vivo studies

Studies in which laboratory animals receive high exposures in a controlled environment provide an important basis for evaluating the safety of environmental, occupational, and drug exposures. These approaches are widely used by health agencies to assess risks to humans from medicines, chemicals, and physical agents (Health Canada, 2000; WHO, 2010; IARC, 2002 preamble; USEPA, 2002; USEPA, 2005). From a public health perspective, long-term (chronic) studies in which animals undergo exposure over most of their lifetime, or during their entire pregnancy, are of high importance in assessing potential risks of cancer and other adverse effects. In these long-term studies, researchers examine a large number of parameters and anatomical sites to assess changes and adverse effects in body organs, cells, and tissues.

These data are used in the hazard identification step of the risk assessment process to determine whether an environmental exposure is likely to produce cancer or damage organs and tissues. Health Canada mandates that lifetime *in vivo* studies or *in vivo* studies of exposures during critical sensitive periods are conducted to assess potential toxicity to humans (Health Canada, 1994). Furthermore, the U.S. Environmental Protection Agency's position is that, "...*the absence of tumors in well-conducted, long-term animal studies in at least two species provides*

reasonable assurance that an agent may not be a carcinogenic concern for humans" (USEPA, 2005, pp. 2-22).

2.4.1.2 In vitro studies

In vitro studies are used to investigate the mechanisms for effects that are observed in living organisms. The relative value of *in vitro* tests to human health risk assessment is less than that of *in vivo* and epidemiologic studies because responses of cells and tissues outside the body may not reflect the response of those same cells if maintained in an intact living system, so their relevance cannot be assumed (IARC, 1992). It may be difficult to extrapolate from simple cellular systems to complex, higher organisms to predict risks to health because the mechanism underlying effects observed *in vitro* may not correspond to the mechanism underlying complex processes like carcinogenesis. In addition, the results of *in vitro* studies cannot be interpreted in terms of potential human health risks unless they are performed in a well-studied and validated test system. For these reasons, the IARC and other agencies treat data from *in vitro* studies as supplementary to data obtained from epidemiologic and *in vivo* studies.

Convincing evidence for a mechanism that explains an effect observed in experimental or epidemiologic studies can add weight to the assessment of cause and effect, and in some cases may clarify reasons for different results among species, or between animals and humans. *In vitro* studies, however, are not used directly by any health agency to assess risks to human health. Therefore, this report focuses on epidemiologic studies and also discusses *in vivo* experimental research with relevance to carcinogenesis but does not address *in vitro* research nor the conclusions of scientific panels regarding *in* vitro data.

2.4.2 Experimental methods for cancer research

Cancer research in the laboratory includes studies of various stages of cancer development. Research has established that cells may take several steps to change from ordinary cells to the uncontrolled growth typical of cancer. Cancer usually begins with a mutation, that is, an irreversible change in the genetic material of the cell, a process also called cancer initiation or cancer induction. Additional steps (also called cancer promotion), must also occur for a cancerous cell to develop into a tumor. A carcinogenic agent may affect either or both the initiation and promotion phases of cancer development. Exposures that affect both initiation and promotion are sometimes called complete carcinogens.

In vitro assays isolate specific cells or microorganisms in glassware in the laboratory to assess the likelihood that exposure to the agent can cause mutations, a step considered necessary in the initiation of cancer. Initiation tests have also been developed in animals, in which scientists expose them for less than lifetime periods to determine whether an exposure caused changes typical for early stage cancers in specific tissues such as liver, breast, or skin.

Other tests are designed to ascertain whether a specific exposure can stimulate tumor growth (i.e., promotion) in an animal in which the cellular changes typical of initiation have already occurred. Studies of promotion typically include two steps: first, exposing the experimental animals to a chemical known to initiate cancer, and second, exposing the animals to the agent to be tested as a promoter. The occurrence of cancer in animals exposed to an initiator and the potential promoter is compared to the occurrence of cancer that develops in animals exposed only to the initiator.

The failure of early EMF research to produce mutations in the DNA of cells *in vitro* was a factor in directing scientists to focus on studies of promotion.

2.4.3 Experimental methods for developmental toxicity

Studies in animals also are used to assess whether an exposure can pose a risk to the unborn children of pregnant women. Experimental studies in pregnant animals provide a means for isolating the exposure in question from the myriad of other factors that can affect prenatal development. The results of these well-controlled *in vivo* studies are used by regulatory agencies to assess prenatal risk and help set human exposure limits (NTP, 2015; USEPA, 1991, 1998).

To test the potential for an exposure to affect fetal development, pregnant mammals such as mice, rats, or rabbits are exposed from the time the embryo is implanted in the uterus to the day before delivery. Variations in study design include preconception exposure of the female in addition to exposure during gestation, and even further exposure after the animal is born. Protocols generally specify that doses be set below the levels known to cause maternal toxicity, that unexposed controls are maintained at the same time period, and that the animals' health is monitored throughout the study. Endpoints measured include maternal body weight and weight change, the numbers and percent of live offspring, fetal body weight, the sex ratio, and external, soft tissue, or skeletal variations and malformations. The uterus can also be examined to assess the number of implantations and fetuses that have been lost, as an indication of miscarriage (USEPA, 1998).

2.4.4 Evaluating the cumulative body of experimental evidence

Key factors in evaluating individual experimental studies for a weight-of-evidence review include the details of the protocol; the plan for selecting animals and conducting and analyzing the study; the adequacy of the dose levels selected; the way in which the study was actually conducted, including adherence to good laboratory practices in animal housing and monitoring; and the evaluation of the effects on toxicity, tumors, or malformations, considering both biological and statistical issues (USEPA, 2005).

As an example of a protocol, consider the long-term *in vivo* study, a major tool for determining whether a chemical can produce cancer in humans. Standard protocols usually specify at least 50 animals of each sex per dose level, in each of three different dose groups. One of these is a high-level dose group termed the maximum tolerated dose, which is close to, but below, the level that increases mortality or produces significant morbidity. Additional dose levels are used below this maximum. An unexposed group, or control, is maintained under the same conditions during the same time period for comparison. This study design permits a separate evaluation of the incidence rate for each tumor type in the exposed group compared to the unexposed control group. Statistical methods are used to assess the role of chance in any differences in the rates between exposed and unexposed, or among the dose groups. If effects are observed in a study, other studies are conducted because similarity of results in different studies, laboratories, and species strengthens the evidence.

Specific methods are used to reduce subjectivity and avoid systematic error, or bias, in scientific experiments (NRC, 1997). These are summarized in Table 2, including the random assignment of subjects to control or comparison groups, the unbiased collection of information (e.g., researchers are not aware of, or are "blind" to the exposure), and the need for replication of results. As with *Hill's criteria*, each guideline for evaluating causation in experimental studies is

not met with a simple "yes" or "no," rather, they serve as guidance for weighing the evidence to reach a decision about cause-and-effect. The more firmly these criteria are met by the studies, the more convincing the evidence.

Avoiding unwanted effects	Experimental techniques should be chosen to avoid effects of intervening factors such as microshocks, noise, corona discharges, vibrations and chemicals.		
Exposure classification	Extreme care should be taken to determine the effective EMF field, voltage, or current in the organism.		
Sensitivity	The sensitivity of the experiments should be adequate to ensure a reasonable probability that an effect would be detected if it existed.		
Objectivity	The experimental and observational techniques, methods and conditions should be objective. "Blind" scoring (where the investigator making the observations is unaware of the experimental variable being tested) should be used whenever there is a possibility of investigator bias. "Double-blind" protocols (where neither the investigator making the observations nor the experimental subject are aware of the experimental variable being tested) should be used in studies of people when the experimental subjects' perceptions may be unwittingly influenced.		
Statistical significance	If an effect is claimed, the result should be demonstrated at a level where chance is an unlikely explanation.		
Consistency	The results of a given experiment should be internally consistent among different ways of analyzing the data, and consistent across studies with respect to the effects of interest.		
Quantifiable results	The results should be quantifiable and replicable. In the absence of independent confirmation, a result should not be viewed as definitive.		
Appropriateness of methodologies	The biological and engineering methodologies should be sound and appropriate for the experiment.		

Table 2. Criteria for evaluating experimenta	I studies as applied to EMF exposures*
--	--

*These criteria were adapted from NRC (1997).

3 Conclusions of weight-of-evidence reviews of EMF and health

Scientists, scientific organizations, and regulatory agencies worldwide use the weight-ofevidence approach to assess potential health risks associated with exposures. These expert groups typically include many scientists with diverse skills and background that reflect the different research approaches required to answer questions about health. Using a weight-ofevidence approach as an analytic framework, each group provides its scientific consensus based on a review of the evidence.

3.1 Weight of evidence reviews by national and international scientific agencies

The following scientific organizations have assembled multidisciplinary panels of scientists to conduct weight-of-evidence reviews and arrive at conclusions about the possible risks associated with ELF EMF (in ascending, chronological order of their most recent publication):¹⁰

- The National Institute for Environmental Health Sciences assembled a 30person Working Group to review the cumulative body of epidemiologic and experimental data and provide conclusions and recommendations to the US government (NIEHS, 1998, 1999).
- The IARC completed a full carcinogenic evaluation of EMF in 2002.
- The Federal-Provincial-Territorial Radiation Protection Committee (FPTRPC), an intergovernmental, Canadian committee assembled to harmonize the standards and practices for radiation protection within federal, provincial, and territorial jurisdictions, conducted a review in 1998 and an update in 2005 (FPTRPC, 1998; FPTRPC, 2005). The FPTRPC most recently

¹⁰ We are aware of other published summaries of the EMF research. With an increase in transmission infrastructure development and the advent of the Internet, the release of reviews and summaries now occurs regularly. This update is restricted to summaries that used a weight-of-evidence approach, and for which a multidisciplinary scientific panel reviewed the epidemiologic and experimental evidence (either in its entirety or since the organization's previous report), and offered conclusions about causality. Other reviews and summaries that did not follow this approach are not addressed because they do not assist in making science-based risk assessments and conclusions. Specifically, the Bioinitiative (BI) Group's report that was posted on the internet is not included in our report, because, among other shortcomings, the BI report is not a comprehensive review of the literature and is not based on the scientific weight-of-evidence method.

released a statement from their Working Group in November 2008 summarizing their opinion on exposure to EMF (FPTRPC, 2008).¹¹

- The National Radiological Protection Board¹² of the United Kingdom issued full evaluations of the research in 1992, 2001, and 2004, with supplemental updates and topic-specific reports published in the interim and subsequent to their last full evaluation in 2004 (NRPB, 1992, 1994a, 1994b, 2001a, 2001b, 2004; HPA, 2006). In a letter addressing a related topic in 2009, the Director of the HPA reiterated their position with regard to ELF EMF and appropriate precautionary measures (HMG, 2009).
- The WHO released a review in June 2007 as part of its International EMF Program to assess the scientific evidence of possible health effects of EMF in the frequency range from 0 to 300 Gigahertz.
- The Health Council of the Netherlands, using other major scientific reviews as a starting point, evaluated recent studies in several periodic reports (HCN, 2001; HCN, 2004; HCN, 2005; HCN, 2007; HCN, 2009a). The HCN also released an advisory letter that addressed the topic of power lines and Alzheimer's disease (HCN, 2009b).
- The European Commission funded the European Health Risk Assessment Network on Electromagnetic Fields Exposure (EFHRAN), a network of experts convened to perform health risk assessments and provide scientifically-based recommendations to the Commission. EFHRAN consulted other major reviews and evaluated epidemiologic and experimental research published after August 2008 to provide an updated health assessment (EFHRAN, 2010, 2012).
- The International Commission on Non-Ionizing Radiation Protection (ICNIRP), the formally recognized organization for providing guidance on standards for non-ionizing radiation exposure for the WHO, published a review of the cumulative body of epidemiologic and experimental data on ELF-EMF in 2003. The ICNIRP released exposure guidelines in 2010 that updated their 1998 exposure guidelines. For both guidelines, they relied heavily on previous reviews of the literature related to long-term exposure,

¹¹ Health Canada refers to the FTPRPC as the authority on issues related to EMF. The FPTRPC established an ELF Working Group to carry out periodic reviews, recommend appropriate actions, and provide position statements that reflect the common opinion of intergovernmental authorities.

¹² The National Radiological Protection Board merged with the Health Protection Agency in April 2005 to form its Radiation Protection Division, and in April 2013, the Health Protection Agency became part of Public Health England.
but provided some relevant conclusions as part of their update process (ICNIRP, 1998, 2010).

- The European Union's Scientific Committee on Emerging and Newly Identified Health Risks (SCENIHR) issued its most recent report to the Health Directorate of the European Commission in 2015 that updated previous conclusions (SSC, 1998; CSTEE, 2001; SCENIHR, 2007, 2009, 2015).
- The Swedish Radiation Protection Authority, using other major scientific reviews as a starting point, evaluated current studies in several annual reports published in 2007 and 2008 (SSI, 2007, 2008). The Swedish Radiation Safety Authority, which superseded the Swedish Radiation Protection Authority in 2008, and has "national collective responsibility within the areas of radiation protection and nuclear safety" including EMF research, continues to publish annual reports (SSM, 2010, 2013, 2014, 2015, 2016, 2018).

The most comprehensive assessment of EMF was conducted by the WHO and published in June 2007; their report updated a previous evaluation of ELF EMF by the IARC in 2002. Exponent's 2007 report focused on the conclusions of WHO (2007) and provided an update by reviewing literature published from December 2005 (the approximate cut-off date for WHO) through September 2007. Exponent's 2010 report reviewed research through January 2010; the 2012 report reviewed the research through March 1, 2012; and the 2017 report reviewed the research through December 31, 2016. The current report will again focus on describing and updating the conclusions of the WHO (2007) report, while noting the other scientific organizations that have published their reviews in the interim.

Overall, the published conclusions of scientific review panels have been consistent. None of the panels concluded that either electric fields or magnetic fields are a known or likely cause of any adverse health effect at the long-term, low exposure levels found in the environment. The only known effects of exposure to EMF are acute or short-term effects (such as nerve and muscle stimulation). Existing guidelines from ICNIRP are set to limit short-term exposure at levels much higher than those encountered in public locations, including publicly-accessible areas near electrical facilities.

Most of the uncertainty and controversy surrounding magnetic-field exposure is related to the research on childhood leukemia. Some epidemiologic studies reported that children with

leukemia were more likely to live closer to power lines, or have higher estimates of magneticfield exposure, compared to children without leukemia; other epidemiologic studies did not report this statistical association. When a number of relevant studies were combined in a single analysis, no association was evident at lower exposure levels, but a weak association was reported between childhood leukemia and estimates of average magnetic-field exposure greater than 3-4 mG (Ahlbom et al., 2000; Greenland et al., 2000). These analyses provide some evidence for an association between magnetic fields and childhood leukemia; however, because of the inherent uncertainty associated with observational epidemiologic studies, the results of these pooled analyses were considered to provide only limited epidemiologic support for a causal relationship; chance, bias and confounding could not be ruled out with reasonable confidence. Further, *in vivo* studies have not found that magnetic fields induce or promote cancer in animals exposed for their entire lifespan under highly-controlled conditions, nor have *in vitro* studies found a cellular mechanism by which magnetic fields could induce carcinogenesis.

Considering all the evidence together, the WHO, as well as other scientific panels, classified magnetic fields as a *possible* cause of childhood leukemia (NRPB, 2001a; IARC, 2002; ICNIRP, 2003; HCN, 2004; WHO, 2007). The term *possible* denotes an exposure for which epidemiologic evidence points to a statistical association, but other explanations cannot be ruled out as the cause of that statistical association (e.g., bias and confounding) and experimental evidence does not support a cause-and-effect relationship (Figure 3).

While much additional research has been published since the WHO evaluation, the main conclusions of scientific organizations remained consistent—the scientific evidence does not establish that exposure to low level ELF EMF is the cause of any cancer (including childhood leukemia) or non-cancer adverse health effects (WHO, 2007; HMG, 2009; EFHRAN, 2012; ICNIRP, 2010; SCENIHR, 2015; SSM, 2018). The WHO and more recent reviews, however, continue to recommend further research to reconcile results from epidemiologic studies on childhood leukemia and the lack of evidence from experimental studies through innovative research. Researchers believe that the development of childhood leukemia, like any other cancer, is influenced by a multitude of different factors, such as genetics, environmental exposures, and infectious agents (see, e.g., Buffler et al., 2005; McNally and Parker, 2006).

Although some questions remain, the epidemiologic evidence does not support a cause-andeffect relationship between magnetic fields and adult leukemia/lymphoma or brain cancer, with the data being described as inadequate or weak (WHO, 2007; EFHRAN, 2012; SCENIHR, 2015; SSM, 2016, 2018). Scientific organizations have concluded that there is strong evidence in support of *no* relationship between magnetic fields and breast cancer or cardiovascular disease (WHO, 2007; SSI, 2008; ICNIRP, 2010; EFHRAN, 2012; SSM, 2016). Although two epidemiologic studies reported a statistical association between peak magnetic-field exposure and miscarriage, a serious bias in how these studies were conducted was identified and various scientific panels concluded that these biases preclude making any conclusions about associations between magnetic-field exposure and miscarriage (HCN, 2004; NRPB, 2004; WHO, 2007; ICNIRP, 2010; SCENIHR, 2015). While an association between some neurodegenerative diseases (i.e., Alzheimer's disease and amyotrophic lateral sclerosis [ALS]) and estimates of higher average occupational magnetic-field exposure has been reported in earlier studies, more recent studies showed mixed results; scientific panels have described this research as weak and inadequate and recommended additional research in this area (WHO, 2007; HCN, 2009b; ICNIRP, 2010; EFHRAN, 2012; SCENIHR, 2015; SSM, 2016).

In summary, reviews published by scientific organizations using weight-of-evidence methods have concluded that the cumulative body of research to date does not support the hypothesis that electric or magnetic fields cause any long-term adverse health effects at the levels we encounter in our everyday environments.

The Working Group of the FPTRPC concluded the following with respect to ELF EMF and health in a statement released in 2008:

In summary, it is the opinion of the Federal-Provincial-Territorial Radiation Protection Committee that there is insufficient scientific evidence showing exposure to EMFs from power lines can cause adverse health effects such as cancer.

The FPTRPC conclusion is consistent with statements by Health Canada on its website:

There is no conclusive evidence of any harm caused by exposures at levels found in Canadian homes and schools, including those located just outside the boundaries of power line corridors.¹³

3.2 Standards and guidelines for limiting exposure to EMF

3.2.1 Status of EMF guidelines

Two international scientific organizations, ICNIRP and the International Committee for Electromagnetic Safety (ICES), have published guidelines for limiting public exposure to EMF (ICES, 2002; ICNIRP, 2010). The health outcomes examined in most EMF epidemiologic and *in vivo* studies primarily have addressed magnetic fields, mainly because structures and vegetation provide some shielding that limits residential exposure to electric fields from power lines; however, these EMF guidelines recommend limits for both electric and magnetic fields.

These guideline limits are set to prevent known and established effects after consideration of the scientific evidence regarding potential effects of both long-term and short-term exposures. Because the only established effects are the short-term direct, acute health effects (i.e., perception, annoyance, and the stimulation of nerves and muscles) that can occur at high levels of exposure, the guidelines are set to protect against these acute effects. With respect to long-term effects, the ICNIRP review concluded the following:

It is the view of ICNIRP that the currently existing scientific evidence that prolonged exposure to low frequency magnetic fields is causally related with an increased risk of childhood leukemia is too weak to form the basis for exposure guidelines. In particular, if the relationship is not causal, then no benefit to health will accrue from reducing exposure. (ICNIRP, 2010; p. 824)

Although ICNIRP and ICES have the same objectives¹⁴ and used similar methods, the recommended limits for exposure of the general

¹³ <u>https://www.canada.ca/en/health-canada/services/home-garden-safety/electric-magnetic-fields-power-lines-electrical-appliances.html;</u> website update on July 6, 2016; accessed on February 20, 2019.

¹⁴ The scope of ICES is the "Development of standards for the safe use of electromagnetic energy in the range of 0 Hz to 300 GHz relative to the hazards of exposure to man ... to such energy." ICES encourages balanced international volunteer participation of the public, the scientific and engineering community, agencies of governments, producers, and users. ICNIRP is an independent group of approximately 40 experts assembled from around the world. It is the formally recognized, non-governmental organization charged with developing

public to EMF at the frequencies used to transmit electricity differ, as seen in Table 3.

Organization recommending limit	Magnetic fields	Electric fields
ICNIRP restriction level (2010)	2,000 mG	4.2 kV/m
ICES maximum permissible exposure (2002)	9,040 mG	5 kV/m 10 kV/m*

Table 3. Reference levels for whole body exposure to 60-Hz fields: general public

*This is an exception within transmission line rights of way because people do not spend a substantial amount of time in rights of way and very specific conditions are needed before a response is likely to occur (i.e., a person must be well insulated from ground and must contact a grounded conductor) (ICES, 2002, p. 27).

ICNIRP recommends screening values for magnetic fields of 2,000 mG for the general public and 10,000 mG for workers (ICNIRP, 2010). The ICES recommends a screening value of 9,040 mG for magnetic-field exposure (ICES, 2002). The ICNIRP screening value for general public exposure to electric fields is 4.2 kV/m, and the ICES screening value for general public exposure to electric fields is 5 kV/m. Both organizations allow higher exposure levels if it can be demonstrated that exposure does not produce current densities or electric fields within tissues that exceed basic restrictions on internal current densities or electric fields.

In Canada, there are no national standards or guidance for limiting residential or occupational exposure to 60-Hz ELF EMF based on either acute or long-term health effects. Rather, the only Canadian standards specify maximum levels and duration of exposure to *radio frequency fields*, that is, fields with a frequency over 3,000 Hz (Health Canada, Safety Code 6, 2015). Health Canada, which monitors the scientific research on EMF and human health as part of its mission to improve the health of Canadians, takes the following position and references the ICNIRP guidelines on its website:

Health Canada does not consider that any precautionary measures are needed regarding daily exposures to EMFs at ELFs. There is no conclusive evidence of any harm caused by exposures at levels found in Canadian homes and schools, including those located just outside the boundaries of power line corridors. ... International exposure guidelines for exposure to EMFs at ELFs have been established by

safety guidance for non-ionizing radiation for the WHO, the International Labour Organization, and the European Union.

the International Commission on Non-Ionizing Radiation Protection (ICNIRP). These guidelines are not based on a consideration of risks related to cancer. Rather, the point of the guidelines is to make sure that exposures to EMFs do not cause electric currents or fields in the body that are stronger than the ones produced naturally by the brain, nerves and heart. EMF exposures in Canadian homes, schools and offices are far below these guidelines.¹⁵

The sections below discuss the similarities and differences between the ICNIRP and ICES standards, and the public health implications of the differences.

3.2.2 Comparison of ICES and ICNIRP guidelines

In both the ICES and ICNIRP standard setting process, a group of scientists conducted extensive reviews of the scientific research regarding health effects. The scientists reviewed the epidemiologic and experimental evidence and concluded that the evidence was insufficient to warrant the development of standards on the basis of hypothesized long-term health effects, such as cancers. Each organization reached a consensus that the most sensitive endpoints—the substantiated adverse effects that would occur at the lowest level of exposure—are short-term reactions to electrostimulation of nerves and muscle. These are direct, acute reactions to high levels of exposure, not severe or life-threatening events.

Each organization developed its recommended exposure limit in two steps. The first step was to identify the lowest level of electrical forces inside the body that is likely to produce the stimulation of nerves and muscle. This internal level, or dose, is further lowered by safety factors to develop what is referred to as the basic restriction. As the term indicates, the basic restriction is the limit for internal dose recommended for exposed populations. This internal dose limit is the foundation of both the ICNIRP and ICES standards because both electric fields and magnetic fields can induce electrical forces in the body.

The ICNIRP and ICES basic restrictions are set well below the value at which an adverse effect was observed in experiments. This is because they incorporate dose reduction factors, also

¹⁵ <u>https://www.canada.ca/en/health-canada/services/home-garden-safety/electric-magnetic-fields-power-lines-electrical-appliances.html;</u> website updated on July 6, 2016; accessed on February 20, 2019.

known as safety factors, to account for potential sources of uncertainty. For example, both groups consider the potentially higher sensitivity in vulnerable groups as a reason for using a safety factor.

The second step in the standard setting process involves developing the reference level. A reference level is developed because a basic restriction, i.e., the biological exposure limit, cannot be directly measured. The reference level is the measurable level of electric field or magnetic field at the location of interest; these levels are outside of the body, and are used as a screening value to ensure that the internal level identified as the basic restriction in not exceeded under any circumstance. Nevertheless, these reference levels represent conservative limits, meaning that if the reference level (i.e., the screening level) is exceeded, it does not necessarily follow that the basic restriction is exceeded. As ICNIRP explains, "In many practical exposure situations external power frequency electric fields at the reference levels will induce current densities in central nervous tissues that are well below the basic restrictions. Recent dosimetry calculations indicate that the reference levels for power-frequency magnetic fields are conservative guidelines relative to meeting the basic restrictions on current density for both public and occupational exposures" (ICNIRP, 1998).

3.2.3 Implications for human health

The underlying question for people who make decisions about public health and safety is whether the ICNIRP reference value (4.2 kV/m) implies greater safety simply because it is lower and includes a larger safety factor. In developing public health standards, safety factors are used when uncertainty is recognized, and the general rule is that smaller safety factors are needed as the relevant information on risk to humans is improved. Although ICNIRP uses a larger safety factor, it applies that safety factor to a higher estimated threshold level. ICES uses a smaller safety factor, but has used highly specific data on human responses, leading to a lower, presumably more precise, estimated threshold level. It is essential to understand that for effects where thresholds are identified, the goal of the standard setting process is to set the exposure limit where no effects will occur in the population. Therefore, further lowering of the exposure limit is not expected to have any additional health benefit. For additional perspective on the question of the safety of exceeding ICNRIP exposure limits up to the level of the ICES limits,

consider that ICNIRP states that EMF guidelines are conservative, and that the ICNIRP recommended limit for occupational exposure is considerably higher at 8.3 kV/m than the 5 kV/m for the general public (ICNIRP, 1998, 2010). Clearly, ICNIRP meant neither workers nor the public to be harmed by electric fields.

3.3 Precautionary approaches

3.3.1 General definition

A precautionary policy for risk management of possible, but unproven, adverse effects emerged in Europe in the 1970s regarding environmental issues. The precautionary principle refers to the idea that, when evidence does not support the suggestion that an exposure is a cause of a particular disease but where a risk is perceived or uncertainty exists, precautionary measures may be taken that are proportional to the perceived level of risk, with science as the basis for estimating that risk. A key element of precautionary approaches is the recognition that a real risk from the exposure may not exist, and its necessary corollary is that the reduction of exposure may not decrease any unconfirmed adverse effects in the population.

The European Commission prepared a report in 2000 to clarify the precautionary principle because this idea had been subject to controversy and variability in interpretation.¹⁶ Their report explained that the implementation of the precautionary principle should be science based, starting with a complete scientific evaluation, and the range of actions taken should depend on the extent of the risk and the degree of uncertainty surrounding the occurrence of adverse effects. They provided guidelines for the application of the precautionary principle or other risk management measures specifying five general principles: proportionality, non-discrimination, consistency, examination of costs and benefits of actions, and examination of scientific developments.¹⁷

¹⁶ Commission of the European Communities, Communication on the Precautionary Principle, Brussels 03 February 2000 (<u>http://europa.eu.int/comm./off/com/health_consumer/precaution.htm</u>).

¹⁷ Proportionality: "Measures...must not be disproportionate to the desired level of protection and must not aim at zero risk."

Nondiscrimination: "comparable situations should not be treated differently and... different situations should not be treated in the same way, unless there are objective grounds for doing so."

A variant of the precautionary principle called prudent avoidance has been favored as a policy option for EMF by some national and local governments. The WHO describes this as "*using simple, easily achievable, low to modest (prudent) cost measures to reduce individual or public EMF exposure, even in the absence of certainty that the measure would reduce risk*" (WHO, 2002).

3.3.2 WHO recommendations regarding precautionary measures for EMF

The scientific evaluation completed by the WHO also discusses general policy strategies for risk management, and provides a summary table of different policy strategies employed worldwide specifically for EMF exposure in the general public (WHO, 2007, Chapter 13). The WHO recommended the following precautionary measures (WHO, 2007, adapted from pp. 372-373):

- Countries are encouraged to adopt international science-based guidelines.
- Provided that the health, social, and economic benefits of electric power are not compromised, implementing very low-cost precautionary procedures to reduce exposures is reasonable and warranted.
- Policy-makers and community planners should implement very low-cost measures when constructing new facilities and designing new equipment including appliances.
- Changes to engineering practice to reduce ELF exposure from equipment or devices should be considered, provided that they yield other additional benefits, such as greater safety or involve little or no cost.

Consistency: "measures...should be comparable in nature and scope with measures already taken in equivalent areas in which all the scientific data are available."

Examination of the benefits and costs of action or lack of action: "This examination should include an economic cost/benefit analysis when this is appropriate and feasible. However, other analysis methods...may also be relevant."

Examination of scientific developments: "The measures must be of a provisional nature pending the availability of more reliable scientific data"... "Scientific research shall be continued with a view to obtaining more complete data."

- When changes to existing ELF sources are contemplated, ELF field reductions should be considered alongside safety, reliability, and economic aspects.
- Local authorities should enforce wiring regulations to reduce unintentional ground currents when building new or rewiring existing facilities, while maintaining safety. Proactive measures to identify violations or existing problems in wiring would be expensive and unlikely to be justified.
- National authorities should implement an effective and open communication strategy to enable informed decision-making by all stakeholders; this should include information on how individuals can reduce their own exposure.
- Local authorities should improve planning of ELF EMF-emitting facilities, including better consultation between industry, local government, and citizens when siting major ELF EMF-emitting sources.
- Government and industry should promote research programs to reduce the uncertainty of the scientific evidence on the health effects of ELF field exposure.

In summary, the general recommendation of the WHO is as follows:

Countries are encouraged to adopt international science-based guidelines. In the case of EMF, the international harmonization of standard setting is a goal that countries should aim for (WHO, 2006). If precautionary measures are considered to complement the standards, they should be applied in such a way that they do not undermine the science-based guidelines (WHO, 2007, p. 367).

3.3.3 Canadian perspective on precautionary approaches

The Government of Canada has published "A Framework for the Application of Precaution in Science-based Decision Making About Risk" (2003). One of the basic general principles is that sound scientific information must be the basis for both deciding whether or not to implement precautionary measures and determining what precautionary measures, if any, are implemented. The document clarifies that "Scientific advisors should give weight to peer-reviewed science and aim at sound and reasonable evidence on which to base their judgments" (p. 8).

The FPTRPC stated the following with respect to precautionary measures in 2008: "In the context of power-frequency EMFs, health risks to the public from such exposures have not been established; therefore, it is the opinion of the FPTRPC that any precautionary measures applied to power lines should favour low cost or no cost options."

Health Canada recommended no precautionary measures to the public in a statement updated in 2016:

Health Canada does not consider that any precautionary measures are needed regarding daily exposures to EMFs at ELFs. There is no conclusive evidence of any harm caused by exposures at levels found in Canadian homes and schools, including those located just outside the boundaries of power line corridors.¹⁸

A framework for applying the precautionary principle to public health issues in Canada has been proposed by four Canadian public health physicians that closely matches the conceptual approach recommended by the European Commission and the approach of the FPTRPC and Health Canada in addressing EMF health concerns (Weir et al., 2010).

¹⁸ <u>https://www.canada.ca/en/health-canada/services/home-garden-safety/electric-magnetic-fields-power-lines-electrical-appliances.html</u>; website update on July 6, 2016; accessed on February 15, 2019.

4 Human Health Research

This section provides a summary and assessment of the literature published up to December 31, 2018, to determine whether recent findings are consistent with the conclusions of the scientific panels reviewed in Section 3, particularly the conclusions of the WHO's evaluation. In several previous reports, Exponent reviewed the literature through December 31, 2016 (Exponent, 2007, 2010, 2012, 2017).

This assessment reviews literature indexed in Pub-Med between January 1, 2017, and December 31, 2018. In carrying out this update, we considered the totality of the science (not just the new information) to determine if changes in the national and international health risk assessments were warranted. This assessment uses a weight-of-evidence approach with standard epidemiologic principles and *Hill's criteria* as an analytic foundation. All relevant research discussed below is taken into consideration and more weight is assigned to studies that are well-designed and well-conducted, because studies with better methods provide stronger evidence. Therefore, this assessment reflects the current knowledge of research related to EMF and the health concerns reviewed.

As noted by the ICNIRP and IARC, there has been no consistent or strong evidence to explain how EMF exposure could affect biological processes in cells and tissues. In addition, such data are supplementary to epidemiologic and *in vivo* studies, and are used rarely by health agencies to directly identify hazards to human health. For that reason, this review systematically addresses epidemiologic studies of various health concerns and *in vivo* studies relevant for carcinogenesis,

A structured literature review was conducted to identify new epidemiologic and *in vivo* peerreviewed research published on 50- or 60-Hz alternating current (AC) ELF EMF between January 1, 2017, and December 31, 2018. A large number of search strings referencing the exposure and health outcomes of interest, as well as authors that regularly publish in this area, were included as search terms in the PubMed database.¹⁹ This report focuses on the health outcomes that have received the most attention—cancer, reproductive or developmental effects, and neurodegenerative diseases – as well as electrical hypersensitivity. To be included in this review, epidemiologic studies on these health outcomes must have assessed EMF exposure beyond a self-reported job title.²⁰ Many other health effects have been studied (suicide, depression, cardiovascular effects, effects on the immune system, etc.), but for brevity and because research on these topics have evolved slowly, they are not summarized here. We note, however, that for these outcomes no substantive evidence has been identified by any previous comprehensive reviews. The possible effects of EMF on implantable cardiac devices and fauna/flora are discussed separately in Sections 5 and 6, respectively. The WHO report continues to remain a good resource for the status of research on these other areas of health research (WHO, 2007).

4.1 Cancer

4.1.1 Childhood leukemia

Since the late 1970s, numerous epidemiologic studies have evaluated the relationship between childhood leukemia and some proxy of magnetic-field exposure. When independently evaluated, the studies showed mixed and varying results. Some of the largest and most advanced studies did not show a clear relationship between magnetic-field exposure and leukemia (Linet et al., 1997; UKCCS, 2000; McBride et al., 1999). When two, independent pooled analyses combined the data from several of these studies, however, results showed an approximate two-fold statistically significant association between average magnetic-field exposure above 3-4 mG and childhood leukemia (Ahlbom et al., 2000; Greenland et al., 2000). This result means that the children with leukemia in these studies were about two times more likely to have had average

¹⁹ PubMed is a service of the U.S. National Library of Medicine that includes over 26 million citations from MEDLINE and other life science journals for biomedical articles back to the 1950s. PubMed includes links to full text articles and other related resources (<u>http://www.ncbi.nlm.nih.gov/PubMed/</u>).

Studies that only report associations between the health outcome under investigation and job titles that are presumed to have high levels of magnetic-field exposure were identified and scanned, but are not evaluated further in this report for several reasons. First, job titles are a crude method of estimating exposure because they do not capture the variety of a person's occupational history or the variety of exposures a person may encounter within one occupation. Furthermore, hypothesis-generating case-control analyses that calculate associations for many occupations are subject to the bias associated with multiple comparisons. These studies provide relatively little information in a weight-of-evidence review, particularly when studies are available with more thorough exposure evaluations (as is the case for the large number of studies related to magnetic-field exposures).

magnetic-field exposure above 3-4 mG, compared to the children in the control group. Average exposure at this level is rare; several surveys show that approximately 0.5-7 percent of children have time-averaged exposures in excess of 3 mG, and 0.4-3.3 percent have time-averaged exposures in excess of 4 mG (WHO, 2007).²¹ Because of the rarity of exposure to magnetic fields in the 3-4 mG range, analyses have suggested that a small proportion of childhood leukemia cases would be attributed to magnetic fields, if a true relationship existed (Greenland and Kheifets, 2006; Kheifets et al., 2006). It also means that it is difficult to draw conclusions about a very small number of childhood leukemia. A more recent pooled analysis that combined data from studies published between 2000 and 2010 (Kheifets et al., 2010a) reported results comparable to those reported by the earlier pooled analyses (Ahlbom et al., 2000; Greenland et al., 2000), but the reported association in the highest exposure category was weaker than reported previously and no longer statistically significant based on the more recent studies. The studies included in the more recent pooled analysis had limitations similar to those included in the earlier analyses.

The epidemiologic studies of childhood leukemia and EMF were limited in many ways, such that chance, bias, and confounding could not be ruled out as explanations for the association, which has been the overall conclusion of the reviews issued by the IARC (2002), WHO (2007), SCENIHR (2015), and other agencies. Thus, it was unclear whether exposure to magnetic fields in the range of 3-4 mG had any relationship with the development of childhood leukemia or whether the association was simply a consequence of chance, due to small numbers of cases with magnetic field exposure above background levels, or some error in some aspects of study design, conduct, or analysis. In addition, experimental studies did not suggest that magnetic fields are carcinogenic—these studies did not indicate any consistent increase in cancer in animals when they were exposed to high levels of magnetic fields over the course of their lifetime (*see "In vivo* studies of carcinogenesis" below), and there was no known mechanism by which magnetic fields cause cancer.

²¹ The failure to understand the difference between calculated or measured spot values of the magnetic field and estimates of long-term average magnetic-field exposure above 4 mG has been discussed by Bailey and Wagner (2008) as noted by Health Canada (2010).

A number of large case-control studies on EMF and childhood leukemia have been published in recent years from several European countries and from the United States (Sermage-Faure et al., 2013; Bunch et al., 2014, 2015, 2016; Pedersen et al., 2014a, 2014b, 2015; Crespi et al., 2016). These studies, which incorporated large sample sizes and methodological advancements, showed no statistically significant associations between estimates of residential exposure to EMF and childhood leukemia. While these studies provided no further support for an association and somewhat weaken the overall evidence, the previously reported associations remain largely unexplained and the overall conclusion on the epidemiologic data remains that it provides limited evidence for an association. This conclusion also expressed by the most recent reviews by scientific organizations (e.g., SCENIHR, 2015; SSM, 2018).

Relevant studies 2017-2018

Several recent publications have presented further analyses on the same study populations used in two of the case-control studies (Bunch et al., 2014; Crespi et al., 2016) summarized in the previous Exponent report (Exponent, 2017). Swanson and Bunch (2018) reanalyzed the same study population used in Bunch et al. (2014)'s study of childhood leukemia in relation to residential proximity to high-voltage power lines in the United Kingdom. In their re-analysis, Swanson and Bunch (2018) used finer categories of distance from power lines (e.g., cut-points of every 50-meter distance) and broader groupings of diagnosis date (e.g., 1960-1979, 1980-1999, and 2000-on) to match the categories used in a recent pooled analysis of childhood leukemia and distance to power lines (Amoon et al., 2018a, summarized below). Swanson and Bunch (2018) reported that no overall associations between distance categories and childhood leukemia were observed for the period including 1980 and later, and that associations for the earlier period (1960-1979) showed no monotonic or consistent pattern with distance. Thus, the authors concluded that their finding "*weakens the evidence that any elevated risks are related to magnetic fields*" (Swanson and Bunch, 2018, p. N30).

Kheifets et al. (2017a) and Amoon et al. (2018b) both conducted additional analyses using the same California Power Line Study (CAPS) population as Crespi et al. (2016). Kheifets et al. (2017a) reported on childhood leukemia and calculated magnetic fields from power lines. Magnetic-field levels at birth address were calculated using geographic information systems,

aerial imagery, historical information on load and phasing, and site visits (Vergara et al., 2015). In the main analyses by Kheifets et al. (2017a), which included 5,788 cases of leukemia and a similar number of primary controls with geocode accuracy, the authors reported no consistent pattern of association. Compared to the lowest exposure category (<1 mG), they reported a slight, statistically non-significant, negative association in the intermediate exposure categories (1 to 2 mG and 2 to 4 mG) and a small, statistically non-significant, positive association in the highest exposure category (\geq 4 mG). The authors reported similar results in subgroup and sensitivity analyses and commented that all estimates had wide confidence intervals. The authors concluded that their study "does not in itself provide clear evidence for risk associated with greater exposure to magnetic fields from power lines, but could be viewed as consistent with previous findings of increased risk" (Kheifets et al., 2017a, p. 1117). Thomas (2018) commented that while the Kheifets et al. (2017a) study had low potential for selection bias due to its record-based methods, the study may be subject to exposure misclassification resulting in bias towards the null because the exposure assessment considered residential proximity only to high-voltage power lines and ignored other sources, including distribution lines.

Amoon et al. (2018b) assessed the potential impact of residential mobility of the study subjects (i.e., moving residences between birth and diagnosis) on the associations reported in Crespi et al. (2016) and Kheifets et al. (2017a). The authors reported that while children that moved tended to be older, lived in housing other than a single-family home, had younger mothers and fewer siblings, and were of lower socioeconomic status, changing residences was not associated with either calculated fields or proximity to \geq 200-kV power lines. Thus, the authors concluded that "[m]obility appears to be an unlikely explanation for the associations observed between power lines [sic] exposure and childhood leukemia" in the previous California studies (Amoon et al., 2018b, p. 459).

A case-control study in Greece examined the association between parental occupational exposures and childhood acute leukemia at a major pediatric hospital in Athens (Kyriakopoulou et al., 2018). The study included 108 cases of acute leukemia, including lymphoblastic leukemia (ALL) and acute myeloid leukemia (AML), under the age of 15, and 108 controls matched on age, gender, and ethnicity. The parents' job titles held during four different exposure periods (1

year before conception, during pregnancy, during breastfeeding, and from birth until diagnosis) were evaluated for exposure (exposed versus unexposed) to four agents (high contact level, chemicals, electromagnetic fields, and ionizing radiation) based on the authors' review of literature and their professional judgment. A total of six cases (5.6%) and six controls (5.6%) were categorized as exposed to electromagnetic²² fields. Therefore, no statistically significant association was observed between magnetic-field exposure and childhood acute leukemia for any of the four periods of exposure. No associations were observed between childhood acute leukemia and the remaining exposure categories; however, the authors did observe that high birth weight and family history of cancer were associated with children who developed leukemia.

A meta-analysis of epidemiologic studies of residential distance to power lines and childhood leukemia was conducted by Amoon et al. (2018a). The authors pooled the data from 11 studies with record-based assessments of residential distance from high-voltage power lines from 10 countries (Australia, Brazil, Denmark, France, Italy, Norway, Sweden, Switzerland, the United Kingdom, and the United States). In total, 29,049 cases and 68,231 controls were included in the analyses. The authors reported no association when proximity to transmission lines with any voltage was investigated; the adjusted OR for residential distance <50 meters, as compared to distances \geq 300 meters, was 1.01 [95% CI, 0.85-1.21]). For power lines with voltages of \geq 200 kV, the adjusted OR (1.13) for distances <50 meters also was not statistically significant (95% CI, 0.92-1.93). The reported associations were slightly stronger for leukemia case diagnosed before 5 years of age and in study periods prior to 1980. Adjustment for various potential confounders (e.g., socioeconomic status, dwelling type, residential mobility) had little effect on the estimated associations.

Magne et al. (2017) conducted a national survey of ELF magnetic-field exposure in France, including a representative sample of close to 1,000 children 0 to 14 years of age. The study was purely an exposure assessment study and the authors did not investigate any health outcome in relation to magnetic-field exposure. The authors conducted 24-hour measurements of ELF magnetic-fields of the included children and reported that 3.1% of the study participants had a

²² In this context, "electromagnetic fields" is used to refer to ELF magnetic fields.

24-hour average exposure >0.4 μ T (4 mG). Only 0.8% of the children, however, had 24-hour average exposure >0.4 μ T (>4 mG) when exposure from alarm clocks was excluded. The authors also reported that none of the children with 24-hour average exposure >0.4 μ T (>4 mG) lived within 125 meters of a 225-kV transmission line or within 200 meters of a 400-kV transmission line.

Kheifets et al. (2017b) conducted a comparative analysis of epidemiologic studies of childhood leukemia that investigated the association between childhood leukemia and ionizing radiation (i.e., radon or gamma radiation) or non-ionizing radiation (i.e., ELF EMF), or both, in an attempt to evaluate to what extent bias, confounding, and other methodological issues might be responsible for the reported associations. The authors reported that while they found some indication of bias, they found little evidence that confounding has a substantial influence on results.

In summary, the results of recent studies do not change the classification of the epidemiologic data as limited. While most of the recently published studies showed no statistically significant associations between estimates of exposures from power lines, and the most recent pooled analyses indicated weaker and statistically non-significant associations (Amoon et al., 2018a), the association between childhood leukemia and magnetic fields observed in some earlier studies remains unexplained.

Authors	Year	Study
Amoon et al.	2018a	Proximity to overhead power lines and childhood leukaemia: an international pooled analysis.
Amoon et al.	2018b	Residential mobility and childhood leukemia.
Kheifets et al.	2017a	Residential magnetic fields exposure and childhood leukemia: a population- based case-control study in California.
Kheifets et al.	2017b	Comparative analyses of studies of childhood leukemia and magnetic fields, radon and gamma radiation.
Kyriakopoulou et al.	2018	Parental occupational exposures and risk of childhood acute leukemia.
Magne et al.	2017	Exposure of children to extremely low frequency magnetic fields in France: Results of the EXPERS study.
Swanson and Bunch	2018	Reanalysis of risks of childhood leukaemia with distance from overhead power lines in the UK.

Table 4.	Studies of childhoo	d leukemia (2017-2018)
----------	---------------------	------------------------

Attachment 1

March 4, 2019

Authors	Year	Study
Comment on Kheifets et	al. (2017a)	
Thomas	2018	Re: Kheifets et al. (2017): Residential magnetic fields exposure and childhood leukemia: a population-based case-control study in California.

4.1.2 Childhood brain cancer

Compared to the research on magnetic fields and childhood leukemia, there have been fewer studies of childhood brain cancer and the evidence linking magnetic fields to childhood brain cancer is considerably weaker. No consistent association has been found, although studies are limited by the small number of participants (since childhood brain cancer is rare). To address the issue of small numbers, both a meta-analysis (Mezei et al., 2008) and a pooled analysis (Kheifets et al., 2010b) were conducted. Overall, no association was reported in the meta-analysis, but a sub-analysis of five studies with information on calculated or measured magnetic fields greater than 3-4 mG found a combined OR that was elevated but not statistically significant (OR=1.68, 95% CI=0.83-3.43) (Mezei et al., 2008). The authors stated that an increased risk of childhood brain tumors could not be excluded at this high exposure level, but that the similarity of this result to the findings of the pooled analyses.

Kheifets et al. (2010b), in direct response to a recommendation by the WHO, pooled data from 10 studies on childhood brain cancer and residential magnetic-field exposure. Similar to the pooled analysis of childhood leukemia (Kheifets et al., 2010a), there were few cases in the upper exposure categories; however, contrary to the childhood leukemia results, no consistent associations were reported for childhood brain cancer (Kheifets et al., 2010b). While some elevated ORs were observed, they were not statistically significant and no dose-response patterns were observed. The authors concluded that their results provide little evidence for an association between magnetic fields and childhood brain cancer.

Subsequent studies have not reported any consistent associations between EMF and childhood brain tumors and thus did not warrant any change in the classification of the epidemiologic evidence as *inadequate*.

Relevant studies 2017-2018

Only one major study has been published on childhood brain cancer since the previous Exponent report (Exponent, 2017). Su et al. (2018) conducted a meta-analysis of epidemiologic studies that investigated the association between parental occupational exposure to ELF magnetic fields and childhood nervous system tumors. The authors included a total of 22 studies (of which 21 were case-control studies) published as of December 2017 in their analysis. For central nervous system (CNS) tumors, they reported no statistically significant association for paternal exposure to ELF magnetic fields but reported a weak statistically significant association (OR = 1.16, 95% CI 1.06, 1.26) for maternal exposure based on a subset of eight studies. For neuroblastoma, the authors reported no association with either maternal or paternal exposure to ELF magnetic fields. The authors also reported the results of several subgroup analyses. Study quality, as assessed by the authors, had inconsistent effects on the associations reported for maternal and paternal exposure. The authors noted that, when based on higher quality studies, observed associations were stronger for maternal exposure but weaker for paternal exposure. In addition, it is noteworthy that associations for CNS tumors were statistically significant only when studies using non-quantitative exposure assessment methods (i.e., relying on job titles only) were pooled; however, no associations were reported based on studies with a quantitative exposure assessment. Limitations of the analysis include the fact that the majority of included studies did not classify the histological subtype of CNS tumor and that several of the studies included persons with tumors diagnosed up to 30 years of age. The authors also reported evidence for publication bias.

In summary, the weight-of-evidence continues to not support an association between magneticfield exposures and the development of childhood brain cancer.

Table 5.	Studies of childhood brain cancer (2017-2018)	
----------	---	--

Authors	Year	Study
Su et al.	2018	Association between parental occupational exposure to extremely low frequency magnetic fields and childhood nervous system tumors risk: A meta-analysis.

4.1.3 Breast cancer

Early studies conducted on breast cancer and electric blanket use and residential and occupational magnetic-field exposure reported inconsistent findings. More recent studies, published around and after 2000, however, tended to be methodologically more advanced and, overall, reported no consistent associations. The WHO, in its 2007 report, concluded that the body of research they reviewed was higher in quality compared with the early studies, and that there was strong support for consensus statements that magnetic-field exposure does not influence the risk of breast cancer.²³ The WHO recommended no further research with respect to breast cancer and magnetic-field exposure, even though the epidemiologic evidence was still classified as *inadequate*.

More recently published large epidemiologic studies have overall reported no consistent or statistically significant associations between EMF and breast cancer among either men or women. Recently published reviews also conclude that the evidence does not suggest a risk (SCENIHR, 2015; SSM, 2016, 2018). The most recently published annual reports by the Swedish Scientific Council on EMF and Health concluded that, with respect to female breast cancer, "*now it is fairly certain that there is no causal relation with exposure to ELF magnetic fields*" (SSM, 2016; p. 7), and with respect to male breast cancer, "*[t]o date, there is no established link between ELF-MF exposure and breast cancer in men.*" (SSM, 2018; p. 49).

²³ The WHO concluded, "Subsequent to the IARC monograph a number of reports have been published concerning the risk of female breast cancer in adults associated with ELF magnetic field exposure. These studies are larger than the previous ones and less susceptible to bias, and overall are negative. With these studies, the evidence for an association between ELF exposure and the risk of breast cancer is weakened considerably and does not support an association of this kind" (WHO 2007, p. 307).

Relevant studies 2017-2018

There have been no new epidemiology studies published on breast cancer and EMF exposure since Exponent's previous report (Exponent, 2017). Therefore, the previous assessment that EMF is not causally linked to breast cancer remains appropriate.

4.1.4 Other adult cancers

In general, scientific panels have concluded that the scientific evidence is inadequate to establish a causal link between other adult cancers and exposure to magnetic fields, but due to the inherent nature of scientific research, the possibility cannot be entirely ruled out (IARC, 2002; WHO 2007). Most epidemiologic studies of EMF and adult cancers (in addition to breast cancer) examined leukemia, lymphoma, and brain cancer, and studies of these outcomes will be discussed in detail below. Fewer studies have been conducted on adult cancers other than leukemia, lymphoma, and cancers of the brain and breast , and no consistently replicated findings were identified by any of the expert panels. Since studies with improved exposure assessment methods do not report stronger or more consistent findings, scientific panels concluded that the evidence for an association is weak and the observed inconsistency is probably due to chance or bias. The IARC classified the epidemiologic data with regard to adult leukemia, lymphoma, and brain cancer as "inadequate" in 2002, and the WHO confirmed this classification in 2007, with the remaining uncertainty attributed mainly to limitations in exposure assessment methods.

Much of the research on EMF and adult cancers is related to occupational exposure, given the higher range of exposure levels encountered in the occupational environment. The main limitations of these studies, however, are the methods used to assess exposure, with early studies relying simply on a person's occupational title (often taken from a death certificate) and later studies linking a person's full or partial occupational history to representative average exposure for each occupation (i.e., a job-exposure matrix). The latter method, while representing a methodological advancement, still has some important limitations as highlighted by Kheifets et al. (2009). While a person's occupation may provide some indication of the overall magnitude of their occupational magnetic-field exposure, it does not take into account the possible variation in exposure due to different job tasks within occupational titles, the frequency and

intensity of contact to relevant exposure sources, or variation by calendar time. Furthermore, since scientists do not know any mechanism by which magnetic fields could lead to cancer, an appropriate exposure metric is also unknown.

4.1.4.1 Adult brain cancer

Epidemiologic studies on EMF and adult brain cancer published following the WHO assessment provided no additional consistent evidence for an association (e.g., Johansen et al., 2007; Coble et al., 2009; Baldi et al., 2011; Marcilio et al., 2011). In a meta-analysis of occupational EMF exposure and leukemia and brain cancer (Khefeits et al., 2008), a small but statistically significant increase of leukemia and brain cancer was reported in relation to the highest estimate of magnetic-field exposure in the individual studies. Several findings, however, led the authors to conclude that magnetic-field exposure is not responsible for the observed associations. For example, the authors reported a weaker association in the more recent studies than the observed association in their previous meta-analysis (Kheifets et al., 1995), whereas a stronger association would be expected if there were a true relationship since the quality of the studies has improved over time. The authors concluded that "the lack of a clear pattern of EMF exposure and outcome risk does not support a hypothesis that these exposures are responsible for the observed excess risk" (Kheifets et al., 1995, p. 677).

Overall, the epidemiologic studies that have been reviewed in our previous reports predominantly support no association between adult brain cancer and residential or occupational EMF exposure. However, an association could not be ruled out *entirely* due to limitations in exposure assessment methods and insufficient data available on specific brain cancer subtypes.

Relevant studies 2017-2018

Carlberg et al. (2017, 2018) published the results of two case-control epidemiologic studies of occupational exposure to ELF EMF and brain cancer. Both studies relied on data from previously published case-control studies in Sweden (Hardell et al., 2006, 2013; Carlberg et al., 2013, 2015). Carlberg et al. (2017) included 1,346 living glioma cases diagnosed between the periods of 1997 to 2003 and 2007 to 2009 and 3,485 controls, ascertained from the Swedish Population Registry, who were matched to cases on sex and 5-year age group. Average and

cumulative occupational exposure to ELF EMF was assessed from self-reported questionnaires on lifetime occupational history and a previously developed job-exposure matrix (Turner et al., 2014). Overall, the authors observed no associations with cumulative exposure to ELF EMF and a weak statistical association in the highest average exposure category (>0.27 μ T [2.7 mG]). Statistically signification associations were reported for grade IV astrocytoma and cumulative and average exposure when restricted to exposure experienced during the more recent exposure periods (1 to 14 years prior to diagnosis). The authors reported no association, however, with more distant exposure periods (15 to 20+ years), and observed no associations for other tumor grades. The authors hypothesized that the observed association for grade IV astrocytoma in the recent exposure periods may be the result of a potential late effect on cancer promotion. Because deceased subjects were excluded from the analyses, and the reported association was limited to tumors of the highest grade (with the highest mortality rate), there is a high likelihood that the reported pattern of results arose due to differential exclusion of rapidly fatal cases among patients with the highest-grade tumors.

Carlberg et al. (2018) included 1,592 meningioma cases and 3,485 controls. The investigators used a similar approach and methods as in the glioma study (Carlberg et al., 2017). The authors reported no trend or association between meningioma development and any of the investigated metrics of occupational exposure to ELF EMF (i.e., average occupational exposure, highest exposed job, or cumulative exposure) regardless of the time windows investigated (i.e., exposure during 1 to 14 years prior to diagnosis, or exposure more than 15 years prior to diagnosis). The authors concluded that occupational ELF-EMF was not associated with an increased risk for meningioma (Carlberg et al., 2018).

Turner et al. (2017) investigated the potential interaction between occupational exposure to ELF magnetic fields and 29 chemical agents – including cadmium, chromium, iron, nickel, solvents, benzo(a)pyrene, polycyclic aromatic hydrocarbons, and environmental tobacco smoke – on brain cancer development within the INTEROCC case-control study. The study included 1,939 glioma and 1,822 meningioma cases, along with 5,404 controls, matched on sex and age. Occupational exposure to both ELF magnetic fields and the chemicals of interest were assessed using job-exposure matrices. The authors reported that there was "*no clear evidence*" for an

interaction between occupational exposure to ELF magnetic fields and occupational exposure to any of the included chemicals for either glioma or meningioma (p. 802).

In summary, recent studies do not provide support for an association between exposure to magnetic fields and brain cancer development. The data remains inadequate (EFHRAN, 2012; SCENIHR, 2015).

Authors	Year	Study
Carlberg et al.	2017	Case-control study on occupational exposure to extremely low-frequency electromagnetic fields and glioma risk.
Carlberg et al.	2018	Case-control study on occupational exposure to extremely low-frequency electromagnetic fields and the association with meningioma.
Turner et al.	2017	Occupational exposure to extremely low frequency magnetic fields and brain tumor risks in the INTEROCC study.

 Table 6.
 Studies of adult brain cancer (2017-2018)

4.1.4.2 Adult leukemia and lymphoma

Similar to adult brain cancer, the WHO classified the epidemiologic evidence with regard to adult leukemia as *"inadequate"* (WHO 2007). Epidemiologic studies of adult leukemia and lymphoma published subsequent to the WHO report (2007) provided no consistent support for an association (e.g., Johansen et al., 2007; Wong et al., 2010; Marcilio et al., 2011). As described above, a small and statistically significant increase of leukemia in relation to the highest estimate of magnetic-field exposure reported in the meta-analysis by Kheifets et al. (2008), but the authors concluded that the overall pattern of results (e.g., there was no consistency in findings by leukemia subtype) did not support a causal relationship between EMF and leukemia.

More recent studies did not provide substantive new evidence in support of an association between EMF exposure and leukemia and lymphoma in adults. Because of scientific uncertain due to study limitations, the scientific body of studies provided inadequate evidence for an association (EFHRAN, 2012; SCENIHR, 2015).

Relevant studies 2017-2018

One large study has been published on adult leukemia and lymphoma since the previous Exponent report (Exponent, 2017). Huss et al. (2018a) conducted a census-based, retrospective cohort study examining exposure to ELF magnetic fields and death from several types of hematopoietic malignancies within the Swiss National Cohort. The study included a total of 3.1 million economically active individuals between 30 and 65 years of age (for men) or 62 years of age (for women) who participated in the Swiss 1990 or 2000 census, or both. The authors evaluated mortality from different malignant neoplasms of the lymphoid and hematopoietic tissue (i.e., various types of acute and chronic leukemias and lymphomas) from 1990 to 2008. Occupational exposure to ELF magnetic fields was assessed using the study subjects' reported job title at the time of the census and a job-exposure matrix that assigned exposure levels of low, medium, or high. None of the hematopoietic cancer types included in the main analyses were statistically associated with medium or high exposure to ELF magnetic fields in the fullyadjusted models. Adjustment for other occupational exposures, including solvents, pesticides, herbicides, metals, and electric shocks, had a very small effect on the risk estimates.

The authors reported statistically significant associations for myeloid leukemia among men who were ever highly exposed at the time of both censuses (HR: 1.31, 95%: 1.02-1.67) and for acute myeloid leukemia among men ever highly exposed at the time of both censuses and additionally during their vocational training (HR: 2.75; 95%: 1.11-6.83). Both estimates were based on a very small number of cases (n = 6 and 5, respectively). Lung cancer mortality, included as an outcome unlikely to be caused by EMF, showed statistically significant associations and a clear exposure-response pattern with exposure to ELF magnetic fields. The authors hypothesized *a priori* that lung cancer was not associated with exposure to ELF magnetic fields (Huss et al., 2018a, p. 468). Thus, this finding clearly indicates that confounding by smoking, which is a well-established cause of both lung cancer and leukemias/lymphomas, remains a major weakness of the study, and may explain the association reported in some of the sub-analyses. The authors concluded that their analysis "*provided no convincing evidence for an increased risk of death*" from hematopoietic cancers in workers occupationally exposed to ELF magnetic fields (Huss et al., 2018a, p. 467).

Page 60 of 116

In the same publication, Huss et al. (2018a) also reported a meta-analysis of 28 epidemiologic studies of occupational exposure to ELF magnetic fields and acute myeloid leukemia published up to September 2017. The authors reported a weak overall association with acute myeloid leukemia, with a summary RR of 1.21 (95% CI: 1.08, 1.37). The authors noted that studies evaluating cumulative exposure were more likely to observe increase risks of acute myeloid leukemia; in the subset of studies that evaluated cumulative exposure (n=4), the summary RR was 1.51 (95% CI: 1.06, 1.37). The authors concluded that their findings were in line with previous meta-analysis results and stated that if any risk of hematopoietic cancers exists from occupational magnetic field exposure "*it appears to be small*" (Huss et al., 2018a, p. 474). In summary, recent results did not provide substantial new evidence for a consistent association between EMF and leukemia overall, leukemia sub-types, or lymphoma in adults that would alter previous conclusions. While some scientific uncertainty remains on a potential relationship between adult lymphohematopoietic malignancies and magnetic-field exposure because of continued deficiencies in study methods, the current database of studies provides inadequate evidence for an association (EFHRAN, 2012; SCENIHR, 2015).

 Table 7.
 Studies of adult leukemia/lymphoma (2017-2018)

Authors	Year	Study
Huss et al.	2018a	Occupational extremely low frequency magnetic fields (ELF-MF) exposure and hematolymphopoietic cancers - Swiss National Cohort analysis and updated meta-analysis.

4.1.5 In vivo studies of carcinogenesis

It is standard procedure to conduct studies on laboratory animals to determine whether exposure to a specific agent leads to the development of cancer (USEPA, 2005). This approach is used because all known human carcinogens also have been shown to cause cancer in laboratory animals and such studies are better suited to determining causation than epidemiology studies (IARC, 2002).

Magnetic field bioassays

The major focus of interest is on chronic bioassay studies in which animals, including those with a particular genetic susceptibility to cancer, are exposed at high levels over their entire

lifespan (or a large part of it) and tissue evaluations are performed to assess the incidence of tumors in many organs. These studies are judged to be one of the gold standards for identifying carcinogenic agents and are often considered when establishing regulatory actions.

The 2007 WHO review described four large-scale, long-term studies of rodents exposed to magnetic fields over the course of their lifetime that did not report increases in any type of cancer (Mandeville et al., 1997; Yasui et al., 1997; McCormick et al., 1999; Boorman et al., 1999a, 1999b). Fam and Mikhail (1996) reported that some animals developed a type of lymphoma similar to childhood ALL, but other studies that exposed transgenic mice predisposed to develop leukemias to ELF magnetic fields did not show an increased incidence of this lymphoma type (Harris et al., 1998; McCormick et al., 1999; Sommer and Lerchl, 2004). More recent studies also do not report effects of chronic magnetic field exposure at levels from 20 mG to 10,000 mG on the incidence of lymphomas or other cancer types (Qi et al., 2015; Soffritti et al., 2016a, 2016b).

Magnetic field exposure + known carcinogens

Other studies test whether the exposure of interest, in combination with a known carcinogen, produces a promotional or co-carcinogenetic effect, or whether the exposure in combination with a known carcinogen and a known promoter produces a co-promotional effect. These types of studies can be problematic in their interpretation because of the sometimes limited nature of the interaction of treatments with particular tissues and because of the complexity of the study designs and conditions.

Most studies reviewed by the WHO did not find evidence that magnetic-field exposure, when combined with chemical carcinogens, affected the development of tumors in skin, liver, etc. For over a decade, however, one laboratory in Germany has reported that the incidence of mammary tumors caused by 7, 12-dimethylbenz[a]anthracene (DMBA) was generally, but inconsistently, increased by magnetic-field exposure (Löscher et al., 1993, 1994, 1997; Baum et al., 1995; Löscher and Mevissen, 1995; Mevissen et al., 1993a, 1993b, 1996a, 1996b, 1998). The reported influence of magnetic fields on the carcinogenic effects of DMBA as reported by the German laboratory could not be replicated in studies from laboratories supported by the U.S. National Toxicology Program (Anderson et al., 1999; Boorman et al.1999a, 1999b; NTP, 1999). The

WHO concluded that the inconsistent findings across laboratories may be due to differences in experimental protocols and/or the use of different rat sub-strains, only some of which may be susceptible to the promotional effects of magnetic fields on mammary tissue. Based on the research available at the time, the WHO concluded that, "*There is no evidence that ELF exposure alone causes tumours. The evidence that ELF field exposure can enhance tumour development in combination with carcinogens is inadequate*" (WHO 2007, p. 322).

In light of the available evidence that exposure to magnetic fields alone does not increase the occurrence of cancer, most studies published subsequently have investigated the potential promotional or co-carcinogenic effects of magnetic-field exposure. These studies show that long-term exposure to magnetic fields does not alter the incidence of brain tumors or leukemia/lymphoma in rats and mice treated with the chemical initiators DMBA (Negishi et al., 2008), ethylnitrosourea (Chung et al., 2008), or n-butylnitrosourea (Bernard et al., 2008) or the cancer incidence rates or survival time in a strain of mice genetically predisposed to develop leukemia (Chung et al., 2010). The German laboratory continues to report findings similar to earlier work and have explored potential associations of magnetic-field exposure with the expression of certain genes and proteins in the mammary tissue of different rat strains (Fedrowitz and Löscher, 2008, 2012). Other studies have looked for effects of chronic magnetic fields exposure in combination with exposure to ionizing radiation such as x-rays (Soffritti et al., 2016a) or the carcinogen formaldehyde (Soffritti et al. 2016b), but because of multiple flaws in the design and analysis of the data, the results are insufficient to support an interaction between magnetic fields and known carcinogens.

In other studies investigating the therapeutic applications of magnetic fields, much higher level exposures to magnetic fields in combination with cancer treatments have been reported to reduce tumor size or increase the survival of animals injected with tumors (Berg et al., 2010; Wen et al., 2011; El-Bialy and Rageh, 2013; Mahna et al., 2014) or to reduce the development of pre-neoplastic lesions in the livers of rats initiated via chemicals and surgery (Jiménez-Garcia et al., 2010). Such anti-tumor effects are reported for magnetic field exposures greater than 70,000 mG.

Page 63 of 116

Damage to DNA and oxidative stress

Another focus of animal research is on the potential for magnetic fields to damage the DNA directly or in combination with known carcinogenic chemicals or x-rays (Lai and Singh, 2004). Studies have continued to look for evidence of DNA damage with mixed results from magnetic-field exposure alone (Mariucci et al., 2010; Okudan et al., 2010; Alcaraz et al., 2014; Villarini et al., 2013; Wilson et al., 2015; Korr et al., 2014) or in combination with radiation (Miyakoshi et al., 2012; Saha et al., 2014; Woodbine et al., 2015). Where small effects are reported, they are not convincing given that the reported changes in DNA damage markers appear to be unrelated to the strength of the magnetic field.

Without good evidence that either cancer or DNA damage is caused by magnetic-field exposures, a number of studies have investigated the role of magnetic-field exposures on indicators of tissue oxidative stress on the premise that oxidative stress is the mechanism that connects magnetic-field exposure to cancer. However, these studies typically have been of low quality, and not designed to establish any direct relevance of their findings to cancer (Seifirad et al., 2014; Glinka et al., 2013; Hassan and Abdelkawi, 2014; Deng et al., 2013; Cui et al., 2012; Duan et al., 2013; Manikonda et al., 2014; Martínez-Sámano et al., 2012; Akdag et al., 2013; Kiray et al., 2013).

Relevant studies 2017-2018

The research themes that have been pursued in animal studies of cancer and biological processes possibly related to cancer before 2017 have continued to be addressed in more recent *in vivo* studies.

Magnetic field bioassay

The Ramazinni laboratory that previously investigated the combined effects of magnetic fields and x-rays or formaldehyde in chronic exposure bioassays recently reported on the effects of life-long exposure to magnetic fields alone from levels of 200 mG to 10,000 mG (Bua et al., 2018).

Bua et al. (2018) reported no effect of magnetic-field exposure on the incidence of total tumors in any group exposed to magnetic fields or on food and water consumption, body weight, or survival. The incidences of the specific types of malignancies reported, specifically mammary gland tumors, Schwannomas of the heart, thyroid C-cell carcinomas, and hemolymphoreticular neoplasia (HLRN) also were not increased by magnetic field exposure. Unexpectedly, Bua et al (2018) reported a statistically significant 8.4% decrease in malignant tumors in male rats following lifelong exposure to a 1,000 mG magnetic field. Based on these data, Bua et al. concluded that the study "provided no evidence of any carcinogenic effect related to the exposure of ELF EMF alone" (p. 274) This result is consistent with a previous report from this same laboratory (Soffritti, 2010). Contrary to good experimental practice, the control group in this study was the same as used in other studies from this laboratory (Soffritti, 2010; Soffritti et al., 2016a). Further, this study suffers from the same flaws in design and data analyses as the other studies from ,the Ramazinni laboratory, including the failure to analyze the results by cage rather than per individual animal.

Magnetic field exposure + known carcinogens

As noted previously, several investigators have investigated the capability of very strong magnetic fields to enhance the anti-tumor effects of compounds used to treat cancer.

Only one study was recently reported in this category. Yuan et al. (2018) injected G401 nephroblastoma cells into nude mice (that have an impaired immune system due to a genetic mutation that causes deterioration of the thymus gland). With the immune system partially deactivated in the nude mice, tumors grow rapidly from the injected kidney cells. On the 7th day after injection of cells, groups of 8 mice were treated for 15 days with: 1) 50-Hz AC magnetic fields (intensity varying from 10,000 mG to 25,000 mG combined with DC magnetic fields (intensity between 30-40,000 mG), 2) the magnetic field with the anti-tumor drug, cisplatin – 3x/week, 3) just cisplatin – 3x/week, or 4) sham-exposure to the magnetic field and cisplatin (control). Exposures each day consisted of 3-4 treatment sessions. In each session, magnetic field exposure consisted of 8rounds (combinations of AC and DC magnetic fields of varying intensity) lasting 10 minutes each.

Yuan et al. (2018) reported that neither the magnetic field nor cisplatin treatments alone affected the weight of the tumors. However, when the mice were treated with both magnetic fields and cisplatin, the average weight of the tumors decreased by about 40%. The effects of the treatments on kidney and liver enzymes released into blood were measured to monitor organ toxicity. A statistically increase in levels of AST and ALT was observed in mice with the combined treatment (described as a "trivial" side effect). None of the treatments affected the weight of the mice measured 3 days after the last treatments.

Markers of immune function and oxidative stress

Various short-term studies have been conducted to investigate potential mechanisms related to carcinogenesis, including immune-function changes and oxidative stress.

The immune system is thought to play an important role in the immunosurveillance against cancer cells. Further, ALL, one of the cancers of concern for EMF exposures in children, arises in cells of the immune system. Thus, there is an interest in the potential effects of EMF exposures on immune function. Because most cancers elicit a response from the immune system, blood levels of certain chemokines (important to inducing immune system functions) are reported to increase when various types of cancer occur. Li et al. (2018) investigated the chemokine response of Balb/c mice (100 per group) exposed to 50-Hz magnetic fields at levels of 0 (sham controls), 1,000 mG, 5,000 mG, and 25,000 mG for <1, 1, 10, 30, or 90 days. At each time point, blood was drawn from four mice and the average value reported. The investigators analyzed the samples for nine different chemokines that affect the immune response by promoting pro-inflammatory functions and recruiting immune cells to sites of infection.

The investigators reported that exposure to magnetic fields over 90 days did not affect the body weight of the mice. Nor did the level of magnetic-field exposure have a significant effect on the chemokine levels in blood measured by immunoassay, with two exceptions: MCP-1 and EOTAXIN-1. The change in these chemokines was confirmed by ELISA assay and the clearest increase in levels was at 5,000 mG; magnetic field exposure at the higher level of 25,000 mG reduced the increase in these chemokines. The authors reported that they did not see the

expected dose-dependent rise in chemokines. The study had serious limitations in the experimental design and procedures, including small numbers of mice from each group evaluated at any single timepoint and for any specific chemokine, the failure to randomly allocate animals to treatment groups or to analyze the cage, not the individual animals in the statistical analyses. It is not clear that the differences reported are attributable to magnetic-field exposure *per se*. Further, because chemokines are important in eliciting immune reactions, an increase in chemokine levels may be indicative of a protective effect rather than increased susceptibility to cancer.

Oxidative stress is a condition in which oxygen free radical levels in tissues are elevated and is one mechanism by which DNA damage, as well as other forms of cellular damage, may occur. While there is general agreement that oxidative stress from endogenous cellular processes are an overwhelming source of damage to DNA and other cellular components (de Bont and Larebeke, 2004), whether such mechanisms are activated by magnetic fields is under investigation. Previous in vivo studies that have evaluated whether magnetic-field exposure may be associated with oxidative stress provided mixed results.

Luo et al. (2016) investigated the potential effects of magnetic-field exposure on a variety of physiologic measures related to cellular oxidative processes. This study was predicated upon the theory that prolonged, uncompensated, high levels of oxidative products might contribute to cancer and neurodegenerative disease. The WHO (2007) and SCENIHR (2015) previously reviewed similar studies. Luo et al. (2016) suggested that a decline in superoxide dismutase (SOD) and a rise in malondialdehyde (MDA) in blood and the brain cortex are indicative of oxidative stress in cells.

Luo et al. (2016) randomly assigned male ICR mice in groups of 12 to 50-Hz, 40,000 mG, 60,000 mG, 80,000 mG, or 100,000 mG magnetic fields, or sham-exposure (control) conditions for 4 hours per day and assayed the blood and brain for SOD and MDA levels after 7, 14, 21 and 28 days. The authors observed noticeable and statistically significant changes in these two indicators in the predicted directions with exposures at or above 80,000 mG. The design of the study and the effects reported are similar to those reported in a previous study from this laboratory (Duan et al., 2013.

In the Luo et al. (2016) study, other groups of mice exposed to 80,000 mG magnetic fields also were orally administered 60, 90, or 120 mg of an antioxidant (lotus seedpod procyanidins [LSPC]) for 15 days before magnetic-field exposure and daily thereafter with magnetic-field exposure for an additional 28 days. The highest LSPC dose reversed the changes in SOD activity and MDA levels in the blood and brain cortex compared to mice exposed to magnetic fields only; changes in other oxidative indicators, including catalase, glutathione peroxidase, glutathione reductase, and glutathione-S-transferase, were also reversed. A strength of the study is that the authors tested for effects of magnetic fields at multiple exposure levels and randomized the mice to the experimental groups, which minimizes systematic bias. Yet, while the study was reported in detail, the analysis of the data was not performed blind, the authors reported no controls on noise and vibration from the magnetic-field coils and power supply, and like Bua et al. (2018), the authors did not properly account for the multiple animals exposed in each cage in the statistical analyses.

Zhang et al. (2016b) compared a group of males who worked in 110-410 kilovolt substations to other administrative staff with the utility with insignificant exposure on their urine levels of oxidative stress markers (isoprostane and 8-OHdG) in a simple cross-sectional study. In addition, subjects within each group were given capsules of green tea extract or a placebo twice a day for 12 months. The authors suggested that the higher levels of oxidation products in the urine of the substation workers were related to magnetic field exposure, as the substation and administrative workers did not differ with respect to multiple other factors. Treatment with green tea extract reduced the levels of oxidation products among substation workers, but not administrative workers; the levels increased again after stopping green tea extract treatment.

The subject population could not have been randomly allocated to the substation and administrative groups as claimed and there was no validation of the exposure differences between groups with respect to magnetic fields or other occupational exposures. Further, a substantial number of the study participants (546 or 63%) dropped out of the study before its completion. While the Zhang et al. study provided no evidence to link magnetic fields to higher levels of oxidation products in urine (apart from some other unmeasured factor), it does,

however, provide a reasonable basis to suggest that a daily ingestion of green tea extract may reduce urinary measures of oxidative stress.²⁴

Overall, it is hard to draw any firm conclusions from these studies. Although markers of oxidative stress were generally increased with higher rather than lower magnetic-field exposures, it is not known if this effect is reversible or even biologically relevant. Independent replication of the findings in studies with greater sample sizes and blinded analyses is needed. Moreover, without studies that are specifically designed to quantitatively assess the relationship between markers of oxidative stress and measurements of DNA damage in an established model animal system, any relationship to a carcinogenic process is based on speculation rather than scientific evidence.

In summary, the *in vivo* studies published since the last update do not alter the previous conclusion of the WHO that there is inadequate evidence of carcinogenicity due to ELF EMF exposure.

Reviews of in vivo research

Reviews of *in vivo* research, including studies on carcinogenesis by SSM (2013, 2014, 2015) and SCENIHR (2015) cover a good deal of the research published after 2012. The relevant conclusions are:

SSM (2013):

Other studies indicated increased oxidative stress, again mostly by exposures at levels well above the current exposure limits. One study showed indications for tumour growth inhibition by a 100 mT [1,000,000 mG] field, but with only small numbers of animals. Replication is necessary to obtain more insight. In general, the latest animal studies do not contribute to understanding a mechanism that could explain the association found in epidemiological studies between long term exposure to ELF magnetic fields below 1 μ T [10 mG] and an increased risk of

A very similar study from this same laboratory also reported that another antioxidant, resveratrol, had similar effects in lowering elevated levels of urinary isoprostane and 8-OHdG in power plant workers with average estimated exposures of 1.4 kV/m and 200 mG for more than 20 years but not in control workers (Zhang et al., 2017).

childhood leukaemia. Hence, there is still a need for dedicated studies in this area using new animal models (pp. 28-29).

SSM (2014):

In general, the results of the studies are not very consistent. In some studies a function may be increased and in others decreased, while dose-responses cannot be derived. Most of the results are from single studies that need to be replicated in order to establish whether the observed effects are real or not. Also the large variety of exposure schedules used does not add to get a unified picture. Finally, none of these studies provide information that can be used in the interpretation of the association found in epidemiology studies between ELF magnetic field exposure and an increased risk of childhood leukaemia (p. 36).

SSM (2015):

With the exception of single studies, the quality of the experiments and their description did not substantially improve compared to the previous years ...Furthermore, referring to direct DNA-damage due to "low doses" of ELF-MF or presenting "dose-dependencies" using two groups only is somehow doubtful. Overall and similar to the previous SSM report, the results of the described studies are not very consistent (p. 9).

SCENIHR (2015):

Previously SCENIHR (2009) concluded that animal studies did not provide evidence that exposure to magnetic fields alone caused tumours or enhanced the growth of implanted tumours. The inclusion of more recent studies does not alter that assessment. In addition, these studies do not provide further insight into how magnetic fields could contribute to an increased risk of childhood leukaemia (p. 161).

Based on *in vivo* research published after these reviews and evaluated in this report, these conclusions are still appropriate.²⁵

²⁵ A review of human cytogenetic studies involving exposure to ELF-EMF magnetic fields (most involving occupational exposures) also has drawn the conclusion that "no firm conclusion can be drawn with respect to alleged ELF-EMF induce genetic effects" (Maes and Verschaeve, 2016, p. 2347).
Authors	Year	Study
Bua, L., et al	2018	Results of lifespan exposure to continuous and intermittent extremely low frequency electromagnetic fields (ELFEMF) administered alone to Sprague Dawley rats.
Li, H., et al.	2018	Eotaxin1 and MCP1 serve as circulating indicators in response to power frequency electromagnetic field exposure in mice
Luo, X., et al.	2016	Chemoprotective action of lotus seedpod procyanidins on oxidative stress in mice induced by extremely low-frequency electromagnetic field exposure
Yuan et al.	2018	The antitumor effect of static and extremely low frequency magnetic fields against nephroblastoma and neuroblastoma.
Zhang et al.	2016	Effects of dietary green tea polyphenol supplementation on the health of workers exposed to high-voltage power lines.
Zhang et al.	2017	Resveratrol may reverse the effects of long-term occupational exposure to electromagnetic fields on workers of a power plant.

 Table 8.
 Studies of in vivo carcinogenesis (2017-2018)

4.2 Reproductive and developmental effects

Studies have evaluated the relationship between ELF EMF and fertility, pregnancy outcomes, and prenatal and postnatal developmental effects. The effect of occupational exposures and contact with video display terminals, electric blankets, and heated beds has been studied on miscarriage, infertility, low birth weight, and select birth defects (e.g., neural tube defects, cleft palate defects), but no consistent findings emerged.

In 2002, two studies received considerable attention because of a reported association between peak magnetic-field exposure greater than approximately 16 mG and miscarriage—a prospective cohort study of women in early pregnancy (Li et al., 2002) and a nested case-control study of women who miscarried compared to their late-pregnancy counterparts (Lee et al., 2002). In an accompanying editorial to these two papers, a well-known epidemiologist proposed a hypothesis that the observed association may be the result of behavioral differences between women with healthy pregnancies (i.e., less physically active) and women who miscarried (i.e., more physically active), as opposed to a causal relationship between EMF and miscarriage (Savitz, 2002). Savitz proposed that physical activity is associated with higher likelihood of peak magnetic-field exposures at any given cut-points, and nausea commonly experienced in early, healthy pregnancies and the cumbersomeness of late, healthy pregnancies would reduce

physical activity levels, thereby decreasing the opportunity for exposure to peak magnetic fields. Later studies that reported consistent associations between activity (mobility during the day) and peak magnetic-field exposure metrics (Mezei et al., 2006; Savitz et al., 2006; Lewis et al., 2015) provided empirical support to the notion that the associations observed in Lee et al. (2002) and Li et al. (2002) were not due to a causal relationship but were likely due to behavioral differences between cases and non-cases. Other criticisms of the two studies also considered the timing of EMF measurements of the study subjects. In the Li et al. (2002) study, nearly half of women who had miscarriages in the cohort had their magnetic-field measurements taken after the miscarriage occurred, when changes in physical activity may have already occurred; in the Lee et al. study (2002), all measurements occurred after the miscarriage.

The scientific panels that have considered these two studies concluded that the possibility of bias in the studies precludes making any conclusions about the effect of magnetic fields on miscarriage (NRPB, 2004; FPTRPC, 2005; WHO, 2007). With respect to epidemiologic studies, the WHO concluded *that "On the whole, epidemiological studies have not shown an association between adverse human reproductive outcomes and maternal or paternal exposure to ELF fields. There is some evidence for an increased risk of miscarriage associated with maternal magnetic field exposure, but this evidence is inadequate"* (WHO, 2007, pp. 8-9). The WHO also concluded that, in general, experimental studies provide no consistent or convincing evidence in support of a potential adverse effect of EMF on human reproductive and developmental outcomes, and concluded that "Overall, the evidence for developmental and reproductive effects is inadequate" (WHO, 2007, p. 9).

Epidemiologic studies on pregnancy and reproductive outcomes that were reviewed in the previous Exponent reports provided little new insight in this research area and did not change the classification of the data from earlier assessments as inadequate.

Relevant studies 2017-2018

Several epidemiologic studies investigated the potential association between ELF magnetic-field exposure and miscarriage (Li et al., 2017; Sadeghi et al., 2017) or birth outcomes (Migault et al. 2018). Li et al. (2017) examined the association between magnetic-field exposure and miscarriage in a cohort of 913 pregnant women in California. Exposure was assessed using 24-

hour personal magnetic-field measurements collected on a single day during pregnancy, and the 99th percentile value observed during the 24-hour measurement period was used as the exposure of interest by the authors. Participants were also asked to keep a diary of their activities on the day of measurement. The authors reported an increased risk of miscarriage in women with high magnetic-field exposure (i.e., the 99th percentile value during the 24-hour measurement of \geq 2.5 mG) compared to women with low magnetic-field exposure (<2.5 mG) when measurements were collected on a typical day (defined as a day reflecting participants' typical pattern of work and leisure activities during pregnancy). They reported no association, however, among those women whose magnetic-field exposure was measured on a non-typical day, and no trend was observed for miscarriage risk with increasing magnetic-field exposures >2.5 mG. The authors did not report the overall TWA for the 24-hours of exposure that could be compared to previous studies.

While personal exposure measurements are an improvement in exposure assessment over some of the earlier studies, the collection of only one measurement over a single 24-hour period during pregnancy is a limitation of the Li et al. (2017) study, as day-to-day changes in exposure cannot be captured. No information was provided in the paper on the exact timing of the measurement (i.e., whether the measurement day preceded or followed the occurrence of miscarriage among cases); this is a substantial limitation, as measurements taken following miscarriage in a substantial fraction of cases, was a major criticism of the previous study by the same research team (Li et al., 2002). Li et al. (2017) also failed to measure mobility during the measurement day, a potential major source of confounding in the study (e.g., Savitz, 2002; Mezei et al., 2006; Savitz et al., 2006). Varying levels of mobility between women with healthy pregnancies and women who suffer a miscarriage remain a viable alternative explanation for the findings in both the previous and the current studies.

Iranian scientists (Sadeghi et al., 2017) conducted a case-control analysis of preterm birth and residential distance to high-voltage overhead power lines. The researchers identified 135 cases of live spontaneous preterm birth in an Iranian hospital between 2013 and 2014 and compared their estimated exposure to 150 controls with term live births selected using randomized-digit dialing. The nearest linear distance from the high voltage power lines to the maternal residence

during pregnancy was measured using geographical information system software. Exposure was categorized using distance from power lines (<600 meters vs. \geq 600 meters). The authors reported no statistically significant associations between preterm birth or birth defects and the mothers' residential distance from power lines. One of the main limitations of the study is the reliance on maternal address within 600 meters to high-voltage power lines as a surrogate for exposure. No elevation of ELF EMF levels can be expected for distances from approximately 100 to 600 meters in that zone; thus, no valid conclusions can be drawn from the study with respect to exposure to EMF.

Migault et al. (2018) studied the relationship between maternal cumulative exposure to ELF EMF and two pregnancy outcomes (moderate prematurity and being small for gestational age) within a prospective birth cohort in France. The study included 18,329 infants born at 33 weeks of gestation or more during 2011 and follows the children until 20 years of age. Cumulative exposure to both occupational and residential ELF EMF during pregnancy was assessed using the mothers' self-reported occupation and a job-exposure matrix. The job-exposure matrix included exposure estimates for five non-professional categories (e.g., housewife, student, and unemployed) that were used to estimate residential exposure. The authors observed no statistically significant association between maternal cumulative exposure and moderate prematurity or small for gestational age. The authors noted that the ability to consider both occupational and residential exposure in their cumulative estimates is a strength of the study but suggested that the small sample size in the high exposure categories limited the study's power to detect a potential association.

Using data from the Danish National Birth Cohort, Sudan et al. (2017) conducted a follow-up study to a previously reported association between intrauterine exposure to magnetic fields and childhood asthma (Li et al., 2011). The researchers examined 92,675 children born to 91,661 mothers who were pregnant between 1996 and 2002, and assessed intrauterine exposure of the children using distance from the residence of the mother during pregnancy to the nearest power line. They observed no association between magnetic-field exposure estimated by distance from power lines and asthma development, regardless of how the asthma diagnosis was defined. The authors noted, however, that the majority of mothers and children in the dataset had no

residential exposure from power lines (i.e., lived in a home that was located outside a specified distance to the nearest power line), thus limiting the ability to make firm conclusions. In addition, potential errors in the estimation of distances to power lines, which were used in the calculations of magnetic-field levels, are a limitation of the study's method of exposure assessment (Chang et al., 2014).

In summary, the recent epidemiologic studies evaluated do not provide substantial new evidence in support of an association between EMF and reproductive or developmental outcomes. The classification of the data as inadequate remains appropriate, as studies in this research area still suffer from limitations in study design, sample size, and exposure assessment method.

Authors	Year	Study
Li et al.	2017	Exposure to magnetic field non-ionizing radiation and the risk of miscarriage: a prospective cohort study.
Migault et al.	2018	Maternal cumulative exposure to extremely low frequency electromagnetic fields and pregnancy outcomes in the Elfe cohort.
Sadeghi et al.	2017	Preterm birth among women living within 600 meters of high voltage overhead Power Lines: a case-control study.
Sudan et al.	2017	Re-examining the association between residential exposure to magnetic fields from power lines and childhood asthma in the Danish National Birth Cohort.

 Table 9.
 Studies of reproductive and developmental effects (2017-2018)

4.3 Neurodegenerative disease

Research into the possible effect of magnetic fields on neurodegenerative diseases began in 1995, and the majority of research since then has focused on Alzheimer's disease and a specific type of motor neuron disease called ALS, which is also known as Lou Gehrig's disease. Based on the initial findings on the Alzheimer's disease the NRPB concluded in 2001 that there was *"only very weak evidence to suggest that it [ELF magnetic fields] could cause Alzheimer's disease"* (NRPB, 2001b, p. 21). Early studies on ALS also reported an association between ALS mortality among workers with certain electrical occupations. The review panels, however, were hesitant to conclude that the associations provided strong support for a causal relationship because they felt that an alternative explanation (i.e., electric shocks received at work) may be the source of the observed association.

Also including more recent studies, the WHO panel concluded that there is *"inadequate"* data in support of an association between magnetic fields and Alzheimer's disease or ALS. They stated that *"When evaluated across all the studies, there is only very limited evidence of an association between estimated ELF exposure and [Alzheimer's or ALS] disease risk"* (WHO 2007, p. 194). While a subsequent meta-analysis also reported an association between occupational EMF exposure and Alzheimer's disease (Garcia et al., 2008), its conclusion was necessarily limited by the quality of the studies included in the analysis. A Swiss study that was the first to examine residential proximity to high-voltage power lines and neurodegenerative disease, reported an increase in Alzheimer's disease mortality among people living with 50 meters of transmission lines, but observed no association for ALS, Parkinson's disease, and multiple sclerosis (Huss et al., 2009) spurred interest in research on Alzheimer's disease. Based on a review of the evidence that also considered these studies, the Health Council of the Netherlands EFHRAN review still considered the evidence as "inadequate" for all forms of neurodegenerative diseases (EFHRAN, 2012).

Following the research recommendations of the WHO, scientists conducted epidemiologic research that studied exposure to ELF EMF and development of neurodegenerative diseases. Overall, these studies did not provide consistent and convincing support for an association. Several meta-analyses of these studies reported weak to no evidence of an association between occupational exposure to ELF magnetic fields and neurodegenerative disease (Zhou et al., 2012; Vergara et al., 2013). The authors of these meta-analyses concluded that potential within-study biases, evidence of publication bias, and uncertainties in the various exposure assessments greatly limit the ability to infer an association, if any, between occupational exposure to magnetic fields and neurodegenerative disease.

Relevant studies 2017-2018

Two case-control studies (Koeman et al., 2017; Vinceti et al., 2017) and two meta-analyses (Huss et al., 2018b; Röösli and Jalilian, 2018) have been recently published regarding EMF exposure and ALS. Koeman et al. (2017) conducted a nested case-control analysis within the Netherlands Cohort Study that assessed various occupational exposures, including solvents, pesticides, metals, ELF magnetic fields and electric shocks, and ALS mortality in a cohort of

more than 120,000 men and women. The analysis included 136 ALS cases and a random subset (n=4,344) of the cohort study population. Information on occupational history and potential confounders were assessed using self-administered questionnaires at enrollment. Exposure to various agents in the occupational environment was assessed using job-exposure matrices. The authors reported a statistically significant association among men who were ever highly exposed; however, this was based on a small number of cases (n=9) in the high exposure category. In addition, they reported a statistically significant association between the ALS mortality among men for those with the highest 30 percent or more of cumulative ELF magnetic-field exposure in the cohort; this association was no longer statistically significant when adjusted for the effects of other occupational exposures, including insecticides. They reported no statistically significant associations for other occupational exposures investigated in the study and that due to the overall low number of exposed women, risk analyses for women were "*largely uninformative*."

Vinceti et al. (2017) conducted a population-based, case-control study of magnetic fields from high-voltage power lines and ALS within Italy. The authors included 703 ALS cases, diagnosed between 1998 and 2011, and a sample of 2,737 randomly selected controls from the same provinces. The authors used information on the residential addresses of the cases and controls and geodata on high-voltage power lines (with voltages between 132 and 380 kV) to model magnetic-field exposure at the study subjects' residences. The authors reported no statistically significant associations between ALS and calculated magnetic-field levels, and they observed no exposure-response trend. The authors concluded that their findings "*appear to confirm*" that exposure to magnetic fields from power lines occurring in the general population is not associated with an increased risk of ALS (Vincente et al., 2017, p. 583).

Huss et al. (2018b) conducted a meta-analysis combining data from 20 studies of occupational exposure to ELF magnetic fields and ALS. Overall, the authors reported a weak association with borderline statistical significance for ALS and estimated ELF magnetic-field levels (summary RR 1.14; 95% CI, 1.00-1.30) and a somewhat stronger association in a subset of six studies with full occupational history compared to studies where occupation was available only at certain time points. The authors also reported substantial heterogeneity among studies, evidence for

publication bias, and the lack of a clear exposure-response relationship between estimates of ELF magnetic fields and ALS. Röösli and Jalilian (2018) combined data from five epidemiologic studies that examined residential exposure to ELF magnetic fields from high-voltage power lines and ALS. The authors reported no statistically significant associations; the pooled RR for the most exposed populations (either <200 meters from high-voltage lines or $>0.1 \mu T [>1 mG]$) was 0.71 (95% CI, 0.48-1.07).

Pedersen et al. (2017) updated a prior cohort study (Johansen, 2000) of occupational exposure to ELF magnetic fields and CNS disease, including dementia, motor neuron disease, Parkinson's disease, multiple sclerosis, and epilepsy, among more than 32,000 male electric utility workers in Denmark. The authors identified cases within the occupational cohort of electric utility workers from the Danish National Patient Registry diagnosed from 1982 to 2010. They estimated exposure to ELF magnetic fields using a JEM and company records of job title and area of work, and classified into three categories (<0.1 μ T [<1 mG], 0.1-0.99 μ T [1-9.9 mG], and \geq 1.0 μ T [\geq 10 mG]).

Pederson et al. compared disease incidence of exposed worker within the cohort to: 1) disease incidence in the general population of Danish men (external comparison); and 2) to disease incidence among unexposed workers to account for the potential healthy-worker effect (internal comparison). No consistent pattern of disease association was reported by the authors for any of the investigated outcomes. While the external comparison indicated statistically significant associations for all types of dementia in the highest exposure category of ELF magnetic fields, the internal comparison, which is the more appropriate comparison, reported no such associations. The authors reported no statistically significant increases with exposure to ELF-magnetic fields for motor neuron disease, Parkinson's disease, multiple sclerosis, or epilepsy in either external or internal comparisons.

Checkoway et al. (2018) investigated the association between Parkinsonism²⁶ and occupational exposure to several agents, including endotoxin, solvents, shift work, and magnetic fields,

²⁶ Parkinsonism is defined by Checkoway et al. (2018) as "a syndrome whose cardinal clinical features are bradykinesia, rest tremor, muscle rigidity, and postural instability. Parkinson disease is the most common neurodegenerative form of parkinsonism" (p. 887).

among female Shanghai textile workers. The study included 537 retired cotton factory workers who were at least 50 years of age, and 286 age-matched controls who were retired cotton factory workers not exposed to cotton dust (which was used to define endotoxin exposure). Exposure to magnetic fields was assessed using a JEM. The authors reported no statistically significant associations between occupational exposure to magnetic fields and parkinsonism. In addition, the authors, did not observe statistically significant associations with endotoxin, shift work, or solvent exposure. Huss et al. (2015) conducted a meta-analysis of 11 studies of occupational exposure to ELF magnetic fields and Parkinson's disease. The authors observed no statistically significant association (summary RR 1.05, 95% CI 0.98-1.13) and they reported that overall, there was "no evidence that the exposure to ELF-MF [magnetic fields] increases the risk of Parkinson's disease" (Huss et al., 2015, p. 7348).

Jalilian et al. (2018) conducted a meta-analysis of 20 epidemiologic studies of occupational exposure to ELF magnetic fields and Alzheimer's disease. The authors reported a moderate, but statistically significant overall association for Alzheimer's disease (summary RR 1.63; 95% CI, 1.35-1.96), with weaker associations in cohort studies than in case-control studies. The authors also reported substantial heterogeneity among studies, and evidence for publication bias. Pooling results from studies with *"higher risk"* of bias, as assessed by the authors, resulted in stronger associations, suggesting that bias in the studies likely contributed to the reported associations.

In summary, many of the recent studies on neurodegenerative diseases have represented methodological improvements (e.g., increased sample size, improved exposure assessment, inclusion of incidence cases) compared to previous studies. In spite of these improvements, the overall evidence from these studies do not materially change the assessment that there is no consistent or convincing support for a causal association between EMF exposure and neurodegenerative diseases.

Authors	Year	Study
Checkoway et al.	2018	Occupational exposures and parkinsonism among Shanghai women textile workers.

Table 10.	Studies of	neurodegenerativ	e diseases	(2017-2018)
-----------	------------	------------------	------------	-------------

Authors	Year	Study
Huss et al.	2018b	Occupational exposure to extremely low-frequency magnetic fields and the risk of ALS: A systematic review and meta-analysis.
Jalilian et al.	2018	Occupational exposure to extremely low frequency magnetic fields and risk of Alzheimer disease: A systematic review and meta-analysis.
Koeman et al.	2017	Occupational exposure and amyotrophic lateral sclerosis in a prospective cohort.
Pedersen et al.	2017	Occupational exposure to extremely low-frequency magnetic fields and risk for central nervous system disease: an update of a Danish cohort study among utility workers.
Röösli and Jililian	2018	A meta-analysis on residential exposure to magnetic fields and the risk of amyotrophic lateral sclerosis.
Vinceti et al.	2017	Magnetic fields exposure from high-voltage power lines and risk of amyotrophic lateral sclerosis in two Italian populations.

4.4 Electromagnetic hypersensitivity

The WHO 2007 report discussed anecdotal accounts of persons who reported that they could perceive EMF at levels below accepted thresholds and accounts of persons who believed they had developed a variety of symptoms including sleep disturbances, general fatigue, difficulty concentrating, dizziness, and eyestrain due to EMF exposure. Based on double-blind studies of human volunteers, office workers, and self-reported hypersensitive individuals, however, the WHO review concluded that the perception of EMF and health complaints are not related to exposure. Neither healthy volunteers nor self-identified hypersensitive individuals can reliably distinguish field exposure from sham-exposure. Also, no exposure-related differences were observed in levels of stress hormones or inflammatory mediators. The WHO proposed that electromagnetic hypersensitivity should more appropriately be termed "*idiopathic* environmental intolerance (IEI) with attribution to EMF" and explained that "[t]hese symptoms are not explained by any known medical, psychiatric or psychological disorder, and the term IEI has no medical diagnostic value. IEI individuals cannot detect EMF exposure any more accurately than non-IEI individuals, and well-controlled and conducted double-blind studies have consistently shown that their symptoms are not related to EMF exposure per se" (WHO, 2007, p. 137).

Studies published following the WHO review, overall, supported the conclusion that ELF EMF is not detected by self-identified sensitive subjects or other subjects, and that symptoms are not reliably elicited by exposure to ELF magnetic or electric fields over a range of exposure levels.

The 2012 EFHRAN report also concluded that the available evidence suggests the lack of an effect on people with "electrical hypersensitivity" (EFHRAN, 2012).

Subsequent studies continue to report that persons who identify as having electromagnetic sensitivity are unable to detect exposure to EMF better than chance (Van Moorselaar et al., 2016). Other reports indicate detection of 50-Hz magnetic fields to *"some extent"* or to *"small extent,"* but reporting of symptoms by these individuals were related to perceived exposure (Köteles et al., 2013; Szemerszky et al., 2015).

The 2015 SCENIHR review concluded "The studies published since the 2009 Opinion show discordant results. However, observational studies suffered from weaknesses and do not provide convincing evidence of an effect of ELF exposure on symptoms in the general population and most experimental evidence also points to the absence of any causal effect." (p. 184)

Relevant studies 2017-2018

Most recent research on IEI attributable to EMF (IEI-EMF) has focused on addressing the prevalence of self-reported IEI symptoms in populations and their psychological characteristics (Andrianome et al., 2017; Dömötör et al., 2017; Slottje et al., 207; Claeson et al., 2018; Gruber et al., 2018; Karvala et al., 2018; Huang et al., 2018a,b; Vuokko et al., 2018). Another focus of recent research is on reports that mere perception of exposure to EMF even when no exposure is present can influence feelings of well-being among both persons who do and don't identify as IEI (Boehmert et al., 2018; Eltiti et al., 2018; Verrender et al., 2018)). The attribution of the reported symptoms to a perceived exposure is called the "*nocebo*" effect (i.e., symptoms explained by unconscious psychological reaction as a result of the expectation of an effect, rather than the effect of the exposure itself).

The only new experimental study of the responses of subjects actually exposed to magnetic fields (Van Moorselaar et al., 2017) is a follow-up to a previous double-blind randomized controlled trial that found subjects with self-identified electromagnetic hypersensitivity were not able to detect exposure to EMF better than chance (van Moorselaar et al., 2016).

Some IEI persons had criticized previous studies because they were conducted in laboratories which necessitated travel from home and this source of stress hampered their response to test exposures and that the test exposures did not match what they believed to be the source of their symptom (van Moorselaar et al. 2017). To address these concerns, van Moorselaar et al. (2017) devised a sophisticated and extremely well-controlled study. They visited members of an IEI self-help group at their homes and with the assistance of each volunteer determined what exposure ELF-EMF, RF, etc. they responded to. Then, volunteers were exposed to the type of electromagnetic field that they said they could detect within 15 minutes of exposure onset. Questionnaires were administered immediately before and after testing and after two or four months after testing. Of the 40 participants, only 4 stated that they could detect ELF-EMF. These volunteers were then exposed to magnetic field at levels they had selected ranging from 1.5 to 66 mG or no-field conditions in a double-blind randomized series of tests. None of the participants could reliably detect the presence of ELF-magnetic fields under blinded testing conditions although in the unblinded pre-test they asserted that they could. Similarly, other participants also were found to be unable to detect electromagnetic field at other frequencies and characteristics.

In summary, recent studies did not provide new sufficient evidence to change the overall conclusion that either self-identified individuals with electromagnetic hypersensitivity or members of the general populations can detect EMF exposure encountered in our environment, or that general non-specific symptoms are related to EMF exposure. The recent SCENIHR review stated that "*Overall, existing studies do not provide convincing evidence for a causal relationship between ELF MF exposure and self-reported symptoms*" (SCENIHR, 2015, p. 7).

Authors	Year	Study
van Moorselaar et al.	2017	Effects of personalised exposure on self-rated electromagnetic hypersensitivity and sensibility - A double-blind randomised controlled trial.

5 Possible Effects of ELF Electric and Magnetic Fields on Implantable Cardiac Devices

The sensing system of pacemakers and implantable cardioverter-defibrillators (ICD) is designed to be responsive to the heart's electrical signal. For this reason, other electrical signals can potentially interfere with the normal functioning of pacemakers and ICDs, a phenomenon called electromagnetic interference (EMI). Most sources of EMF are too weak to affect a pacemaker or ICD; however, EMF from certain sources (e.g., some appliances and industrial equipment) may cause interference. This section considers potential EMI associated with ELF EMF to implantable cardiac devices such as pacemakers and defibrillators.

In the presence of electromagnetic fields, devices can respond in different ways, defined as modes. The likelihood of interference occurring and the mode of the response depend on the parameters (e.g., strength, frequency, duty cycle) of the interfering signal, the patient's orientation in the electromagnetic field, the exact location of the device, and the variable parameters of the device that are specific to a patient. Modern devices incorporate various technological safeguards (e.g., shielding by titanium casing and electrical filtering) to minimize the potential for EMI (Dyrda and Khairy, 2008). Experimental research has been conducted to assess whether interference may occur when currents are induced in the patient's body by environmental electric fields and magnetic fields.

In the absence of specific recommendations from medical device manufacturers, the American Conference of Governmental Industrial Hygienists (ACGIH) suggested exposure levels to prevent pacemaker EMI. For electric fields, the ACGIH suggested keeping exposures below 1 kV/m, and for magnetic fields, they recommended exposure not exceed 1 G (ACGIH, 2015). These recommendations are general in nature and do not address that classes of pacemakers from some manufacturers are quite immune to interference even at levels much greater than these recommended guidelines. The ACGIH also recommended that patients consult their physicians and the respective pacemaker manufacturers before following any organizations' guidelines.

Page 83 of 116

Manufacturers of pacemakers and other implantable devices will typically follow the AAMI PC69:2007 or ANSI/AAMI/ISO 14117:2012 (North America) or EN 45502-2-1:2003 / IEC 45502-2-2:2003 (Europe) standards. These standards require a test to verify that the function of the cardiac device is not affected to at least a 2 millivolt peak-to-peak signal applied to the sensing electrodes. This test verifies immunity of a cardiac device²⁷ of at least 0.83 G (root-mean-square magnetic field) at 60 Hz. At 60 Hz, the reference levels in EC 519/99 (also known as 1999/519/EC) are 0.83 G and 4.167 kV/m—the standard assumes that only an electric or magnetic field is present at any time (CEU, 1999).

Moreover, the standard procedure (EN 50527-1:2010) to assess EMF exposure for workers with active implantable medical devices (AIMD) states that the "risk assessment is based on the approach that AIMDs are expected to work uninfluenced as long as the General Public Reference levels of 1999/519/EC (except for static magnetic fields) are not exceeded ..., where the AIMD has been implanted and programmed following good medical practice" (CENELEC, 2010). The procedure recommended by this standard contains steps for assessing that the field levels of EC 519/99 are not exceeded and that AIMD patients do not have higher than normal sensitivity settings on their device for clinical reasons.

Previous studies indicated occurrence of pacing abnormalities at magnetic-field levels that are much higher than the levels a person would encounter on a daily basis. While electric fields did produce interference at levels that can be produced by certain electrical sources (Toivonen et al., 1991; Astridge et al., 1993; Scholten and Silny, 2001; Joosten et al., 2009), most pacemakers were not affected by high levels of electric fields (up to 20 kV/m) and did not exhibit any pacing abnormalities. Joosten et al. (2009) showed that the most sensitive unipolar pacemakers may be affected by electric-field levels between 4.3 kV/m and 6.2 kV/m; however, most modern pacemakers are bipolar devices, which are designed specifically to reduce the potential for EMI. Joosten et al. (2009), for example, found that in Germany in 2007, only 6% of the pacemakers in use had a unipolar sensing system. The conclusions of other researchers are similar (Tiikkaja et al., 2013a,b; Napp et al., 2014; Korpinen et al., 2014; von Olshausen et al., 2016) regarding the

²⁷ The magnetic-field value is calculated using an average area (225 cm²) of a unipolar cardiac device. In rare cases, such as for a large patient with a unipolar implant, the immunity may be lower. For a patient with a bipolar lead configuration, the immunity will be higher.

relative rarity of interference, especially below ICNIRP reference levels, and that interference from use of specific devices is relatively common whilst interference from power lines or substations is not reported.

Relevant studies 2017-2018

Some recent medical case reports describe interference with pacemakers and other implantable cardiac devices produced by electrical currents are conducted from a mechanical heart implanted in the chest (Ahmed et al., 2017), faulting wiring in pump in a swimming pool (Roberto et al., 2017), or hot tub (Shenasa et al., 2018) to an implanted cardioverter defibrillator (ICD). Other studies have tested typical dental devices for potential EMF interference to ICDs without finding cause for concern (Dadalti et al., 2017), or reviewed studies of the application of electrical currents to teeth, if general precautions are followed (AlRahabi and Ghabbani, 2018). While some of these reports indicate the possibility of interference in certain scenarios with these equipment devices, these interference scenarios are not relevant to the exposure to EMF in electric utility environments due to differences in many factors, including, most importantly, the proximity of the interfering sources and the intensity and frequency of the interfering electromagnetic fields.

As in past reviews, there are reports that exposure of ICDs and pacemakers to EMF at sufficient intensities can affect device operation. This was demonstrated by Seckler et al. (2017) who stimulated the hearts of 160 patients in the University Hospital in Aachen, Germany with ICDs to determine what role the position and orientation of the leads from the device to the heard had on thresholds for interference. The maximum 50-Hz electric and magnetic field exposures tested were 30 kV/m and 25,500 mG, respectively. The researchers recommended that the orientation and position of the lead tips be optimized and the appropriate sensitivity of the pacemaker be set at the time of implant as these procedures could effectively minimize interference.

Stunder et al. (2017), also at the University Hospital in Aachen, measured the thresholds for interference from 50-Hz electric and magnetic fields for 119 patients with implanted pacemakers under conditions that were most likely to result in interference. At nominal sensitivity settings the threshold for 5 unipolar pacemakers was 9.6 kV/m or 2000 mG for

electric and magnetic fields, respectively, or 3 kV/m and 1,500 mG when exposed to electric or magnetic fields simultaneously. For 114 bipolar lead devices at nominal sensitivity settings, the lowest interference thresholds were 11.9 kV/m and 3,000 mG for electric and magnetic fields, respectively, or 5.2 kV/m and 2,550 mG when exposed to electric or magnetic fields simultaneously. When these pacemakers were set to their maximum sensitivity the thresholds were far lower, as would be expected. Since the general public would rarely, if ever, encounter a magnetic field from under transmission line or other utility source close to the threshold for interference of any of the devices tested (400 mG) set to maximum sensitivity, pacemaker interference is highly unlikely. On the other hand, when set to maximum sensitivity, about 35% of pacemakers tested in this study would be affected by an electric field of 10 kV/m, a value similar to the electric field (8-12 kV/m) under 765 kV transmission line (Tell et al., 1977). At maximum sensitivity the threshold drops to levels that could be found under some lower voltage lines. The authors place these results in practical context by comparing the theoretical rate of interference for pacemakers at nominal sensitivity for combined electric and magnetic field exposures at reference levels under the ICNIRP and ICES standards as 0-4.4 %, respectively. This compares to 16% of the pacemakers predicted to exhibit interference for a person using a hand drill.

De Rotte et al. (2017) placed ICDs from four different manufacturers in a human phantom dummy within a Piper Dakota single-engine plane to record electromagnetic interference that would be detected as tachycardia (R- or P-waves) or inappropriate defibrillation discharge. The devices were set at maximum sensitivity. No interference was detected during 11, 932 minutes of the testing. No measurements of EMF were made to assess the intensity or frequency of the EMF within the plane.

Lennerz et al. (2018) randomly assigned 108 volunteers with ICDs to sit within 4 of the most popular electric cars in Europe while the car was 'bench-driven' to achieve maximum electromagnetic field generation. Two cardiologists blindly reviewed the volunteers' electrocardiograms for abnormal device function, tachyarrhythmia, or other spurious functioning. The magnetic fields during operation on the test bench and during driving were measured as 301-1165 mG during charging. Fields measured in front seats 5 cm from the implanted device when the cars were not charging were much lower (20-36 mG). The authors indicated that these values were lower than measurements of about 200 mG made in the back seat in a sample of gasoline-powered cars (Halgamuge et al., 2010). No evidence of electromagnetic field interference with the performance of the ICDs was reported in this study.

Authors	Year	Study
De Rotte et al.	2017	Electromagnetic interference in implantable defibrillators in single-engine fixed-wing aircraft
Lennerz et al.	2018	Electric cars and electromagnetic interference with cardiac implantable electronic devices; a cross-sectional evaluation
Seckler et al.	2017	Effect of lead position and orientation on electromagnetic interference in patients with bipolar cardiovascular implantable electronic devices.
Strunder et al.	2017	In vivo study of electromagnetic interference with pacemakers caused by everyday electric and magnetic fields

Table 12. Studies of EMI (2017-2018)

6 Fauna and Flora Research

6.1 Fauna

Previous Exponent reports reviewed the relevant research and concluded that the research to date did not suggest that electric or magnetic fields result in any adverse effects on the health, behavior, or productivity of fauna, including livestock such as cows, sheep, and pigs, a variety of small mammals, deer, elk, birds, or bees.

Relevant studies 2017-2018

In this period, only one relevant study was identified. Shepherd et al. (2018) reported an exquisitely designed study of magnetic fields on the behavior of honey bees which was appropriately analyzed but unfortunately the investigators failed to control for confounding effects of vibration from the coils that produced the magnetic field.

The investigators performed three experiments. First, they attached the bees to plate with adhesive tape and then taught the bees to expect a sugar-water reward after delivery of a flower odor on 5 trials of the pairing of sugar water with the scent. After each of the 5 trials, bees were exposed to 200 mG, 1000 mG, 10,000 mG or no magnetic field for one minute after each trial. Evidence of learning was indicated by the bee extending it tongue-like proboscis (proboscis extension response [PER]) during the 10-second presentation of the odor stimulus before the sugar water was presented. A lower percentage of bees exposed to 200, 1,000 and 10,000 mG magnetic field displayed PER but the reduction was not significantly greater at the higher of these two field levels. Only bees from five of the six hives showed this response. All bees were tested one hour after training but no differences between the PER response at the end of training and the memory test one hour later were observed in any of the groups of bees.

In the second experiment the bees were attached to the plate with a light tether and the frequency of wing beats were measured from analyses of videotapes of exposure to magnetic fields or no magnetic field. Magnetic fields at 7,000 mG significantly increased wing-beat from about 2.5 to 6 beats per second.

The third experiment observed the effect of the 1,000 mG magnetic field on free-flying bees from 6 hives. A netting-enclosed passage from the hive led to a feeding station 1.4 meters away. To get to the feeding station the bees had to fly through the magnetic field generated by 2 wire coils. Over a 15 minute observation period the average number of bees leaving the hive increased and reaching the feeding station increased by 5 bees above pre-study baseline while the average number of bees exposed to 1,000 mG magnetic field declined by 3 bees. Having reached the feeding station (and not turning around *en route*), the number of bees returning to the hive was the same for bees exposed to the magnetic field or not.

While the researchers suggested that bees might encounter the highest levels of magnetic fields investigated in the study from 400 kV transmission lines, their calculations show that highest field levels are found only about 1 m from the phase conductors. As their calculations document, the magnetic fields at ground level where bees are visiting flowers are at least 50fold lower. As for the observed responses of the bees, there is a real question as to whether the bees were responding just to the magnetic field. Apparently the researchers were unaware that the magnetic coils they used to generate the magnetic field also produce noise and vibration at a frequency of 120 Hz in the same way that the electromagnet in audio speakers cause vibration and sound. Researchers in the field of magnetobiology know this and have urged others to take precautions against confounding effects of noise and vibration in magnetic field studies (IARC, 2002; Makinistian et al., 2018). In particular, it is surprising that the researchers did not design their magnetic field coils to minimize noise and vibration because that was a very specific concern of the researchers decades ago who started research on bees and magnetic fields and who design their magnetic field coils to solve this problem (Kirschvink, 1992; Kirschvink et al., 1997). In addition, the electricity flowing through the coils are a source of electric fields which should have been blocked by shielding. As for the bees, they don't have ears but they do have sensory cells that can detect vibration by mechanical contact or through the air by sound up to about 500 Hz and their ability to perceive even the very weak vibrations during honey bee waggle dances and other behaviors as part of normal bee communications has been explored in recent years (Wenner, 1964; Tsujiuchi et al., 2007; Collison, 2016).

In short, this study describes some interesting observations of bees during their experiments but does not clearly identify whether the response of the bees is caused by the magnetic field or

Page 89 of 116

vibrations and noise, or an electric field from the exposure system. Furthermore, it is unclear whether the behavior described in the unusual conditions of this study would occur in the real world where if the bees found locations of high magnetic field to be disturbing they could fly away from those regions an option not available to the captive bees in this study. This very option was mentioned by the researchers "*Moreover, the effects cause by EMF on flight could serve as an aversive stimulus that bees could avoid. There are therefore many questions remaining to be addressed to fully understand how EMFs could impact on bees.*" (p. 6).

Results of studies published since 2012 have not provided substantive new evidence that would alter previous conclusions.

6.2 Flora

Previous Exponent reports described the body of research on the possible effects of EMF on forest species and agriculture crops, concluding that researchers have found no adverse effects on plant responses at the levels of EMF produced by high-voltage transmission lines, excluding some corona-related effects from high-voltage lines on the growth of nearby trees. No peer review studies published since 2012 added substantial new information to our knowledge about effects of EMF on flora that would be relevant to transmission lines.

Glossary

Association – An association is a measure of how things vary together. They are measured by odds ratios and relative risks.

Basic restriction – The basic restriction is the electric field level or current density inside the body that is recommended as a limit to protect exposed populations. The term is used in standards or guidelines that recommend exposure limits.

Bias – Bias refers to any error in the design, conduct or analysis of a study that results in a distorted estimate of an exposure's effect on the risk of disease. For example, the characteristics of persons selected by telephone calls to participate in a study may not accurately reflect those of the entire community and this can introduce error into the study's findings.

Carcinogenesis – Carcinogenesis describes the process of the progression of normal cells to cancerous cells.

Causation or cause – A cause is an exposure or condition of the individual that has been proven through a sound weight-of-evidence review to increase risk of a disease.

Cause-and-effect relationship – A cause-and-effect relationship between an exposure and a disease is a statistically significant association that is determined through a weight-of-evidence review to be causal in nature.

Case-control study – A case-control study compares persons without a disease (controls) to persons with a disease (cases) to see if they differ on any factors or exposures of interest.

Chance – Chance refers to random sampling variation, like a coincidence. An association can be observed between an exposure and disease that is simply the result of a chance occurrence.

Cohort study – A cohort study follows a group of people over a long period of time to observe whether the occurrence of disease differs among exposed and unexposed persons in the group.

Confidence interval – A confidence interval is a range of values for an estimate of effect that has a specified probability (e.g., 95%) of including the "true" estimate of effect. A 95% confidence interval indicates that, if the study were conducted a very large number of times, 95% of the measured estimates would be within the upper and lower confidence limits.

Confounding – Confounding is a situation in which an association is distorted because the exposure is associated with other risk factors for the disease. For example, a link between coffee drinking in mothers and low birth weight babies has been reported in the past. However, some women who drink coffee also smoke cigarettes. It was found that when the smoking habits of the mothers are taken into account, coffee drinking was not associated with low birth weight babies because of the confounding effect of smoking.

Dose-response assessment/relationship – Data from scientific research in which a change in amount, intensity, or duration of exposure is associated with a change in risk of a specified outcome. A pattern of a stronger association with increasing exposure, or dose.

Electric field – The electric field is a property of a location or point in space and its electrical environment, and describes the forces that would be experienced by a charged body in that space by virtue of its charge. The electric field is expressed in measurement units of volts per meter (V/m) or kilovolts per meter (kV/m); a kilovolt per meter is equal to 1,000 V/m.

Electromagnetic spectrum – The range of wavelengths of electromagnetic energy, including visible light, arranged by frequency. Wavelength decreases with increasing frequency; the ELF range includes the power frequencies of 50/60-Hz.

Electromagnetic hypersensitivity – Self-reported responses to or perception of electromagnetic fields, including ELF-EMF, at levels far below exposure limits that may include a wide range symptoms, including sleep disturbances, general fatigue, difficulty in concentrating, dizziness, and eyestrain.

Epidemiology – The study of the frequency and distribution of disease and health events in human populations and the factors that contribute to disease and health events.

Exposure assessment – The step in risk assessment that characterizes the exposure circumstances of the situation under analysis.

Extremely low frequency (ELF) fields – Extremely low frequency refers to electromagnetic fields in the range of 0-300 Hz.

Hazard identification – The identification of adverse effects on health from a specific exposure based on a weight-of-evidence review of the scientific research.

In vitro – Laboratory studies of isolated cells that are artificially maintained in test tubes or culture dishes are called *in vitro* studies, literally "in glass." Researchers expose isolated cells or groups of cells (tissues) to a specific agent under controlled conditions. These studies help explain the mechanisms by which exposures might affect biological processes.

In vivo – Studies in living animals or experimental studies of processes in whole living organisms are called *in vivo* studies. Scientists expose laboratory animals to a specific agent under controlled conditions and look for effects on body function, measures of health, or disease. Experience has shown that effects in laboratory animals can help to predict effects that occur in people.

Initiation – The first stage in the development of cancer, initiation typically results from exposure to an agent that can cause mutations in a cell. Initiation is believed to be irreversible, and increases the likelihood of cancer occurring.

Job-exposure matrix – A job-exposure matrix cross-classifies job titles and exposure estimates. Job-exposure matrices are used to estimate cumulative occupational exposure (e.g., magnetic field exposure) based on an individual's job history.

Magnetic fields – The magnetic field is a state of region in space, and describes the forces that would be experienced by a moving charge (or magnetic material) in proportion to its charge and velocity. The strength of magnetic fields is expressed as magnetic flux density in units called gauss (G), or in milligauss (mG), where 1 G = 1,000 mG.

Meta-analysis – An analytic technique that combines the results of many studies into one summary estimate of the association between a particular exposure and disease.

Nested case-control study – A case-control study in which the cases and controls are drawn from a cohort study's population.

Odds ratio – An odds ratio is a measure of association that describes the ratio of the odds of exposure among persons with a disease to the odds of exposure among persons without a disease. For example, an odds ratio of two would suggest that persons with the disease are two times more likely to have had exposure than persons without the disease.

Pooled analysis – A pooled analysis combines individual-level data across many studies and analyzes the data together to get a summary estimate of the association between a particular exposure and disease.

Precautionary principle – The precautionary principle refers to the idea that, when evidence does not support the suggestion that an exposure is a cause of a particular disease but where a risk is perceived, precautionary measures may be taken that are proportional to the perceived level of risk, with science as the basis for measuring that risk.

Promotion – Promotion is a later stage in cancer development, following initiation. If there is sufficient exposure to the agent, promoters increase the frequency of tumor formation that occurs after initiation.

Reference level – The reference level is a measurable level of electric or magnetic field outside of the body that is used as a screening value. It is a practical measure to determine whether the internal level identified as the basic restriction is likely to be exceeded.

Relative risk – A relative risk is an estimate that compares the risk of disease among persons who are exposed to the risk of disease among persons who are unexposed. For example, a relative risk of two means that that exposed persons in the study is two times more likely to develop the disease than unexposed persons.

Risk characterization – A quantitative estimation of the likelihood of adverse effects that may result from exposure to a specific agent in a specific situation.

Safety factor – A multiplicative factor (usually less than 1.0) incorporated into risk assessments or safety standards to allow for unpredictable types of variation, such as variability in responses from test animals to humans or person-to-person variability.

Selection bias – Selection bias occurs when there are differences in the type of person who participates in the study compared to the type of person who doesn't participate in the study.

Selection bias introduces systematic error into a study, and limits the conclusions and generalizations that can be drawn.

Spot measurement – A spot measurement is an instantaneous magnetic or electric field reading that is taken at one location as an estimate of exposure.

Statistically significant – An association is statistically significant if one can conclude (with an established level of confidence using standard statistical tests) that the association is not due to a chance occurrence.

Time-weighted average (TWA) - The average exposure over a given specified time period (i.e., an 8-hr workday or a 24-hr day) of a person's exposure to a chemical or physical agent. The average is determined by sampling the exposure of interest throughout the time period.

Voltage – Voltage is the difference in electric potential between any two conductors of a circuit. It is the electric 'pressure' that exists between two points and is capable of producing the flow of current through an electrical conductor.

Weight-of-evidence review – A weight-of-evidence review critically evaluates the strength of the evidence for causality for a particular exposure and disease. It entails a comprehensive assessment of *all* relevant scientific research, in which each of the studies is critically evaluated, and more weight is given to studies of better quality.

Wire code categories – Wire coding categories are based on a classification system of homes using characteristics of power lines outside the home (e.g., thickness of the wires) and their distances from the home. This information is used to code the homes into categories based on their predicted magnetic field level.

References

Ahlbom A, Day N, Feychting M, Roman E, Skinner J, Dockerty J, Linet M, McBride M, Michaelis J, Olsen JH, Tynes T, Verkasalo PK. A pooled analysis of magnetic fields and childhood leukaemia. Br J Cancer 83: 692-8, 2000.

Ahmed AS, Patel PJ, Bagga S, Gilge JL, Schleeter T, Lakhani BA, Ravichandran AK, Donnelley S, Allavatam V, Prystowsky EN, Padanilam BJ. Troubleshooting electromagnetic interference in a patient with centrifugal flow left ventricular assist device and subcutaneous implantable cardioverter defibrillator. J Cardiovasc Electrophysiol 29: 477-481, 2017.

Akdag MZ, Dasdag S, Cakir DU, Yokus B, Kizil G, Kizil M. Do 100- and 500-muT ELF magnetic fields alter beta-amyloid protein, protein carbonyl and malondialdehyde in rat brains? Electromagn Biol Med 32: 363-72, 2013.

Alcaraz M, Olmos E, Alcaraz-Saura M, Achel DG, Castillo J. Effect of long-term 50 Hz magnetic field exposure on the micronucleated polychromatic erythrocytes of mice. Electromagn Biol Med 33: 51-7, 2014.

AlRahabi MK and Ghabbani HM. Influence and safety of electronic apex locators in patients with cardiovascular implantable electronic devices: a systematic review. Libyan J Med 14:1547071, 2018.

American Conference of Governmental Industrial Hygienists (ACGIH). 2015 TLVs and BEIs: Based on the Documentation of the Threshold Limit Values for Chemical Substances and Physical Agents & Biological Exposure Indices. Cincinnati: ACGIH Signature Publications, 2015.

Amoon AT, Crespi CM, Ahlbom A, Bhatnagar M, Bray I, Bunch KJ, Clavel J, Feychting M, Hemon D, Johansen C, Kreis C, Malagoli C, Marquant F, Pedersen C, Raaschou-Nielsen O, Röösli M, Spycher BD, Sudan M, Swanson J, Tittarelli A, Tuck DM, Tynes T, Vergara X, Vinceti M, Wunsch-Filho V, Kheifets L. Proximity to overhead power lines and childhood leukaemia: an international pooled analysis. Br J Cancer 119: 364-373, 2018a.

Amoon AT, Oksuzyan S, Crespi CM, Arah OA, Cockburn M, Vergara X, Kheifets L. Residential mobility and childhood leukemia. Environ Res 164: 459-466, 2018b.

Anderson LE, Boorman GA, Morris JE, Sasser LB, Mann PC, Grumbein SL, Hailey JR, McNally A, Sills RC, Haseman JK. Effect of 13 week magnetic field exposures on DMBAinitiated mammary gland carcinomas in female Sprague-Dawley rats. Carcinogenesis 20: 1615-20, 1999.

Andrianome S, Gobert J, Hugueville L, Stéphan-Blanchard E, Telliez F, Selmaoui B. An assessment of the autonomic nervous system in the electrohypersensitive population: a heart rate variability and skin conductance study. J Appl Physiol 123 :1055-1062, 1985, 2017. doi: 10.1152/japplphysiol.00229.2017.

Armstrong BG, Deadman J, McBride ML. The determinants of Canadian children's personal exposures to magnetic fields. Bioelectromagnetics 22: 161-9, 2001.

Astridge PS, Kaye GC, Whitworth S, Kelly P, Camm AJ, Perrins EJ. The response of implanted dual chamber pacemakers to 50 Hz extraneous electrical interference. Pacing Clin Electrophysiol 16: 1966-74, 1993.

Bailey, WH and Wagner, ME. IARC evaluation of ELF magnetic fields: Public understanding of the 0.4- μ T exposure metric. J Expo Sci Environ Epidemiol 18: 233-235, 2008.

Baldi I, Coureau G, Jaffre A, Gruber A, Ducamp S, Provost D, Lebailly P, Vital A, Loiseau H, Salamon R. Occupational and residential exposure to electromagnetic fields and risk of brain tumors in adults: a case-control study in Gironde, France. Int J Cancer 129: 1477-84, 2011.

Baum A, Mevissen M, Kamino K, Mohr U, Löscher W. A histopathological study on alterations in DMBA-induced mammary carcinogenesis in rats with 50 Hz, 100 mT magnetic field exposure. Carcinogenesis 16:119-125, 1995.

Berg H, Gunther B, Hilger I, Radeva M, Traitcheva N, Wollweber L. Bioelectromagnetic field effects on cancer cells and mice tumors. Electromagn Biol Med 29: 132-43, 2010.

Bernard N, Alberdi AJ, Tanguy ML, Brugere H, Helissey P, Hubert C, Gendrey N, Guillosson JJ, Nafziger J. Assessing the potential leukemogenic effects of 50 Hz magnetic fields and their harmonics using an animal leukemia model. J Radiat Res (Tokyo) 49: 565-77, 2008.

Boehmert C, Verrender A, Pauli M, Wiedemann P. Does precautionary information about electromagnetic fields trigger nocebo responses? An experimental risk communication study. Environ Health. 17:36, 2018. doi: 10.1186/s12940-018-0377-y.

Boorman GA, Anderson LE, Morris JE, Sasser LB, Mann PC, Grumbein SL, Hailey JR, McNally A, Sills RC, Haseman JK. Effect of 26 week magnetic field exposures in a DMBA initiation-promotion mammary gland model in Sprague-Dawley rats. Carcinogenesis 20: 899-904, 1999a.

Boorman GA, McCormick DL, Findlay JC, Hailey JR, Gauger JR, Johnson TR, Kovatch RM, Sills RC, Haseman JK. Chronic toxicity/oncogenicity evaluation of 60 Hz (power frequency) magnetic fields in F344/N rats. Toxicol Pathol 27: 267-78, 1999b.

Bua L, Tibaldi E, Falcioni L, Lauriola M, De Angelis L, Gnudi F, Manservigi M, Manservisi F, Manzoli I, Menghetti I, Montella R, Panzacchi S, Sgargi D, Strollo V, Vornoli A, Mandrioli D, Belpoggi F. Results of lifespan exposure to continuous and intermittent extremely low frequency electromagnetic fields (ELFEMF) administered alone to Sprague Dawley rats. Environ Res 164: 271-279, 2018.

Buffler PA, Kwan ML, Reynolds P, Urayama KY. Environmental and genetic risk factors for childhood leukemia: appraising the evidence. Cancer Invest 23: 60-75, 2005.

Bunch KJ, Keegan TJ, Swanson J, Vincent TJ, Murphy MF. Residential distance at birth from overhead high-voltage powerlines: childhood cancer risk in Britain 1962-2008. Br J Cancer 110: 1402-8, 2014.

Bunch KJ, Swanson J, Vincent TJ, Murphy MF. Epidemiological study of power lines and childhood cancer in the UK: further analyses. J Radiol Prot 36: 437-455, 2016.

Bunch KJ, Swanson J, Vincent TJ, Murphy MF. Magnetic fields and childhood cancer: an epidemiological investigation of the effects of high-voltage underground cables. J Radiol Prot 35: 695-705, 2015.

Carlberg M, Soderqvist F, Hansson Mild K, Hardell L. Meningioma patients diagnosed 2007-2009 and the association with use of mobile and cordless phones: a case-control study. Environ Health 12: 60, 2013.

Carlberg M and Hardell L. Pooled analysis of Swedish case-control studies during 1997-2003 and 2007-2009 on meningioma risk associated with the use of mobile and cordless phones. Oncol Rep 33: 3093-3098, 2015.

Carlberg M, Koppel T, Ahonen M, Hardell L. Case-control study on occupational exposure to extremely low-frequency electromagnetic fields and glioma risk. Am J Ind Med 60: 494-503, 2017.

Carlberg M, Koppel T, Ahonen M, Hardell L. Case-Control Study on Occupational Exposure to Extremely Low-Frequency Electromagnetic Fields and the Association with Meningioma. Biomed Res Int 2018: 5912394, 2018.

Chang ET, Adami HO, Bailey WH, Boffetta P, Krieger RI, Moolgavkar SH, Mandel JS. Validity of geographically modeled environmental exposure estimates. Crit Rev Toxicol. 44: 450-66, 2014.

Checkoway H, Ilango S, Li W, Ray RM, Tanner CM, Hu SC, Wang X, Nielsen S, Gao DL, Thomas DB. Occupational exposures and parkinsonism among Shanghai women textile workers. Am J Ind Med 61(11):886-892, 2018.

Chung MK, Kim YB, Ha CS, Myung SH. Lack of a co-promotion effect of 60 Hz rotating magnetic fields on N-ethyl-N-nitrosourea induced neurogenic tumors in F344 rats. Bioelectromagnetics 29: 539-48, 2008.

Chung MK, Yu WJ, Kim YB, Myung SH. Lack of a co-promotion effect of 60 Hz circularly polarized magnetic fields on spontaneous development of lymphoma in AKR mice. Bioelectromagnetics 31: 130-9, 2010.

Claeson AS, Palmquist E, Nordin S. Physical and chemical trigger factors in environmental intolerance. Int J Hyg Environ Health. Apr;221 :586-592, 2018. doi: 10.1016/j.ijheh.2018.02.009.

Coble JB, Dosemeci M, Stewart PA, Blair A, Bowman J, Fine HA, Shapiro WR, Selker RG, Loeffler JS, Black PM, Linet MS, Inskip PD. Occupational exposure to magnetic fields and the risk of brain tumors. Neuro Oncol 11: 242-9, 2009.

Collison C. A closer look: sound generation and hearing. Bee Culture. February 27, 2016.

Council of the European Union (CEU). Council recommendation of 12 July 1999 on the limitation of exposure of the general public to electromagnetic fields (0 Hz to 300 GHz) - 1999/519/EC. Official Journal of the European Communities L199: 59-70, 1999.

Crespi CM, Vergara XP, Hooper C, Oksuzyan S, Wu S, Cockburn M, Kheifets L. Childhood leukaemia and distance from power lines in California: a population-based case-control study. Br J Cancer 115: 122-8, 2016.

Cui Y, Ge Z, Rizak JD, Zhai C, Zhou Z, Gong S, Che Y. Deficits in water maze performance and oxidative stress in the hippocampus and striatum induced by extremely low frequency magnetic field exposure. PLoS One 7: e32196, 2012.

Dadalti MTS, da Cunha A, Araujo MCP, Moraes LGB, Risso PA. Electromagnetic interference of dental equipment with implantable cardioverter defibrillators. Acta Odontol Scand 75: 584-587, 2017.

De Bont R and van Larebeke N. Endogenous DNA damage in humans: a review of quantitative data. Mutagenesis 19: 169-85, 2004.

de Rotte AA, van der Kemp P, Mundy PA, Rienks R, de Rotte AA. Electromagnetic Interference in Implantable Defibrillators in Single-Engine Fixed-Wing Aircraft. Aerosp Med Hum Perform 88: 52-55, 2017.

Deadman JE, Armstrong BG, McBride ML, Gallagher R, Theriault G. Exposures of children in Canada to 60-Hz magnetic and electric fields. Scand J Work Environ Health 25: 368-75, 1999.

Deng Y, Zhang Y, Jia S, Liu J, Liu Y, Xu W, Liu L. Effects of aluminum and extremely low frequency electromagnetic radiation on oxidative stress and memory in brain of mice. Biol Trace Elem Res 156: 243-52, 2013.

Dömötör Z, Szemerszky R, Köteles F. Nature relatedness is connected with modern health worries and electromagnetic hypersensitivity. J Health Psychol. 1:1359105317699681, 2017. doi: 10.1177/1359105317699681.

Duan Y, Wang Z, Zhang H, He Y, Lu R, Zhang R, Sun G, Sun X. The preventive effect of lotus seedpod procyanidins on cognitive impairment and oxidative damage induced by extremely low frequency electromagnetic field exposure. Food Funct. 4:1252-62, 2013. doi: 10.1039/c3fo60116a.

Dyrda K, Khairy P. Implantable rhythm devices and electromagnetic interference: myth or reality? Expert Rev Cardiovasc Ther 6: 823-32, 2008.

El-Bialy NS and Rageh MM. Extremely low-frequency magnetic field enhances the therapeutic efficacy of low-dose cisplatin in the treatment of Ehrlich carcinoma. Biomed Res Int 2013: 189352, 2013.

Eltiti S, Wallace D, Russo R, Fox E. Symptom presentation in idiopathic environmental intolerance with attribution to electromagnetic fields: Evidence for a nocebo effect based on data re-analyzed from two previous provocation studies. Front Psychol. 9:1563, 2018. doi: 10.3389/fpsyg.2018.01563.

European Committee for Electrotechnical Standardization (CENELEC). Procedure for the assessment of the exposure to electromagnetic fields of workers bearing active implantable medical devices - Part 1: General (EN 50527-1). Brussels, Belgium: CENELEC, 2010

European Health Risk Assessment Network on Electromagnetic Fields Exposure (EFHRAN). Report on the Analysis of Risks Associated to Exposure to EMF: *In Vitro* and *In Vivo* (Animals) Studies. Milan, Italy 2010.

European Health Risk Assessment Network on Electromagnetic Fields Exposure (EFHRAN). Risk Analysis of Human Exposure to Electromagnetic Fields (Revised). Report D2 of the EFHRAN Project. Milan, Italy: EFHRAN, 2012.

Exponent, Inc. Electric and Magnetic Fields and Health: Review of the Scientific Research from March 1, 2012, to December 31, 2016. Prepared for BC Hydro, Inc., Exponent, 2017.

Exponent, Inc. EMF and Health – Comprehensive Review and Update of the Scientific Research, January 15, 2010 through March 1, 2012. Prepared for BC Hydro, Inc., Exponent, 2012.

Exponent, Inc. EMF and Health – Review and Update of the Scientific Research. Prepared for BC Hydro, Inc., Exponent, 2007.

Exponent, Inc. EMF and Health – Review and Update of the Scientific Research, September 2007 through January 2010. Prepared for BC Hydro, Inc., Exponent, 2010.

Fam WZ, Mikhail EL. Lymphoma induced in mice chronically exposed to very strong low-frequency electromagnetic field. Cancer Lett 105: 257-69, 1996.

Federal-Provincial-Territorial Radiation Protection Committee (FPTRPC). Health effects and exposure guidelines related to extremely low frequency (ELF) 50/60 Hz electric and magnetic fields – an overview. Prepared by the ELF Working Group. Ottawa: Health Canada, 1998.

Federal-Provincial-Territorial Radiation Protection Committee (FPTRPC). 2008. "Response Statement to Public Concerns Regarding Electric and Magnetic Fields (EMFs) from Electrical Power Transmission and Distribution Lines." Accessed January 24, 2017. http://www.hcsc.gc.ca/ewh-semt/radiation/fpt-radprotect/emf-cem-eng.php. Federal-Provincial-Territorial Radiation Protection Working Group. Health Effects and Exposure Guidelines Related to Extremely Low Frequency Electric and Magnetic Fields - An Overview. Ottawa: FPTRPC Committee, 2005.

Fedrowitz M and Löscher W. Exposure of Fischer 344 rats to a weak power frequency magnetic field facilitates mammary tumorigenesis in the DMBA model of breast cancer. Carcinogenesis 29: 186-93, 2008.

Fedrowitz M and Löscher W. Gene expression in the mammary gland tissue of female Fischer 344 and Lewis rats after magnetic field exposure (50 Hz, 100 muT) for 2 weeks. Int J Radiat Biol 88: 425-9, 2012.

Glinka M, Sieroń A, Birkner E, Cieślar G. Influence of extremely low-frequency magnetic field on the activity of antioxidant enzymes during skin wound healing in rats. Electromagn Biol Med 32:463-70, 2013.

Gobba F, Bravo G, Rossi P, Contessa GM, Scaringi M. Occupational and environmental exposure to extremely low frequency-magnetic fields: a personal monitoring study in a large group of workers in Italy. J Expo Sci Environ Epidemiol 21: 634-645, 2011.

Government of Canada, Privy Council Office. A framework for the application of precaution in science-based decision making about risk. Ottawa: Privy Council Office, 2003.

Greenland S and Kheifets L. Leukemia attributable to residential magnetic fields: results from analyses allowing for study biases. Risk Anal 26: 471-482, 2006.

Greenland S, Sheppard AR, Kaune WT, Poole C, Kelsh MA. A pooled analysis of magnetic fields, wire codes, and childhood leukemia. Childhood Leukemia-EMF Study Group. Epidemiology 11: 624-34, 2000.

Gruber MJ, Palmquist E, Nordin S. Characteristics of perceived electromagnetic hypersensitivity in the general population. Scand J Psychol. 59 :422-427, 2018. doi: 10.1111/sjop.12449.

Halgamuge MN, Abeyrathne CD, Mendis P. Measurement and analysis of electromagnetic fields from trams, trains and hybrid cars. Radiat Prot Dosimetry 141:255-268, 2010.

Hardell L, Carlberg M, Hansson Mild K. Pooled analysis of two case-control studies on use of cellular and cordless telephones and the risk for malignant brain tumours diagnosed in 1997-2003. Int Arch Occup Environ Health 79: 630-639, 2006.

Hardell L, Carlberg M, Soderqvist F, Mild KH. Case-control study of the association between malignant brain tumours diagnosed between 2007 and 2009 and mobile and cordless phone use. Int J Oncol 43: 1833-1845, 2013.

Harris AW, Basten A, Gebski V, Noonan D, Finnie J, Bath ML, Bangay MJ, Repacholi MH. A test of lymphoma induction by long-term exposure of E mu-Pim1 transgenic mice to 50 Hz magnetic fields. Radiat Res 149:300-307, 1998.

Hassan NS, Abdelkawi SA. Assessing of plasma protein denaturation induced by exposure to cadmium, electromagnetic fields and their combined actions on rat. Electromagn Biol Med 33: 147-53, 2014.

Health Canada. Health Canada Decision-Making Framework for Identifying, Assessing, and Managing Health Risk. Ottawa, Canada: Health Canada, 2000.

Health Canada. Human Health Risk Assessment for Priority Substances. Environmental Health Directorate. Canadian Environmental Protection Act. Ottawa: Health Canada, 1994.

Health Canada. Limits of Human Exposure to Radiofrequency Electromagnetic Energy in the Frequency Range from 3 kHz to 300 GHz. Safety Code 6 (2015). Ottawa, Ontario: Health Canada, 2015.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Electromagnetic Fields: Annual Update 2001. Publication No. 2001/14. The Hague: Health Council of the Netherlands, 2001.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Electromagnetic Fields: Annual Update 2003. Publication No. 2004/01. The Hague: Health Council of the Netherlands, 2004.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Electromagnetic Fields: Annual Update 2005. Publication No. 2005/14. The Hague: Health Council of the Netherlands, 2005.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Electromagnetic Fields: Annual Update 2006. Publication No. 2007/06. The Hague: Health Council of the Netherlands, 2007.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Electromagnetic Fields: Annual Update. Publication No. 2009/02. The Hague: Health Council of the Netherlands, 2009a.

Health Council of the Netherlands (HCN) ELF Electromagnetic Fields Committee. Advisory letter - Power lines and Alzheimer's disease to the Minister of Housing, Spatial Planning and the Environment. The Hague: Health Council of the Netherlands, 2009b.

Health Protection Agency (HPA). Power frequency electromagnetic fields, melatonin and the risk of breast cancer: report of an independent advisory group on non-ionising radiation. Documents of the Health Protection Agency. Series B: Radiation, Chemical and Environmental Hazards. Oxfordshire: Health Protection Agency, 2006.

Her Majesty's Government (HMG). Government response to the stakeholder advisory group on extremely low frequency electric and magnetic fields (ELF EMFs) (SAGE) recommendations. 2009.

Hill AB. The Environment and Disease: Association or Causation? Proc R Soc Med 58: 295-300, 1965.

Huang PC, Cheng MT, Guo HR. Representative survey on idiopathic environmental intolerance attributed to electromagnetic fields in Taiwan and comparison with the international literature. Environ Health. 17:5, 2018a. doi: 10.1186/s12940-018-0351-8.

Huang PC, Li KH, Guo HR. Association between media coverage and prevalence of idiopathic environmental intolerance attributed to electromagnetic field in Taiwan. Environ Res. 161:329-335, 2018b. doi: 10.1016/j.envres.2017.11.034.

Huss A, Koeman T, Kromhout H, Vermeulen R. Extremely Low Frequency Magnetic Field Exposure and Parkinson's Disease--A Systematic Review and Meta-Analysis of the Data. Int J Environ Res Public Health 12: 7348-56, 2015.

Huss A, Spoerri A, Egger M, Röösli M. Residence near power lines and mortality from neurodegenerative diseases: longitudinal study of the Swiss population. Am J Epidemiol 169: 167-75, 2009.

Huss A, Spoerri A, Egger M, Kromhout H, Vermeulen R. Occupational extremely low frequency magnetic fields (ELF-MF) exposure and hematolymphopoietic cancers - Swiss National Cohort analysis and updated meta-analysis. Environ Res 164:467-474, 2018a.

Huss A, Peters S, Vermeulen R. Occupational exposure to extremely low-frequency magnetic fields and the risk of ALS: A systematic review and meta-analysis. Bioelectromagnetics 39: 156-163, 2018b.

Jalilian H, Teshnizi SH, Rösli M, Neghab M. Occupational exposure to extremely low frequency magnetic fields and risk of Alzheimer disease: A systematic review and metaanalysis. Neurotoxicology 69: 242-252, 2017.

Johansen C. Exposure to electromagnetic fields and risk of central nervous system disease in utility workers. Epidemiology 11(5):539-43, 2000.

International Agency for Research on Cancer (IARC). IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. Volume 80: Static and Extremely Low-Frequency (ELF) Electric and Magnetic Fields. Lyon, France: IARC Press, 2002.

International Agency for Research on Cancer (IARC). IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. Preamble. Lyon, France: IARC, Press, 2006.

International Agency for Research on Cancer (IARC). Mechanisms for Carcinogenesis in Risk Identification. Lyon, France: IARC Press, 1992.

International Commission for Non-Ionizing Radiation Protection (ICNIRP). ICNIRP Statement. General Approach to Protection Against Non-Ionizing Radiation. Health Phys 82: 540-548, 2002.

International Commission on Non-Ionizing Radiation Protection (ICNIRP). Guidelines for limiting exposure to time-varying electric, magnetic, and electromagnetic fields (up to 300 GHz). Health Phys 74: 494-522, 1998.

International Commission on Non-Ionizing Radiation Protection (ICNIRP). Exposure to Static and Low Frequency Electromagnetic Fields, Biological Effects and Health Consequences (0-100 kHz) – Review of the Scientific Evidence on Dosimetry, Biological Effects, Epidemiological Observations, and Health Consequences Concerning Exposure to Static and Low Frequency Electromagnetic Fields (0-100 kHz). Matthes R, McKinlay AF, Bernhardt JH, Vecchia P, Beyret B (eds.). Munich: ICNIRP, 2003.

International Commission on Non-ionizing Radiation Protection (ICNIRP). Guidelines for limiting exposure to time-varying electric and magnetic fields (1 Hz to 100 kHz). Health Phys 99: 818-36, 2010.

International Committee on Electromagnetic Safety (ICES). IEEE Standard for Safety Levels with Respect to Human Exposure to Electromagnetic Fields 0 to 3 kHz. Piscataway, NJ: IEEE, 2002.

Jiménez-Garcia MN, Robledo JA, Aparicio-Bautists DI, Rodriguez-Segura MA, Villa-Trevino S, Godina-Nava JJ. Anti-proliferative effect of extremely low frequency electromagnetic field on preneoplastic lesions formation in the rat liver. BMC Cancer 10: 1-12, 2010.

Johansen C, Raaschou Nielsen O, Olsen JH, Schüz J. Risk for leukaemia and brain and breast cancer among Danish utility workers: a second follow-up. Occup Environ Med 64: 782-4, 2007.

Joosten S, Pammler K, Silny J. The influence of anatomical and physiological parameters on the interference voltage at the input of unipolar cardiac pacemakers in low frequency electric fields. Phys Med Biol 54: 591-609, 2009.

Karvala K, Sainio M, Palmquist E, Claeson AS, Nyback MH, Nordin S. Building-Related Environmental Intolerance and Associated Health in the General Population. Int J Environ Res Public Health. 15: pii: E2047, 2018. doi: 10.3390/ijerph15092047.

Kheifets L, Ahlbom A, Crespi CM, Draper G, Hagihara J, Lowenthal RM, Mezei G, Oksuzyan S, Schüz J, Swanson J, Tittarelli A, Vinceti M, Wunsch Filho V. Pooled analysis of recent studies on magnetic fields and childhood leukaemia. Br J Cancer 103: 1128-35, 2010a.

Kheifets L, Ahlbom A, Crespi CM, Feychting M, Johansen C, Monroe J, Murphy MF, Oksuzyan S, Preston-Martin S, Roman E, Saito T, Savitz D, Schüz J, Simpson J, Swanson J, Tynes T, Verkasalo P, Mezei G. A pooled analysis of extremely low-frequency magnetic fields and childhood brain tumors. Am J Epidemiol 172: 752-61, 2010b.

Kheifets L, Bowman JD, Checkoway H, Feychting M, Harrington JM, Kavet R, Marsh G, Mezei G, Renew DC, van Wijngaarden E. Future needs of occupational epidemiology of extremely low frequency electric and magnetic fields: review and recommendations. Occup Environ Med 66: 72-80, 2009.

Kheifets L, Crespi CM, Hooper C, Cockburn M, Amoon AT, Vergara XP. Residential magnetic fields exposure and childhood leukemia: a population-based case-control study in California. Cancer Causes Control 28: 1117-1123, 2017a.

Kheifets L, Swanson J, Yuan Y, Kusters C, Vergara X. Comparative analyses of studies of childhood leukemia and magnetic fields, radon and gamma radiation. J Radiol Prot 37: 459-491, 2017b.

Kheifets L, Monroe J, Vergara X, Mezei G, Afifi AA. Occupational electromagnetic fields and leukemia and brain cancer: an update to two meta-analyses. J Occup Environ Med 50: 677-88, 2008.

Kheifets L, Swanson J, Greenland S. Childhood leukemia, electric and magnetic fields, and temporal trends. Bioelectromagnetics 27: 545-52, 2006.

Kheifets L1, Bowman JD, Checkoway H, Feychting M, Harrington JM, Kavet R, Marsh G, Mezei G, Renew DC, van Wijngaarden E.

Kheifets LI, Afifi AA, Buffler PA, Zhang ZW. Occupational electric and magnetic field exposure and brain cancer: a meta-analysis. J Occup Environ Med 37: 1327-41, 1995.

Kiray A1, Tayefi H, Kiray M, Bagriyanik HA, Pekcetin C, Ergur BU, Ozogul C. The effects of exposure to electromagnetic field on rat myocardium. Toxicol Ind Health 29:418-25, 2013.

Kirschvink J, Padmanabha S, Boyce C, Oglesby J. Measurement of the threshold sensitivity of honeybees to weak, extremely low-frequency magnetic fields. J Exp Biol. 200:1363-8, 1997.

Kirschvink JL. Uniform magnetic fields and double-wrapped coil systems: improved techniques for the design of bioelectromagnetic experiments. Bioelectromagnetics. 13:401-11, 1992.

Korpinen L, Kuisti H, Elovaara J, Virtanen V. Implantable cardioverter defibrillators in electric and magnetic fields of 400 kV power lines. Pacing Clin Electrophysiol 37: 297-303, 2014.

Koeman T, Slottje P, Schouten LJ, Peters S, Huss A, Veldink JH, Kromhout H, van den Brandt PA, Vermeulen R. Occupational exposure and amyotrophic lateral sclerosis in a prospective cohort. Occup Environ Med 74: 578-585, 2017.

Korr H, Angstman NB, Born TB, Bosse K, Brauns B, Demmler M, Fueller K, Kántor O, Kever BM, Rahimyar N, Salimi S, Silny J, Schmitz C. No evidence of persisting unrepaired nuclear DNA single strand breaks in distinct types of cells in the brain, kidney, and liver of adult mice after continuous eight-week 50 Hz magnetic field exposure with flux density of 0.1 mT or 1.0 mT. PLoS One 9:e109774, 2014.

Köteles F, Szemerszky R, Gubányi M, Körmendi J, Szekrényesi C, Lloyd R, Molnár L, Drozdovszky O, Bárdos G. Idiopathic environmental intolerance attributed to electromagnetic fields (IEI-EMF) and electrosensibility (ES) - are they connected? Int J Hyg Environ Health 216(3):362-70, 2013.

Kyriakopoulou A, Meimeti E, Moisoglou I, Psarrou A, Provatopoulou X, Dounias G. Parental Occupational Exposures and Risk of Childhood Acute Leukemia. Mater Sociomed. 30(3):209-214, 2018.

Lai H and Singh NP. Magnetic-field-induced DNA strand breaks in brain cells of the rat. Environ Health Perspect 112: 687-94, 2004.

Lee GM, Neutra RR, Hristova L, Yost M, Hiatt RA. A nested case-control study of residential and personal magnetic field measures and miscarriages. Epidemiology 13: 21-31, 2002.

Lennerz C, O'Connor M, Horlbeck L, Michel J, Weigand S, Grebmer C, Blazek P, Brkic A, Semmler V, Haller B, Reents T, Hessling G, Deisenhofer I, Whittaker P, Lienkamp M, Kolb C. Electric Cars and Electromagnetic Interference With Cardiac Implantable Electronic Devices: A Cross-sectional Evaluation. Ann Intern Med 169: 350-352, 2018.

Lewis RC, Evenson KR, Savitz DA, Meeker JD. Temporal variability of daily personal magnetic field exposure metrics in pregnant women. J Expo Sci Environ Epidemiol 25: 58-64, 2015.

Li DK, Odouli R, Wi S, Janevic T, Golditch I, Bracken TD, Senior R, Rankin R, Iriye R. A population-based prospective cohort study of personal exposure to magnetic fields during pregnancy and the risk of miscarriage. Epidemiology 13: 9-20, 2002.

Li DK, Chen H, Odouli R. Maternal Exposure to Magnetic Fields During Pregnancy in Relation to the Risk of Asthma in Offspring. Arch Pediatr Adolesc Med 165: 945-950, 2011.

Li DK, Chen H, Ferber JR, Odouli R, Quesenberry C. Exposure to Magnetic Field Non-Ionizing Radiation and the Risk of Miscarriage: A Prospective Cohort Study. Sci Rep 7: 17541, 2017.

Li H, Lin L, Li L, Zhou L, Hao S, Zhang Y, Ding Z. Eotaxin1 and MCP1 serve as circulating indicators in response to power frequency electromagnetic field exposure in mice. Mol Med Rep 18: 2832-2840, 2018.

Linet MS, Hatch EE, Kleinerman RA, Robison LL, Kaune WT, Friedman DR, Severson RK, Haines CM, Hartsock CT, Niwa S, Wacholder S, Tarone RE. Residential exposure to magnetic fields and acute lymphoblastic leukemia in children. N Engl J Med 337: 1-7, 1997.

Löscher W and Mevissen M. Linear relationship between flux density and tumor co-promoting effect of prolonged magnetic field exposure in a breast cancer model. Cancer Lett 96: 175-80, 1995.

Löscher W, Mevissen M, Haussler B. Seasonal influence on 7,12-dimethylbenz[a]anthraceneinduced mammary carcinogenesis in Spague-Dawley rats under controlled laboratory conditions. Pharmacol Toxicol 81: 265-270, 1997.

Löscher W, Mevissen M, Lehmacher W, Stamm A. Tumor promotion in a breast cancer model by exposure to a weak alternating magnetic field. Cancer Lett 71: 75-81, 1993.

Löscher W, Wahnschaffe U, Mevissen M, Lerchl A, Stamm A. Effects of weak alternating magnetic fields on nocturnal melatonin production and mammary carcinogenesis in rats. Oncology 51: 288-295, 1994.

Luo X, Chen M, Duan Y, Duan W, Zhang H, He Y, Yin C, Sun G, Sun X. Chemoprotective action of lotus seedpod procyanidins on oxidative stress in mice induced by extremely low-frequency electromagnetic field exposure. Biomed Pharmacother 82: 640-648, 2016.

Maes A and Verschaeve L. Genetic damage in humans exposed to extremely low-frequency electromagnetic fields. Arch Toxicol 90: 2337-48, 2016.

Magne I, Souques M, Bureau I, Duburcq A, Remy E, Lambrozo J. Exposure of children to extremely low frequency magnetic fields in France: Results of the EXPERS study. J Expo Sci Environ Epidemiol 27: 505-512, 2017.

Mahna A, Firoozabadi SM, Shankayi Z. The effect of ELF magnetic field on tumor growth after electrochemotherapy. J Membr Biol 247: 9-15, 2014.

Makinistian L, Muehsam DJ, Bersani F, Belyaev I. Some recommendations for experimental work in magnetobiology, revisited. Bioelectromagnetics. 39:556-564, 2018. doi: 10.1002/bem.22144. Epub 2018 Oct 10.

Mandeville R, Franco E, Sidrac-Ghali S, Paris-Nadon L, Rocheleau N, Mercier G, Desy M, Gaboury L. Evaluation of the potential carcinogenicity of 60 Hz linear sinusoidal continuous-wave magnetic fields in Fisher F344 rats. FASEB Journal 11:1127-1136, 1997.

Manikonda PK, Rajendra P, Devendranath D, Gunasekaran B, Channakeshava, Aradhya SR, Sashidhar RB, Subramanyam C. Extremely low frequency magnetic fields induce oxidative stress in rat brain. Gen Physiol Biophys 33: 81-90, 2014.

Marcilio I, Gouveia N, Pereira Filho ML, Kheifets L. Adult mortality from leukemia, brain cancer, amyotrophic lateral sclerosis and magnetic fields from power lines: a case-control study in Brazil. Rev Bras Epidemiol 14: 580-8, 2011.

Mariucci G, Villarini M, Moretti M, Taha E, Conte C, Minelli A, Aristei C, Ambrosini MV. Brain DNA damage and 70-kDa heat shock protein expression in CD1 mice exposed to extremely low frequency magnetic fields. Int J Radiat Biol 86: 701-10, 2010.

Martínez-Sámano J, Torres-Durán PV, Juárez-Oropeza MA, Verdugo-Díaz L. Effect of acute extremely low frequency electromagnetic field exposure on the antioxidant status and lipid levels in rat brain. Arch Med Res 43: 183-9, 2012.

McBride ML, Gallagher RP, Theriault G, Armstrong BG, Tamaro S, Spinelli JJ, Deadman JE, Fincham S, Robson D, Choi W. Power-frequency electric and magnetic fields and risk of childhood leukemia in Canada. Am J Epidemiol 149: 831-42, 1999.
McCormick DL, Boorman GA, Findlay JC, Hailey JR, Johnson TR, Gauger JR, Pletcher JM, Sills RC, and Haseman JK. Chronic toxicity/oncogenicity evaluation of 60 Hz (power frequency) magnetic fields in B6C3F1 mice. Toxicol Pathol 27:279-85, 1999.

McNally RJ and Parker L. Environmental factors and childhood acute leukemias and lymphomas. Leuk Lymphoma 47: 583-98, 2006.

Mee T, Whatmough P, Broad L, Dunn C, Maslanyj M, Allen S, Muir K, McKinney PA, van Tongeren M. Occupational exposure of UK adults to extremely low frequency magnetic fields. Occup Environ Med 66: 619-27, 2009.

Mevissen M, Haussler M, Lerchl A, Löscher W. Acceleration of mammary tumorigenesis by exposure of 7,12-dimethylbenz[a]anthracene-treated female rats in a 50-Hz, 100-microT magnetic field: replication study. J Toxicol Environ Health 53: 401-418, 1998.

Mevissen M, Lerchl A, Löscher W. Study on pineal function and DMBA-induced breast cancer formation in rats during exposure to a 100-mg, 50-Hz magnetic field. J Toxicol Environ Health 48: 169-185, 1996a.

Mevissen M, Lerchl A, Szamel M, Löscher W. Exposure of DMBA-treated female rats in a 50-Hz, 50 microTesla magnetic field: effects on mammary tumor growth, melatonin levels, and T lymphocyte activation. Carcinogenesis 17: 903-10, 1996b.

Mevissen M, Stamm A, Buntenkotter S, Zwingleberg R, Wahnschaffe U, Löscher W. Effects of magnetic fields on mmmary tumor development induced by 7,12-dimethylbenz(a)anthracene in rats. Bioelectromagnetics 14: 131-143, 1993a.

Mevissen M, Stamm A, Buntenkotter S, Zwingleberg R, Wahnschaffe U, Löscher W. Effects of AC Magnetic Field on DMBA-induced Mammary Carcinogenesis in Sprague-Dawley Rats. In Electricity and Magnetism in Biology and Medicine, edited by M. Blank. San Francisco, CA: San Francisco Press, 1993b.

Mezei G, Bracken TD, Senior R, Kavet R. Analyses of magnetic-field peak-exposure summary measures. J Expo Sci Environ Epidemiol 16: 477-85, 2006.

Mezei G, Gadallah M, Kheifets L. Residential magnetic field exposure and childhood brain cancer: a meta-analysis. Epidemiology 19: 424-30, 2008.

Migault L, Piel C, Carles C, Delva F, Lacourt A, Cardis E, Zaros C, de Seze R, Baldi I, Bouvier G. Maternal cumulative exposure to extremely low frequency electromagnetic fields and pregnancy outcomes in the Elfe cohort. Environ Int.112:165-173, 2018.

Miyakoshi Y, Kajihara C, Shimizu H, Yanagisawa H. Tempol suppresses micronuclei formation in astrocytes of newborn rats exposed to 50-Hz, 10-mT electromagnetic fields under bleomycin administration. Mutat Res 747: 138-41, 2012.

Napp A, Joosten S, Stunder D, Knackstedt C, Zink M, Bellmann B, Marx N, Schauerte P, Silny J. Electromagnetic interference with implantable cardioverter-defibrillators at power frequency: an in vivo study. Circulation. 129(4):441-50, 2014.

National Institute of Environmental Health Sciences (NIEHS). Assessment of Health Effects from Exposure to Power-line Frequency Electric and Magnetic Fields: Working Group Report. Research Triangle Park, NC: National Institute of Environmental Health Sciences of the U.S. National Institute of Health, 1998.

National Institute of Environmental Health Sciences (NIEHS). EMF Questions & Answsers (NIH Publication 02-4493). Research Triangle Park, NC: National Institute of Environmental Health Sciences of the U.S. National Institute of Health, 2002.

National Institute of Environmental Health Sciences (NIEHS). Health Effects from Exposure to Power Line Frequency Electric and Magnetic Fields. NIH Publication No. 99-4493. Research Triangle Park, NC: National Institute of Environmental Health Sciences of the U.S. National Institute of Health, 1999.

National Radiological Protection Board (NRPB). Electromagnetic Fields and the Risk of Cancer. Report of an Advisory Group on Non-ionizing Radiation. Volume 3, Number 1. Chilton, UK: NRPB, 1992.

National Radiological Protection Board (NRPB). Electromagnetic Fields and the Risk of Cancer. Supplementary Report by the Advisory Group on Non-ionising Radiation. Volume 5, Number 2. Chilton, UK: NRPB, 1994a.

National Radiological Protection Board (NRPB). Health Effects Related to the Use of Visual Display Units. Report by the Advisory Group on Non-ionising Radiation. Volume 5, Number 2. Chilton, UK: NRPB, 1994b.

National Radiological Protection Board (NRPB). ELF Electromagnetic Fields and the Risk of Cancer: Report of an Advisory Group on Non-ionising Radiation. Volume 12, No. 1. Chilton: NRPB, 2001a.

National Radiological Protection Board (NRPB). ELF Electromagnetic Fields and Neurodegenerative Disease. Volume 12, Number 4. Chilton, UK: NRPB, 2001b.

National Radiological Protection Board (NRPB). Review of the Scientific Evidence for Limiting Exposure to Electromagnetic Fields (0-300 GHz). Chilton, UK: NRPB, 2004.

National Research Council (NRC). Possible Health Effects of Exposure to Residential Electric and Magnetic Fields. Washington, D.C.: National Academy Press, 1997.

National Toxicology Program (NTP). Toxicology and Carcinogenesis Studies of 60-Hz Magentic Fields in F344/N Rats and B6C3F1 Mice (Whole Body Exposure Studies), NIH Publication No. 99-3978. Research Triangle Park, NC: US Department of Health and Human Services, Public Health Service, National Institutes of Health, 1999. National Toxicology Program (NTP). 2015. "Descriptions of NTP Study Types." U.S. Department of Health and Human Services. <u>https://ntp.niehs.nih.gov/testing/types/index.html</u>.

Negishi T, Imai S, Shibuya K, Nishimura I, Shigemitsu T. Lack of promotion effects of 50 Hz magnetic fields on 7,12-dimethylbenz(a)anthracene-induced malignant lymphoma/lymphatic leukemia in mice. Bioelectromagnetics 29: 29-38, 2008.

Okudan N, Celik I, Salbacak A, Cicekcibasi AE, Buyukmumcu M, Gokbel H. Effects of longterm 50 Hz magnetic field exposure on the micro nucleated polychromatic erythrocyte and blood lymphocyte frequency and argyrophilic nucleolar organizer regions in lymphocytes of mice. Neuro Endocrinol Lett 31: 208-14, 2010.

Pedersen C, Raaschou-Nielsen O, Rod NH, Frei P, Poulsen AH, Johansen C, Schuz J. Distance from residence to power line and risk of childhood leukemia: a population-based case-control study in Denmark. Cancer Causes Control 25: 171-7, 2014a.

Pedersen C, Brauner EV, Rod NH, Albieri V, Andersen CE, Ulbak K, Hertel O, Johansen C, Schuz J, Raaschou-Nielsen O. Distance to high-voltage power lines and risk of childhood leukemia--an analysis of confounding by and interaction with other potential risk factors. PLoS One 9: e107096, 2014b.

Pedersen C, Johansen C, Schuz J, Olsen JH, Raaschou-Nielsen O. Residential exposure to extremely low-frequency magnetic fields and risk of childhood leukaemia, CNS tumour and lymphoma in Denmark. Br J Cancer 113: 1370-4, 2015.

Pedersen C, Poulsen AH, Rod NH, Frei P, Hansen J, Grell K, Raaschou-Nielsen O, Schüz J, Johansen C. Occupational exposure to extremely low-frequency magnetic fields and risk for central nervous system disease: an update of a Danish cohort study among utility workers. Int Arch Occup Environ Health 90: 619-628, 2017.

Qi G, Zuo X, Zhou L, Aoki E, Okamula A, Watanebe M, Wang H, Wu Q, Lu H, Tuncel H, Watanabe H, Zeng S, Shimamoto F. Effects of extremely low-frequency electromagnetic fields (ELF-EMF) exposure on B6C3F1 mice. Environ Health Prev Med 20: 287-93, 2015.

Roberto ES, Aung TT, Hassan A, Wase A. Electromagnetic Interference from Swimming Pool Generator Current Causing Inappropriate ICD Discharges. Case Rep Cardiol 2017: 6714307, 2017.

Röösli M and Jalilian H. A meta-analysis on residential exposure to magnetic fields and the risk of amyotrophic lateral sclerosis. Rev Environ Health 33: 295-299, 2018.

Rothman KJ and Greenland S. Modern Epidemiology. Philadelphia: Lippencott-Raven Publishers, 1998.

Sadeghi T, Ahmadi A, Javadian M, Gholamian SA, Delavar MA, Esmailzadeh S, Ahmadi B, Hadighi MSH. Preterm birth among women living within 600 meters of high voltage overhead Power Lines: a case-control study. Rom J Intern Med 55: 145-150, 2017.

Saha S, Woodbine L, Haines J, Coster M, Ricket N, Barazzuol L, Ainsbury E, Sienkiewicz Z, Jeggo P. Increased apoptosis and DNA double-strand breaks in the embryonic mouse brain in response to very low-dose X-rays but not 50 Hz magnetic fields. JR Soc Interface 11: 20140783, 2014.

Savitz DA. Magnetic fields and miscarriage. Epidemiology 13: 1-4, 2002.

Savitz DA, Herring AH, Mezei G, Evenson KR, Terry JW, Jr., Kavet R. Physical activity and magnetic field exposure in pregnancy. Epidemiology 17: 222-5, 2006.

Scholten A and Silny J. The interference threshold of cardiac pacemakers in electric 50 Hz fields. J Med Eng Technol 25: 1-11, 2001.

Scientific Committee of Emerging and Newly Identified Health Risks (SCENIHR) for the Directorate-General for Health & Consumers of the European Commission. Possible Effects of Electromagnetic Fields (EMF) on Human Health. Brussels, Belgium: European Commission, 2007.

Scientific Committee on Emerging and Newly Identified Health Risks (SCENIHR). Health Effects of Exposure to EMF. Brussels, Belgium: European Commission, 2009.

Scientific Committee on Emerging and Newly Identified Health Risks (SCENIHR). Memorandum on the Use of the Scientific Literature for Human Health Risk Assessment Purposes – Weighing of the Evidence and Expression of Uncertainty. Brussels: European Commission, 2012.

Scientific Committee on Emerging and Newly Identified Health Risks (SCENIHR). Opinion on Potential Health Effects of Exposure to Electromagnetic Fields (EMF). Brussels, Belgium: European Commission, 2015.

Scientific Committee on Health, Environment and Emerging Risks (SCHEER). Memorandum on weight of evidence and uncertainties, Revision 2018. Brussels: European Commission, 2018.

Scientific Committee on Toxicity, Ecotoxicity, and the Environment (CSTEE). Opinion on Possible Effects of Electromagnetic Fields (EMF), Radio Frequency Fields (RF) and Microwave Radiation on Human Health. Brussels: European Commission, 2001.

Scientific Steering Committee of the European Commission (SSC). 1998. "Opinion on possible health effects from exposure to electromagnetic fields (0 Hz- 300 GHz) - Report and opinion adopted at the meeting of the Scientific Steering Committee of 25-26 June 1998." Accessed January 24, 2017. https://ec.europa.eu/food/sites/food/files/safety/docs/sci-com_ssc_out19_en.pdf.

Seckler T, Stunder D, Schikowsky C, Joosten S, Zink MD, Kraus T, Marx N, Napp A. Effect of lead position and orientation on electromagnetic interference in patients with bipolar cardiovascular implantable electronic devices. Europace 19: 319-328, 2017.

Seifirad S, Farzampour S, Nourbakhsh M, Amoli MM, Razzaghy-Azar M, Larijani B. Effects of extremely low frequency electromagnetic fields on paraoxonase serum activity and lipid peroxidation metabolites in rat. J Diabetes Metab Disord 13: 85, 2014.

Sermage-Faure C, Demoury C, Rudant J, Goujon-Bellec S, Guyot-Goubin A, Deschamps F, Hemon D, Clavel J. Childhood leukaemia close to high-voltage power lines--the Geocap study, 2002-2007. Br J Cancer 108: 1899-906, 2013.

Shenasa M, Heidary S, Shenasa H. Inappropriate ICD shock due to hot tub-induced external electrical interference. J Electrocardiol 51: 852-855, 2018.

Shepherd S, Lima MAP, Oliveira EE, Sharkh SM, Jackson CW, Newland PL. Extremely Low Frequency Electromagnetic Fields impair the Cognitive and Motor Abilities of Honey Bees. Sci Rep. 8: 7932. 2018. doi: 10.1038/s41598-018-26185-y.

Slottje P, van Moorselaar I, van Strien R, Vermeulen R, Kromhout H, Huss A. Electromagnetic hypersensitivity (EHS) in occupational and primary health care: A nation-wide survey among general practitioners, occupational physicians and hygienists in the Netherlands. Int J Hyg Environ Health. 220(2 Pt B):395-400, 2017. doi: 10.1016/j.ijheh.2016.11.013.

Soffritti M, Belpoggi F, Lauriola M, Tibaldi E, Manser M, Accurso D, Chiozzotto D, Giuliani L. Mega-experiments on the carcinogenicity of Extremely Low Frequency Magnetic Fields (ELFMF) on Sprague-Dawley rats exposed from fetal life until spontaneous death: plant of the project and early results on mammary carcinogenesis. In Non-Thermal Effects and Mechanisms of Interaction between Electromagnetic Fields and Living Matter-An ICEMS Monograph (Eur J Oncol - Library Vol 5), edited by L. Giuliani and M. Soffritti. Bologna, Italy: National Institute for the Study and Control of Cancer and Environmental Diseases "Bernardino Ramazzini", 2010.

Soffritti M, Tibaldi E, Padovani M, Hoel DG, Giuliani L, Bua L, Lauriola M, Falcioni L, Manservigi M, Manservisi F, Belpoggi F. Synergism between sinusoidal-50 Hz magnetic field and formaldehyde in triggering carcinogenic effects in male Sprague-Dawley rats. Am J Ind Med 59: 509-21, 2016a.

Soffritti M, Tibaldi E, Padovani M, Hoel DG, Giuliani L, Bua L, Lauriola M, Falcioni L, Manservigi M, Manservisi F, Panzacchi S, Belpoggi F. Life-span exposure to sinusoidal-50 Hz magnetic field and acute low-dose gamma radiation induce carcinogenic effects in Sprague-Dawley rats. Int J Radiat Biol 92: 202-14, 2016b.

Sommer AM and Lerchl A. The risk of lymphoma in AKR/J mice does not rise with chronic exposure to 50 Hz magnetic fields (1 microT and 100 microT). Radiat Res 162:194-200, 2004.

Stunder D, Seckler T, Joosten S, Zink MD, Driessen S, Kraus, T, Marx N; Napp A. In vivo study of electromagnetic interference with pacemakers cause by everyday electric and magnetic fields. Circulation135:907-909, 2017. DOI: 10.1161/CIRCULATIONAHA.116.024558.

Su L, Zhao C, Jin Y, Lei Y, Lu L, Chen G. Association between parental occupational exposure to extremely low frequency magnetic fields and childhood nervous system tumors risk: A metaanalysis. Sci Total Environ 642: 1406-1414, 2018.

Sudan M, Arah OA, Becker T, Levy Y, Sigsgaard T, Olsen J, Vergara X, Kheifets L. Reexamining the association between residential exposure to magnetic fields from power lines and childhood asthma in the Danish National Birth Cohort. PLoS One 12: e0177651, 2017.

Susser M. What is a cause and how do we know one? A grammar for pragmatic epidemiology. Am J Epidemiol 133:635-48, 1991.

Swanson J and Bunch KJ. Reanalysis of risks of childhood leukaemia with distance from overhead power lines in the UK. J Radiol Prot 38: N30-N35, 2018.

Swedish Radiation Protection Authority (SSI). Fifth annual report from SSI's Independent Expert Group on Electromagnetic Fields, 2007. SSI Rapport 2008:12. Stockholm, Sweden: SSI, 2008.

Swedish Radiation Protection Authority (SSI). Fourth annual report from SSI's Independent Expert Group on Electromagnetic Fields, 2006: Recent Research on EMF and Health Risks. SSI Rapport 2007:04. Stockholm, Sweden: SSI, 2007.

Swedish Radiation Safety Authority (SSM). Recent Research on EMF and Health Risks: Seventh annual report from SSMs independent Expert Group on Electromagnetic Fields 2010. Report Number: 2010:44. Stockholm, Sweden: SSM, 2010.

Swedish Radiation Safety Authority (SSM). Research 2013:19. Eighth Report from SSM:s Scientific Council on Electromagnetic Fields. Stockholm, Sweden: SSM, 2013.

Swedish Radiation Safety Authority (SSM). Research 2014:16. Recent Research on EMF and Health Risk. Ninth report from SSM's Scientific Council on Electromagnetic Fields. Stockholm, Sweden: SSM, 2014.

Swedish Radiation Safety Authority (SSM). Research 2015:19. Recent Research on EMF and Health Risk - Tenth report from SSM's Scientific Council on Electromagnetic Fields. Stockholm, Sweden: SSM, 2015.

Swedish Radiation Safety Authority (SSM). Research 2016:15. Recent Research on EMF and Health Risk - Eleventh report from SSM's Scientific Council on Electromagnetic Fields, 2016. Including Thirteen years of electromagnetic field research monitored by SSM's Scientific Council on EMF and health: How has the evidence changed over time? Stockholm, Sweden: SSM, 2016.

Swedish Radiation Safety Authority (SSM). Research 2018:09. Recent Research on EMF and Health Risk - Twelfth report from SSM's Scientific Council on Electromagnetic Fields, 2017. . Stockholm, Sweden: Swedish Radiation Safety Authority (SSM), 2018.

Szemerszky R, Gubanyi M, Arvai D, Domotor Z, Koteles F. Is there a connection between electrosensitivity and electrosensibility? A Replication Study. Int J Behav Med 22: 755-63, 2015.

Tell RA, Nelson JC, Lambdin DL, Athey TW, Hankin, NN, James, Jr, DE. An Examnaton of Electric fields Under EHV Overhead Power Transmission Lines. U.S. Environmental Protection Agency: Silver Spring Maryland, April, 1977 EPA-520/2-76-008

Thomas DC. Re: Kheifets et al. (2017): Residential magnetic fields exposure and childhood leukemia: a population-based case-control study in California. Cancer Causes Control, 2018 [Epub ahead of print].

Tiikkaja M, Aro AL, Alanko T, Lindholm H, Sistonen H, Hartikainen JE, Toivonen L, Juutilainen J, Hietanen M. Testing of common electromagnetic environments for risk of interference with cardiac pacemaker function. Saf Health Work 4: 156-9, 2013a.

Tiikkaja M, Aro AL, Alanko T, Lindholm H, Sistonen H, Hartikainen JE, Toivonen L, Juutilainen J, Hietanen M. Electromagnetic interference with cardiac pacemakers and implantable cardioverter-defibrillators from low-frequency electromagnetic fields in vivo. Europace 15: 388-94, 2013b.

Toivonen L, Valjus J, Hongisto M, Metso R. The influence of elevated 50 Hz electric and magnetic fields on implanted cardiac pacemakers: The role of the lead configuration and programming of the sensitivity. Pacing Clin Electrophysiol 14: 2114-2122, 1991.

Turner MC, Benke G, Bowman JD, Figuerola J, Fleming S, Hours M, Kincl L, Krewski D, McLean D, Parent ME, Richardson L, Sadetzki S, Schlaefer K, Schlehofer B, Schüz J, Siemiatycki J, Tongeren MV, Cardis E. Interactions between occupational exposure to extremely low frequency magnetic fields and chemicals for brain tumour risk in the INTEROCC study. Occup Environ Med. 74 :802-809, 2017. doi: 10.1136/oemed-2016-104080.

Tsujiuchi S, Sivan-Loukianova E, Eberl DF, Kitagawa Y, Kadowaki T. Dynamic range compression in the honey bee auditory system toward waggle dance sounds. PLoS One. Feb 21: e2, 2007.

Turner MC, Benke G, Bowman JD, Figuerola J, Fleming S, Hours M, Kincl L, Krewski D, McLean D, Parent ME, Richardson L, Sadetzki S, Schlaefer K, Schlehofer B, Schuz J, Siemiatycki J, Van Tongeren M, Cardis E. Occupational exposure to extremely low frequency magnetic fields and brain tumour risks in the INTEROCC study. Cancer Epidemiol Biomarkers Prev, 2014.

Turner MC, Benke G, Bowman JD, Figuerola J, Fleming S, Hours M, Kincl L, Krewski D, McLean D, Parent ME, Richardson L, Sadetzki S, Schlaefer K, Schlehofer B, Schüz J, Siemiatycki J, Tongeren MV, Cardis E. Interactions between occupational exposure to extremely low frequency magnetic fields and chemicals for brain tumour risk in the INTEROCC study. Occup Environ Med 74: 802-809, 2017.

US Environmental Protection Agency (USEPA). Guidelines for Developmental Toxicity Risk Assessment. EPA/600/FR-91/001. Washington, DC: US Environmental Protection Agency, Risk Assessment Forum, 1991.

US Environmental Protection Agency (USEPA). Health Effects Test Guidelines OPPTS 870.3700 Prenatal Developmental Toxicity Study, EPA 712-C-980207. Washington, DC: US Environmental Protection Agency, 1998.

US Environmental Protection Agency (USEPA). A Review of the Reference Dose and Reference Concentration Processes. EPA/630/P-02/002F, Final Report. Washington, DC: US EPA, 2002.UK Childhood Cancer Study Investigators (UKCCS). Childhood cancer and residential proximity to power lines. Br J Cancer 83: 1573-80, 2000.

U.S. Environmental Protection Agency (USEPA). Guidelines for Carcinogen Risk Assessment and Supplemental Guidance for Assessing Susceptibility from Early-Life Exposure to Carcinogens. EPA/630/P-03/001F. Washington, D.C.: USEPA, 2005.

van Moorselaar I, Slottje P, Heller P, van Strien R, Kromhout H, Murbach M, Kuster N, Vermeulen R, Huss A. Effects of personalised exposure on self-rated electromagnetic hypersensitivity and sensibility - A double-blind randomised controlled trial. Environ Int, 2016.

van Moorselaar I, Slottje P, Heller P, van Strien R, Kromhout H, Murbach M, Kuster N, Vermeulen R, Huss A. Effects of personalised exposure on self-rated electromagnetic hypersensitivity and sensibility - A double-blind randomised controlled trial. Environ Int. 99: 255-262, 2017. doi: 10.1016/j.envint.2016.11.031.

Vergara X, Kheifets L, Greenland S, Oksuzyan S, Cho YS, Mezei G. Occupational exposure to extremely low-frequency magnetic fields and neurodegenerative disease: a meta-analysis. J Occup Environ Med 55: 135-46, 2013.

Vergara XP, Kavet R, Crespi CM, Hooper C, Silva JM, Kheifets L. Estimating magnetic fields of homes near transmission lines in the California Power Line Study. Environ Res. 140:514-23, 2015.

Verrender A, Loughran SP, Dalecki A, Freudenstein F, Croft RJ. Can explicit suggestions about the harmfulness of EMF exposure exacerbate a nocebo response in healthy controls? Environ Res. 166:409-417, 2018.

Vinceti M, Malagoli C, Fabbi S, Kheifets L, Violi F, Poli M, Caldara S, Sesti D, Violanti S, Zanichelli P, Notari B, Fava R, Arena A, Calzolari R, Filippini T, Iacuzio L, Arcolin E, Mandrioli J, Fini N, Odone A, Signorelli C, Patti F, Zappia M, Pietrini V, Oleari P, Teggi S, Ghermandi G, Dimartino A, Ledda C, Mauceri C, Sciacca S, Fiore M, Ferrante M. Magnetic fields exposure from high-voltage power lines and risk of amyotrophic lateral sclerosis in two Italian populations. Amyotroph Lateral Scler Frontotemporal Degener 18: 583-589, 2017.

Villarini M, Ambrosini MV, Moretti M, Dominici L, Taha E, Piobbico D, Gambelunghe C, Mariucci G. Brain hsp70 expression and DNA damage in mice exposed to extremely low frequency magnetic fields: a dose-response study. Int J Radiat Biol 89: 562-70, 2013.

von Olshausen G, Rondak IC, Lennerz C, Semmler V, Grebmer C, Reents T, Ammar-Busch S, Buiatti A, Bourier F, Deisenhofer I, Kolb C. Electromagnetic interference in implantable cardioverter defibrillators: present but rare. Clin Res Cardiol 105: 657-65, 2016.

Vuokko A, Karvala K, Lampi J, Keski-Nisula L, Pasanen M, Voutilainen R, Pekkanen J, Sainio M. Environmental Intolerance, Symptoms and Disability Among Fertile-Aged Women. Int J Environ Res Public Health. 15: pii: E293, 2018.

Weir E, Schabas R, Wilson K, Mackie C. Commentary: A Canadian framework for applying the precautionary principle to public health issues. Revue Canadienne de Santé Publique 101: 396-398, 2010.

Wen J, Jiang S, Chen B. The effect of 100 Hz magnetic field combined with X-ray on hepatoma-implanted mice. Bioelectromagnetics 32: 322-4, 2011.

Wenner AM. Sound Communication in Honeybees. Sci. Amer. 210:116-124, 1964.

Wilson JW, Haines J, Sienkiewicz Z, Dubrova YE. The effects of extremely low frequency magnetic fields on mutation induction in mice. Mutat Res 773: 22-6, 2015.

Wong O, Harris F, Wang Y, Fu H. A hospital-based case-control study of non-Hodgkin lymphoid neoplasms in Shanghai: analysis of personal characteristics, lifestyle, and environmental risk factors by subtypes of the WHO classification. J Occup Environ Med 52: 39-53, 2010.

Woodbine L, Haines J, Coster M, Barazzuol L, Ainsbury E, Sienkiewicz Z, Jeggo P. The rate of X-ray-induced DNA double-strand break repair in the embryonic mouse brain is unaffected by exposure to 50 Hz magnetic fields. Int J Radiat Biol 91: 495-9, 2015.

World Health Organization (WHO). Environmental Health Criteria 238: Extremely Low Frequency (ELF) Fields. Geneva, Switzerland: WHO, 2007.

World Health Organization (WHO). Environmental Health Criteria 35: Extremely Low Frequency (ELF) Fields. Geneva, Switzerland: WHO, 1984.

World Health Organization (WHO). WHO Handbook on: Establishing a dialogue on risks from electromagnetic fields. Geneva, Switzerland: WHO, 2002.

World Health Organization (WHO). 2006. Framework for Developing Health-Based Standards. Geneva, Switzerland: World Health Organization, 2006.

Yasui M, Kikuchi T, Ogawa M, Otaka Y, Tsuchitani M, Iwata H. Carcinogenicity test of 50 Hz sinusoidal magnetic fields in rats. Bioelectromagnetics 18:531-540, 1997.

Yuan LQ1, Wang C1, Zhu K2, Li HM1, Gu WZ2, Zhou DM1,3, Lai JQ1, Zhou D1, Lv Y1, Tofani S4, Chen X1,3. The antitumor effect of static and extremely low frequency magnetic fields against nephroblastoma and neuroblastoma. Bioelectromagnetics. 39 :375-385, 2018. doi: 10.1002/bem.22124.

Zaffanella LE. Survey of Residential Magnetic Field Sources. Volume 2: Protocol, Data Analysis, and Management. EPRI TR-102759-V2. Palo Alto, CA: EPRI, 1993.

Zaffanella LE, Kavet R, Pappa JR, Sullivan TP. Modeling magnetic fields in residences: validation of the resicalc program. J Expo Anal Environ Epidemiol 7: 241-259, 1997.

Zhang Y, Lai J, Ruan G, Chen C, Wang DW. Meta-analysis of extremely low frequency electromagnetic fields and cancer risk: a pooled analysis of epidemiologic studies. Environ Int 88: 36-43, 2016a.

Zhang Y, Zhang D, Zhu B, Zhang H, Sun Y, Sun C Effects of dietary green tea polyphenol supplementation on the health of workers exposed to high-voltage power lines. Environ Toxicol Pharmacol 46: 183-187, 2016b. doi:10.1016/j.etap.2016.07.016.

Zhang D, Zhang Y, Shu B, Shang H, Sun Y, Sun C. Reservatrol may reverse the effects of longtem occupational exposure to electromagnetic fields on workers of a power plant. Oncotarget. 8:47497-47506, 2017. doi: 10.18632/oncotarget.17668.

Zhou H, Chen G, Chen C, Yu Y, Xu Z. Association between extremely low-frequency electromagnetic fields occupations and amyotrophic lateral sclerosis: a meta-analysis. PLoS One 7: e48354, 2012.